EFFICIENT CODING OF THE PREDICTION RESIDUAL

By

LEGAND L. BURGE, JR.

Bachelor of Science Oklahoma State University Stillwater, Oklahoma 1972

Master of Science Oklahoma State University Stillwater, Oklahoma 1973

Submitted to the Faculty of the Graduate College of the Oklahoma State University in partial fulfillment of the requirements for the Degree of DOCTOR OF PHILOSOPHY December, 1979



EFFICIENT CODING OF THE PREDICTION RESIDUAL

Thesis Approved: 1 0

Dean of the Graduate College

ACKNOWLEDGMENTS

I extend my sincere thanks and gratitude to Dr. Rao Yarlagadda, my thesis adviser and chairman of my doctoral committee, for his dedication and guidance during the term of my graduate program. His interest and encouragement through understanding has contributed significantly to realize the completion of this dissertation. My appreciation is due to Drs. Bennett Basore, Craig S. Sims, and Ronald D. Schaefer, my committee members, for their helpful comments and generous contribution of time throughout this research.

I owe special recognition to Dr. Cheryl Scott for stimulating discussions concerning aspects of speech science.

A special thanks is due to Drs. Charles Bacon, H. Jack Allison and Robert Mulholland for stimulating discussions and encouragement during this work. A sincere appreciation to Dr. Lynn R. Ebbesen and John Perrault for their assistance and accessibility of computer time.

I would like to thank the United States Air Force for providing me the opportunity to complete my degree through the AFIT Civilian Institutions Program. I am grateful to Lt. Col. John Kitch, Jr., and Capt. Samuel Brown, Jr., for their understanding and helpfulness during my academic tenure.

I am grateful to the Defense Communications Engineering Center (DCA) in Reston, Virginia, for their support during the term of the research. The stimulating discussions and kindness awarded me by Mr. Gerald Helm

iii

and Mr. George Moran is appreciated. The assistance of Mr. William Mills and Ms. Elaine Bernd was greatly appreciated during the term.

My sincere gratitude to Ms. Debbie Perrault and Ms. Louise Sumpter for their excellent typing of this dissertation. I would like to acknowledge Mr. Eldon Hardy for the excellent drawing of the figures.

I owe special thanks to Mr. Legand L. Burge, Sr., Mrs. Bobbie J. Burge, Mr. Richard Jones, Mrs. Wilba Jones, Mrs. Annie Dean, Rev. Dr. and Mrs. Joe Edwards, Rev. and Mrs. Richard Thompson, Mr. and Mrs. Glenn Mosely, Mr. Lynn R. Osborn, Mr. Lew Phillips, and Mr. Raymond A. Young for their encouragement throughout this work.

The cooperation, patience and understanding of my wife, Claudette, has been greatly appreciated. My son and daughter, Legand and LeAnn, are acknowledged for the continued source of motivation for which this effort was begun.

i٧

TABLE OF CONTENTS

I. INTRODUCTION 1.1 Statement of the Problem 1.2 Review of the Literature 1.3 Organization of the Thesis 1.3 Organization of the Thesis 21 II. PREDICTION RESIDUAL AND THE PITCH EXTRACTION 22 2.1 Introduction 22 2.3 Model of the Vocal Tract 24 2.4 A Parallel Between Glottal Waveform and the Residual Signal 33 2.5 Review of Linear Prediction Analysis 33 2.6 Short-Time Analysis 34 2.7 Implementation of Operations for the Calculation of the Prediction Residual 54 2.8 A Novel Approach to Pitch Extraction 54 2.8.1 Types of Problems Associated with Pitch Extraction 54 2.8.2 Advantages and Disadvantages for Using the Prediction Residual as a Source for Pitch Extraction 54 2.8.3 A Novel Pitch Extractor 56 2.8.4 Pitch Extraction Results 56 2.9 Summary 56 3.1 Introduction 56 3.2 Coding Methods 56 3.3 Transform Coding 57 3.4.1 Sub-Band Coding and Transform Coding 38 3.5 Determination of Frequency Sub-Bands Based on Articulation Information 56 3.5 Determination of Frequency Sub-Bands	napter			Page
1.1 Statement of the Problem	Ι.	INTRODUCT	ION	. 1
II. PREDICTION RESIDUAL AND THE PITCH EXTRACTION 22 2.1 Introduction 22 2.2 Mechanism of Speech Production 22 2.3 Model of the Vocal Tract 33 2.4 A Parallel Between Glottal Waveform and the Residual Signal 33 2.5 Review of Linear Prediction Analysis 33 2.6 Short-Time Analysis 34 2.7 Implementation of Operations for the Calculation of the Prediction Residual 55 2.8 A Novel Approach to Pitch Extraction 56 2.8.1 Types of Problems Associated with Pitch Extraction 56 2.8.2 Advantages and Disadvantages for Using the Prediction Residual as a Source for Pitch Extraction 56 2.8.3 A Novel Pitch Extractor 56 2.8.4 Pitch Extraction Results 56 2.9 Summary 56 3.1 Introduction 56 3.2 Coding Methods 56 3.3 Transform Coding 77 3.4.1 Sub-Band Coding and Transform Coding 88 3.5 Determination of Frequency Sub-Bands Based on Articulation Index 88 3.6 Transitional Information 59 3.7 Relation of Perception to Intelligible Speech 9 3.8.1 Energy Distribution 9 <td></td> <td>1.1 1.2 1.3</td> <td>Statement of the Problem</td> <td>. 1 . 8 . 20</td>		1.1 1.2 1.3	Statement of the Problem	. 1 . 8 . 20
2.1 Introduction	II.	PREDICTIO	N RESIDUAL AND THE PITCH EXTRACTION	. 22
Residual Signal 33 2.5 Review of Linear Prediction Analysis 34 2.6 Short-Time Analysis 34 2.7 Implementation of Operations for the Calculation of the Prediction Residual 44 2.7 Implementation of Operations for the Calculation of the Prediction Residual 55 2.8 A Novel Approach to Pitch Extraction 56 2.8.1 Types of Problems Associated with Pitch Extraction 56 2.8.2 Advantages and Disadvantages for Using the Prediction Residual as a Source for Pitch Extraction 56 2.8.3 A Novel Pitch Extractor 56 2.8.4 Pitch Extraction Results 56 2.8.5 A Novel Pitch Extractor 56 2.8.4 Pitch Extraction Results 56 2.9 Summary 66 3.1 Introduction 66 3.2 Coding Methods 76 3.4 Sub-Band Coding 78 3.4 Sub-Band Coding 78 3.5 Determination of Frequency Sub-Bands Based on Articulation Index 69 3.6 Transitional Information 69 3.8		2.1 2.2 2.3 2.4	Introduction	22 24 34
of the Prediction Residual		2.5 2.6 2.7	Residual Signal	. 35 . 39 . 48
Pitch Extraction 54 Pitch Extraction 54 2.8.2 Advantages and Disadvantages for Using the Prediction Residual as a Source for Pitch Extraction 54 2.8.3 A Novel Pitch Extractor 54 2.8.4 Pitch Extractor 65 2.9 Summary 65 2.9 Summary 65 3.1 Introduction 64 3.2 Coding Methods 64 3.3 Transform Coding 74 3.4 Sub-Band Coding 74 3.4.1 Sub-Band Coding and Transform Coding 84 3.5 Determination of Frequency Sub-Bands Based on Articulation Index 84 3.6 Transitional Information 94 3.7 Relation of Perception to Intelligible Speech 94 3.8 Basis for Coding the Prediction Residual 94 3.8.1 Energy Distribution 10		2.8	of the Prediction Residual	. 53 . 54
for Pitch Extraction 54 2.8.3 A Novel Pitch Extractor 65 2.8.4 Pitch Extraction Results 66 2.9 Summary 66 3.1 Introduction 66 3.2 Coding Methods 67 3.3 Transform Coding 76 3.4 Sub-Band Coding 76 3.5 Determination of Frequency Sub-Bands Based on 86 3.6 Transitional Information 94 3.7 Relation of Perception to Intelligible Speech 97 3.8 Basis for Coding the Prediction Residual 99 3.8.1 Energy Distribution 10			2.8.2 Advantages and Disadvantages for Using the Prediction Residual as a Source	. 54
III. SUB-BAND CODING OF THE PREDICTION RESIDUAL	•	2.9	for Pitch Extraction	58 61 63 67
 3.1 Introduction	III.	SUB-BAND	CODING OF THE PREDICTION RESIDUAL	. 68
3.5Determination of Frequency Sub-Bands Based on Articulation Index		3.1 3.2 3.3 3.4	Introduction	68 69 76 80 85
3.9 Summary 11		3.5 3.6 3.7 3.8 3.9	Articulation of Frequency Sub-Bands Based on Articulation Index	. 86 . 94 . 97 . 99 . 101 . 110

Chapter

IV.	ENERGY	BASED SUB	-BAND CODING ALGORITHM	•	•	111
	4. 4. 4. 4. 4. 4.	1 Introd 2 Bit Al 3 Sub-Ba 4 Adapti 5 Signal 6 Comput 7 Summar	uction	• • • • •	• • • •	111 112 120 131 135 138 141
۷.	SUMMARY	AND SUGG	ESTIONS FOR FURTHER STUDY	•	•	143
BIBLIO	5. 5. GRAPHY	1 Summar 2 Sugges 5.2.1 5.2.2 5.2.3 5.2.4 5.2.5 5.2.6	y tions for Further Study PARCOR Coefficient Study of Sensitivity Sub-Band Coding Using Subjective Measurements Energy Threshold Matrix Study Integer-Band Coding of the Prediction Residual Prediction Residual and Noise Modeling the Prediction Residual	• • • • •	• • • • • •	143 146 146 146 146 147 147 147 148 149
APPEND	IXES					
	AP	PENDIX A	- DEFINITIONS RELATED TO SPEECH SCIENCE	•	•	161
	AP	PENDIX B	- COMPUTER PROGRAMS FOR CODING THE PREDICTION RESIDUAL	•	•	165
	AP	PENDIX C	- ARTICULATION INDEX	•	•	189
	AP	PENDIX D	- SONAGRAMS	•		194

Page

vi

LIST OF TABLES

Table	Page	
Ι.	Average of Fundamental and Formant Frequencies and Formant Amplitudes of Vowels by 76 Speakers	
II.	Representation of IPA Phonemes with Examples	
III.	Comparison of Fundamental Frequencies	
IV.	Sub-Band Partitioning Example	
۷.	Adjustments to the Spectrum of the Speech Signal 88	
VI.	Adjustments for Noise Spectrum	
VII.	Frequencies Related to One-Third-Octave-Bank Method 91	
VIII.	Frequencies Related to Octave Method	
IX.	Sentence Data	
Χ.	Phonemic Data	
XI.	Energy by Phoneme for the Predictive Residual 105	
XII.	Phoneme Energy Groupings	
XIII.	Symbolic Representation of Energy Distribution 108	
XIV.	Energy Threshold Matrix	
XV.	Symbolic Representation of Bit Distribution	
XVI.	<u>A</u> <u>Priori</u> Bit Matrix Distribution	
XVII.	Sub-Band Coder Cutoff Frequencies	
XVIII.	Integer-Band Sampling Cutoff Frequencies for 8000 Hertz Sampling Rate	
XIX.	Sub-Band Coder Parameters Relative to High Energy Phonemes . 126	
XX.	Sub-Band Coder Parameters Relative to Low Energy Phonemes . 127	

vii

lable		Page
XXI.	Sub-Band Coder Parameters Relative to Noise Energy Phonemes	128
XXII.	Representation of Samples for a Frame for High Energy Sound	129
XXIII.	Signal-to-Noise Performance Measurement for Several Phonemes	137
XXIV.	Data Blocks for Processing and Storage	138
xxv.	Frequency Bands of Equal Contribution to Articulation Index	192

T . I . 1

LIST OF FIGURES

Figu	re	P	age
1.	Model of Speech Production as the Response of a Quasi- Stationary Linear System	•	3
2.	Block Diagram of LPC Analysis	•	5
3.	Prediction Residual Formed by Speech Through an Inverse Filter	•	23
4.	Cross-Sectional View of the Human Vocal Tract System \ldots .	•	26
5.	Distinctive Feature of the Phonemes of English Indicating the Presence or Absence of a Feature	•	33
6.	Speech Waveform and Its Prediction Residual for the Phoneme /æ/ over 256 Sample Interval	•	37
7.	Spectrum of Residual Signal for the Phoneme / ae /	•	38
8.	Implementation for Generation of Forward and Backward Prediction Errors	•	41
9.	Detailed Structure of PARCOR Implementation for jth Stage	•	49
10.	Analysis and Synthesis Models for Lattice Structure	•	50
11.	Sequence of Operations Related to Calculation of the Prediction Residual, r(n)	•	55
12.	Two Methods of Determining Pitch Period	•,	57
13.	Prediction Residual Waveform for Phoneme /æ/ over 256 Sample Segment	•	60
14.	Block Diagram of Pitch Extraction Method	•	64
15.	Basic Component of Channel Vocoder	•	71
16.	Block Diagram of Channel Vocoder Analyzer	•	72
17.	Block Diagram of Channel Vocoder Synthesizer	•	73
18.	LPC Vocoder	•	74

Figu	Page	
19.	Block Diagram of the Implementation of Transform Coding 77	
20.	Partitioning of Frequency Spectrum into Four Sub-Bands 83	
21.	Sequence of Operations Relating to the nth Sub-Band 84	
22.	Relation Between AI and Various Measures of Speech Intelligibility	
23.	Transitional Cueing for Consonant-Vowel for Phonemes /a/ and /d/	I
24.	Normalized Energy Distribution by Sub-Band	
25.	Power Density of Speech Signal	
26.	Distribution of the Normalization Factor	
27.	Flow Chart for Bit Allocation	
28.	Bit Distribution for Sub-Bands by Energy Bands	
29.	Bit Distribution for Phonemes by Frame	
30.	Characteristic of the Adaptive Uniform Quantizer 133	
31.	Flow Chart for Coding Residual Signal	1
32.	Composite Articulation Index vs. Cutoff Frequencies of Ideal Lowpass Filters	

LIST OF SYMBOLS

	M	-	Total number of data points
ی مربعہ کا آباد	s _n	-	Signal
	a _i	-	Forward lattice LPC filter coefficients
	^b i	-	Backward lattice LPC filter coefficients
	R(k)	- 1	Residual signal k = 0, 1,, M-1
	p	· • <u>-</u>	Order of the LPC filter
	F ₁ , F ₂ ,	-	Formant frequencies
	r _l (m)	-	Output of the £th bandpass filter
			$m = 0, 1,, M-1; \ell = 1, 2,, B$
	r _e (k)	-	Decimated signal r(k)
	ω _L	-	Bandpass filters are assumed to be contiguous;
			$\ensuremath{\mathfrak{l}}$ gives the lowest frequency corresponding to
			the £th bandpass filter
	h _n (k)	-	Impulse response of nth lowpass filter
			associated with the nth sub-band
	W _R	-	Bandwidth of the <code>lth bandpass filter</code>
	^ω n	-	Edges of the bandpass filter
	e _n (k)	-	Modulated signal of sub-band
	N, N _i	-	Samples available per frame
	f _o	-	Fundamental frequency
	f ₁ , f ₂ ,	-	Harmonics of the fundamental
	fs	-	Original sampling frequency
	f _{si}	-	Sampling frequency for the ith sub-band

xi

g(t), G(ω)	-	Represents glottis excitation function in the
		time and frequency domains
$V(t)$, $V(\omega)$	-	Represents vocal tract filter in the time and
		frequency domains
s(t), S(ω)	-	Represents speech waveform in time and fre-
		quency domains
x(n), y(n), x(k), y(k)	-	Speech sample sequency
x _f (n)	-	Forward prediction of x(n)
x _b (n)	-	Backward prediction of x(n)
e _f (m)	-	Forward prediction error or prediction resid-
		ual signal
e _{fn} (k)	-	Output of the nth bandpass filter
e _b (k)	-	Backward prediction error
R _i	-	Autocorrelation of x(n)
Ej	-	Residual signal at the jth stage
C _j	-	Cross correlation between forward and backward
		prediction errors
k _{j+1}	-	Partial correlation coefficients (PARCOR) of
		(j+l)st stage
A _p (z)	-	Transfer function between forward prediction
		error and speech sequence for pth stage
B _p (z)	-	Transfer function between backward prediction
		error and speech sequence for pth stage
r _{j+l}	-	Reflection coefficient of (j+l)st stage
σ ²		Variance of speech source
у	-	Transform coding coefficients
A	-	n x n matrix

D	-	Mean-square overall distortion
J _i ,	-	Number of bits/sample
δ	-	Correction factor for a quantizer
R	-	Average bit rate
R _{xx} , R _{yy}	-	Covariance matrices for x and y
λ _i	-	Eigenvalue of R _{XX}
k _{ij}	-	Bits/sample corresponding to the ith energy
		and jth frequency band
E _{fn} (k)	_	Discrete Fourier transform coefficient
E _n	-	Energy of the nth sub-band
E _{ij}	-	Energy of the signal for the ith energy band
		and jth frequency band
E ^T ij	· -	Energy threshold for the ith energy band and
		jth frequency band
k ^A ij	-	<u>A priori</u> bits/sample corresponding to the ith
		energy band and jth frequency band
σi	-	Normalization factor
^α n	-	Discrete Cosine Transform coefficient
Δ	-	Quantizer step size
ψ	-	Quantizer array
Q _e	-	Quantization value array
ς _ℓ	-	Quantizer level

xiii

CHAPTER I

INTRODUCTION

1.1 Statement of the Problem

The structural unit of speech composition is the speech sound called the phoneme.[†] Its variations are called allophones. It can be said also that phonemes relate to the linguistic basis of a language. However, phonemes are not "bricks," i.e., the human has been endowed with the ability to communicate in a continuous mode. Because we speak in an uninterrupted fashion in order to complete our thoughts, the phonemic structure connects itself by transitional cues for the perception of certain phonemes [1]. It is this transitional information that is needed for absolute discrimination of speech and speech-like sounds [2]. It is the transitional information that is needed for efficient excitation of a speech synthesizer.

To synthesize intelligible speech, the perceptual aspects of speech sounds have to be used. In other words, the ability for humans to discriminate and differentiate a speech sound with their over-learned senses must be incorporated into the speech synthesis technique. The speech synthesis must include perceptual enhancement, and the inclusion of transitional information (that is, frequency shifts). Transitional information is the loci of frequency determined by the place of

[†]Some of the words related to the science of the speech waveform are defined in APPENDIX A.

articulation that connects the phonemes. Phonemes are the basic speech sound element used to make a word. One can also say that a phoneme is an idealized structural unit of language which serves to keep words apart. It is an astonishing fact as to how the human brain stores rules to keep track to one's language for communicating. The object of speech synthesis is to come as close as possible to this occurrence.

The history of synthetic voice coding had its origination with H. W. Dudly in 1939 [3] [4]. The Dudley speech reproduction model consists of a filter representing the vocal tract resonance characteristics driven by an artificially synthesized excitation signal. The filter and the excitation signal parameters are updated periodically. To determine the filter characteristics, Dudley used the Fourier spectrum of the speech as a basis. The excitation signal consists of a pulse train for voiced sounds and random noise for unvoiced sounds. The model that Dudley has represented is essentially the basis of many methods today [5] [6] [7]. Some of these ideas are discussed below.

A basic model of the speech waveform is to assume a linear quasi time-invariant system which responds to a periodic or noiselike excitation. This linear time invariant system represents the vocal tract. If the vocal tract is assumed to be fixed, then the output of the system is a convolution between the excitation and vocal tract transfer function (see Figure 1).

Recently considerable interest has been given to methods of digital analysis and synthesis of speech assuming the presented model. A method that has proven to be efficient for encoding the speechwave is linear prediction [6]. The linear predictive encoder was developed to improve the channel vocoder voice quality and intelligibility [7]. The difference



Figure 1. Model of Speech Production as the Response of A Quasi-Stationary Linear System (a) Time Domain Characterization, (b) Frequency-Domain Characterization (After Oppenheim, 1978)

between the linear predictive coded (LPC) vocoder and the channel vocoder is the filter. There are two types of LPC vocoders, a pitch-excited and a residual-excited. The difference between the two is how the excitation signal is characterized for the synthesis filter. In the pitch-excited LPC vocoder, the model of the vocal tract, with glottal flow and radiation, is represented by the predictor coefficients. These coefficients are transmitted together with the information regarding the excitation of the speech, i.e., pitch, voiced/unvoiced decision and the gain. Much research has been done toward the pitch-excited LPC vocoder. Two methods have been discovered, the autocorrelation [8] [9] and the covariance [6] methods. The residual-excited methods can be characterized the same way. However, instead of using pitch, voiced/unvoiced desicion and gain, the residual is encoded and transmitted. The residual is the difference between the actual and predicted speech signals. This technique also carries the name adaptive predictive coding (APC). The channel vocoder, on the other hand, uses a set of narrowband filters whereas the linear predictor uses an all pole digital filter. The linear predictive filter describes the frequency response of the vocal tract system by the predictor coefficients. Its function is to decompose the speech into two waveforms. One waveform represents the parameters that are time-varying such as predictor coefficients, partial correlation coefficients and other parameters that represent the formant frequency characteristics. The other waveform is the prediction residual. Figure 2 describes a block diagram of the LPC analysis.

The prediction residual is the ideal signal for an excitation function for the linear predictive analysis and synthesis model because it contains the actual information instead of the pseudo-model, a pulse



S



train or random noise [10]. In addition, phasing information is embedded in the prediction residual. Furthermore, since the analysis filter is the inverse of the synthesis filter, the decomposed waveform can be reconstructed to form the input speech waveform by updating the parameters [7] [11]. The prediction residual also follows the actual speech excitation model g(t) in Figure 1 [12]. The function, g(t), represents the glottal pulse which is also called the glottal volume velocity at the vocal cords or glottis. In order to ideally model the voice reproduction system, it is necessary to use a system whose properties are similar acoustically to the glottis and vocal tract. It is best to model the excitation signal with an analogous function to the glottis waveform for input to the vocal tract. It is well known that for nonnasal voiced speech sounds, the transfer functions have no zeros [5]. For these particular sounds, the vocal tract filters can be approximated by an all pole filter. It is also known that the shape and periodicity of the glottis excitation are subject to large variations [12]. However, with the linear predictive model the features of the glottal flow, the vocal tract, and the radiation, which is the output from the mouth, are included into a single recursive filter. To separate the glottal flow from the vocal tract involves a deconvolution. Some authors have avoided this separation of the source function; however, the artificial excitation used by them represents only a good approximation to the prediction residual for unvoiced sounds. Moreover, for voiced sounds, the artificial excitation could be improved. The prediction residual should be used for the excitation function, because it contains the following characteristics:

1. It is repetitive at the pitch frequency.

2. It has basically a flat amplitude spectrum; however, it includes

details that relate to the suprasegmentals of the individual and of the spoken words.

3. It includes the noisiness of opening and closing of the glottal mechanism indicating phase information.

4. It includes the fact that voiced fricatives and stops are a combination of noise and a repetitive signal.

Noting the speech characteristics in the residual signal, several authors have investigated the coding aspects of the prediction residual [13-32]. However, the speech intelligibility aspects, such as Articulation Index (AI) [29], have not been used in these. The Articulation Index concept has been used effectively in the sub-band coding of speech [36]. The sub-band coding, based upon AI, allows for an efficient bit distribution in coding. This thesis combines all these ideas and presents an efficient method of coding the prediction residual using the concepts of sub-band coding. A literature survey related to these areas is presented in the next section.

One important aspect of coding is bit rate. For certain narrow band rates, the coding of the prediction residual is not feasible [13]. Also, it has been shown that 9,600 bits/second is feasible for transmission of residual and filter parameters, and is practical over voice grade lines [35]. In the future, lower data rates have to be used for cost effectiveness. At present, rates below 6,000 bits/second yield speech quality of a synthetic nature. Rates between 6,000 bits/second and 16,000 bits/ second demonstrates good communication quality. Studies have shown and present operating equipment demonstrate that a 16,000 bits/second transmission rate and above yield toll telephone quality. The thrust of the governmental community for designing voice switch networks has been

recently toward 9,600 bits/second rate. At this rate, the communicators can comprehend the language spoken; however, there is some drop-off in **spe**aker recognition but not as drastic as at rates closer to 6,000 bits/ second. With the advent of mircroprocessing systems more sophisticated algorithms can be implemented with small monetary investments. This thesis presents the coding and decoding of the residual signal using subband coding at a data rate of 9,600 bits/second.

1.2 Review of the Literature

Predictive systems related to speech have evolved through the years. A brief survey of these systems is presented below. In earlier studies of predictive coding systems with applications to speech signals, the linear predictors were limited to fixed coefficients in an interval [17]. In more recent studies, it was found that since the speech signal has nonstationary properties, the linear predictor does not efficiently predict the signal at each interval. In work by Atal and Schroeder [6], an adaptive predictive system took into account the quasi-periodicity of speech signals. In addition to being the classic forerunner for adaptive predictive coding (APC) of speech signals, this is a more elaborate predictor than the one with fixed coefficients which is suited for characteristics of speech sounds. Basically, the residual signal along with the predictor provides sufficient information for the receiver to regenerate the input. In this, pitch is determined from the residual signal. Atal and Schroeder [22] have examined predictive coding of speech signals recently. They have shown that speech quality can be improved by masking quantizer noise over the speech signal. Atal and Hanauer [5] described an efficient encoding of the speech wave by representing it in terms of time-varying

parameters related to a transfer function of the vocal tract and by modeling the excitation.

In work by Dunn [13], the linear predictive coded residual signal was generated by a feed-forward linear predictive coding (LPC) analyzer and encoded using delta modulation (DM). The signal was transmitted at a bit rate of 9,600 bits/second. Gibson, Jones, and Melsa [14] have introduced a method called sequential adaptive prediction which utilized differential pulse code modulation (DPCM) with an adaptive quantizer and an adaptive predictor using Kalman filtering. This work was improved upon by Cohn and Melsa [15] using adaptive differential pulse code modulation (ADPCM) for encoding the prediction residual. A method using the Kalman filter for the adaptive predictive encoder was introduced by Goldberg and others [16]. This system was real time APC that was implemented on a minicompu-An adaptive residual coding using an adaptive predictor, adaptive ter. quantizer, and a variable length coder was studied by Qureshi and Forney [18]. In these studies, a class of speech digitization algorithms is described for use at bit rates of 9,600 to 16,000 bits/second. These systems involve an adaptive predictor, an adaptive quantizer, and a variable length coder. This is a practical version of a residual encoder previously studied by Melsa and others [14]. Most recently, the method of variable length coding of the prediction residual was studied by Berouti and Makhoul [19]. This system of APC uses a noise spectral shaping filter to solve the granular noise quantization problem and an indefinite quantizer to solve the overload quantizing problem.

A voice-excited predictive coder (VEPC) by Esteban and others [20] uses a baseband excitation of the residual and splitband coding by signal decimation/interpolation. Furthermore, quadrature mirror filters are

implemented in order that the aliasing properties could be taken advantage of in the synthesizer.

The most recent work by Cohn and Melsa [21] [23] involves the implementation of a speech coding algorithm for digital transmission of speech at 9,600 bits/second using a sequential, adaptive linear predictive coder, an adaptive source coder, and multipath tree-searching algorithm to generate quality speech. This is an extension of the previous work done on a residual encoder which was an improved ADPCM system for speech digitization. Chang [24] has extended this work and incorporated a noise resistant code for transmission.

In work by Magill and others [25], a feed-forward LPC analyzer was used with an encoding method of Adaptive Delta Modulation (ADM) and an experimental method of encoding the residual by DPCM. This is referred to as a residual excited linear predictive (RELP) vocoder. It combines the advantages of linear predictive coding and voice-excited vocoding.

Recently, Dankberg and Wong [26] have implemented a new version of the RELP vocoder. Their results have included a development of a pitch predicted ADPCM residual encoder and a harmonic generator. Viswanathan and others [27] considered the use of voice-excited linear predictive (VELP) and RELP coders for speech. They have studied in detail the various aspects of these coders and have attempted to maximize speech quality as a result. They also studied the advantages and disadvantages of baseband residual transmission and baseband speech transmission.

In recent work, Kang [28] studied the development of a narrowband voice digitizer that improves speech quality, intelligibility and reliability. The principle of LPC is used in implementing the lattice filter for the analysis and synthesis. Itakura and Saito [9] [30] have used

the lattice method for LPC analysis of speech. The thrust has been for improved quantization of partial correlation (PARCOR) coefficients. Makhoul [31] has presented a class of stable and efficient lattice methods for linear prediction of speech. In this, an indepth study is made on PARCOR coefficients. If the all pole function is stable, then the lattice obtained from this is stable; furthermore, since the PARCOR coefficients are bounded, stability is guaranteed and an efficient quantization method can be used.

In work by Flanagan [32], it is shown that the residual approximates the glottal waveform. In any excitation system, the closer one can approximate the physical model, the better response one gets from the system. Flanagan's work enhances this concept to use the residual waveform as the excitation to the speech synthesizer.

Rabiner and others [33] have studied the LPC error signal. The work investigated the variation of the prediction error as a function of position in an analysis frame within a single stationary speech segment. The error signal has the frequency range of the actual speech.

The work of Goodman [34] found the analog signal can be divided into several nonoverlapping frequency bands. Each band can be sampled and quantized independently. The result is an improvement in encoding efficiency over straight sampling and quantizing of signals that are spectrum peaked. Crochiere and others [36] [37] have applied this to speech signals in the digital domain. This is referred to as sub-band coding (SBC). This approach provides a means of controlling and reducing quantization noise in the coding.

A pilot study of speech waveform coding techniques were studied by Tribolet and others [38]. The study compared subjective ratings to the

various quality (objective) measures for speech waveform coders. Tribolet and others examined four different speech waveform coder algorithms for low-bit rate applications, and studied these relationships for overall objective and subjective ratings for quality. The algorithms were: adaptive differential PCM with a fixed predictor (ADPCM-F), sub-band coding (SBC), ADPCM with a variable predictor (ADPCM-V) and adaptive transform coding (ATC). The transmission rates studied were 24,000, 16,000, and 9,600 bits/second. The objective measures used were a conventional signal-to-noise ratio, frequency weighted signal-to-noise ratio, log likelihood ratio, and an articulatory bandwidth measure. The results of the study were that if complexity/cost was of no concern, then ATC is the most attractive of the group coders. However, if complexity/cost was a concern, then SBC is an attractive choice. ADPCM-F had the poorest quality for its complexity; ADPCM-V was the most costly for its quality. The transform coding and the sub-band coding will be explained in detail in Chapter II.

In the work by Barabell and Crochiere [39] a new design of the subband coding has been implemented for low-bit rate coding of speech. This study applied quadrature filters to SBC. This method has also employed pitch prediction within the sub-bands. Crochiere [40] has implemented a novel approach for pitch extraction in the SBC. The method uses digital linear phase shifters based on a bandpass interpolation scheme to achieve the non-integer delays necessary in the feedback loop for the pitch predictors. It uses the fractional sample delay in the pitch loop and permits the processing of the pitch prediction in each sub-band to be performed at the sub-band sampling rate which contributes to the efficiency of the algorithm. Pitch detection algorithms that have been mentioned above have one basic goal. That is, make a voiced or unvoiced decision and during certain periods of voiced sounds, estimate the pitch period.

There are three areas of categorization for pitch detectors. First, there is a group that uses time-domain properties of speech signals. These pitch detectors operate directly on the speech waveform in order to estimate the pitch period. The measurements that are usually taken are minimum and maximum amplitude, zero-crossing and autocorrelation measurements. With these detectors, it is assumed the formant structure has been minimized by preprocessing the speech. A second category for pitch detection algorithms uses frequency-domain properties of speech signals. A periodic signal in the time-domain will consist of a series of impulses in the frequency-domain located at the fundamental frequency and its harmonics. Therefore, one can make measurements in the frequency domain to determine the pitch period. The final group combines both time and frequency-domain concepts of the speech signals in order to determine pitch period. This is a technique that is used which flattens the signal with frequency-domain techniques and subsequently uses autocorrelation measures to estimate the pitch period. These are called hybrid techniques. Previous work of the pitch detection algorithms and related works that have been published will be discussed.

There are several documented pitch extraction methods that have been published recently. In earlier methods, analysis of the speech time waveform were attempted by visual inspection of spectrograms which involved the manual determination of pitch [41]. At this time the authors noted the requirement for an automatic scheme of some kind. Pinson [42] used the method of Mathews, Miller, and David [41] to estimate a time-domain

synchronous pitch which in turn was used to determine frequencies and bandwidths of vowel formants.

Sondhi [43] introduced three methods for finding the pitch period. The first method spectrum flattens the signal and corrects the phase to synchronize harmonics. A second method by Sondhi also flattens the spectrum but adds an autocorrelation to determine pitch. The third method center clips the speech signal and uses autocorrelation for determination of pitch. Using the method by Sondhi, a real-time digital hardware pitch detector was implemented by Dubnowski, Schafer, and Rabiner [44].

There are also methods that make use of the power spectrum in the determination of the pitch. One such method is called cepstrum pitch determination. The cepstrum is defined as the power spectrum of the logarithm of the power spectrum, or mathematically expressed, the cepstrum, $Q(\tau)$ [45] [46], is

$$Q(\tau) = \left[\int_0^\infty \log|F(\omega)|^2 \cos(\omega\tau) d\omega\right]^2 \qquad (1.1)$$

where f(t) is the speech signal, ω is the frequency in radians, and

$$F(\omega) = \int_{-\infty}^{\infty} f(t) e^{-j\omega t} dt \qquad (1.2)$$

More recently, using digital inverse filtering techniques, Markel has innovated a method for estimating the fundamental frequency of voiced speech using time-domain analysis. This method has been referred to as a simplified inverse filter tracking (SIFT) algorithm [47]. The pitch period is estimated by an interpolation of the autocorrelation function in the neighborhood of the peak of the autocorrelation function. Another recent algorithm that determines the fundamental frequency of sampled speech is implemented by segmenting the signal into pitch periods. This is done by identifying the beginning of each pitch period. This algorithm is called the data reduction pitch detector by Miller [48]. To obtain the appropriate identity of the beginning of the pitch period, the method detects the cycles of the waveform based on intervals between major zero crossings. The rest of the algorithm determines principal cycles, which correspond to true pitch periods.

In work presented by Gold [49], it is assumed that pitch extraction could be obtained by a visual inspection of the speech wave and is the best obtainable. The computer program contains essentially four sections. First, a voiced/unvoiced decision is made and the two portions are separated. Each voiced portion is labeled as relative maximum, then the peak detector is compiled. The third decision is to determine the spacing; this in turn determines which samples will be called pitch peaks. Finally, a procedure is necessary to eliminate spurious peaks and add into the speech missing pitch peaks. The program is implemented such that editing can make the best pitch selection.

The work of Gold and Rabiner [50] using parallel processing for estimating pitch is a modified version of Gold [49]. A series of measurements are made to find the peaks and valleys of the signals. There are six cases used to determine this. Each is followed to determine if the sample will be an impulse or zero. The rules of this are:

 An impulse equal to the peak of the signal occurs at the point of each peak in time.

2. An impulse equal to the difference between the signal present peak and the past peak amplitude occurs at the point of each peak in time.

3. An impulse equal to the difference between the signal present peak and the past peak amplitude occurs at the point of each peak in time. (If the difference is negative, then it is set to zero.)

4. An impulse equal to the negative of the peak of the signal occurs at each negative peak in time.

5. An impulse equal to the negative of the peak at each negative peak plus the peak of the preceding negative peak occurs at each negative peak in time.

6. An impulse equal to the negative of the peak at each negative peak, plus the negative of the preceding local minimum occurs at each negative peak. (If this difference is negative, then the impulse is set to zero.)

From this technique six estimates are formed. These estimates are combined with the two most recent estimates for each of the six pitch detectors. The values are then compared within an acceptable tolerance; the decision is made for the most occurrences. This value is declared the pitch at that time. An unvoiced decision is made when there is an inconsistency between the comparisons for the pitch period.

Another method by Atal [51] is based upon LPC. This detector initializes with a voiced/unvoiced decision. Upon being classified as voiced, the speech is low-pass filtered and then decimated by five to one. The method uses a 41-pole LPC analysis on 40 ms seconds of frame data to generate the speech harmonics. Then, a Newton transformation is used to spectrally flatten the speech. A peak picker determines the pitch period at the five to one decimated rating. Then, the signal is interpolated and a higher resolution is used to obtain the pitch period.

The average magnitude difference function (AMDF) pitch extractor [52] is a variation of autocorrelation analysis to determine the pitch period of voiced speech sounds. This method takes advantage of the periodicity of voiced speech. It calculates a difference function that at multiples of the pitch period will dip sharply when the delayed speech and original speech are compared. The AMDF function is implemented with subtraction, addition, and absolute value operations, whereas autocorrelation methods use addition and multiplication operations. For this reason, the AMDF function is attractive for real-time operations.

Another real-time pitch extraction method, based on linear predictive techniques, is presented by Maksym [53]. The method employs a nonstationary error process from the adaptive predictive coder by Atal [5]. The algorithm in addition to pitch period extraction also detects voiced speech. The basis of the method uses a predictive one-bit quantizer with an adaptive algorithm for determining prediction coefficients. Since the method operates on the short-term prediction of the speech waveform, the presence of the glottal excitation can be detected.

A semiautomatic pitch detector (SAPD) [54] has been presented by McGonegal, Rabiner, and Rosenberg. This method semiautomatically determines the pitch contour of an utterance. An autocorrelation of the speech is generated. The cepstrum of the unfiltered speech is computed. These displays are shown on a scope on a frame-by-frame basis. The computed pitch period for each waveform is marked by and is displayed to the user. With the incorporation of the three waveforms, an extremely accurate measure is found. The processing is lengthy for an utterance; however, robustness and accuracy of the results can be a trade-off for many applications.

A recent method for estimating pitch period in the presence of noise of voiced sounds is based on a maximum likelihood formulation [55]. This **sch**eme is designed to be resistant to white, Gaussian noise. A new signal is formed from the speech signal with a maximizing function to enhance the peaks for short periods. The function is formed by an autocorrelation of the speech. It provides accurate estimates of the pitch period and can be used to determine formant structure. It is compared with the cepstrum method to perform better under the white noise conditions.

An automatic pitch extraction method was developed by Markel [56] which also determines formant frequency tracking. This method is similar to the cepstral analysis. The technique uses two FFT's to obtain the sequence from which the pitch is extracted. The difference between this method and the cepstral method is the procedure for determining the voiced/unvoiced decision.

An accurate method based on the prediction residual is the method by Atal and Hanauer [5]. The speech is low-pass filtered and each sample is raised to a third power to emphasize the high amplitudes of the speech waveform. A pitch-synchronous correlation analysis is performed of the cubed speech. A voiced/unvoiced decision is made in this technique. A second method is based on a linear prediction representation of the speech waveform. Each sample is predicted from the previous n samples, and therefore the correlation is not good at the beginning of the pitch period. The error is large at the beginning. The basis of the technique is to use peak picking for the pitch detection.

Another accurate method has been described by Itakura and Saito [57]. This method determines the prediciton error signal by the method of lattice filter formulation. The pitch period is determined by computing

autocorrelation coefficients of the residual. A set threshold compares the autocorrelation for a voiced/unvoiced decision with the pitch period.

A two stage method was developed by Boll [58] to determine the pitch period. The method is based on the Itakura [57] algorithm. It is built by adding the initialization of each frame based on the preceeding frame results. The portion of the autocorrelation function of the residual in the range where a pitch pulse is expected and the basis of the <u>a priori</u> information is computed in each frame. The savings in computation is significant.

Two methods were developed by Barnwell and others [59]. These algorithms are: 1) the multiband pitch period (MBPP) estimator, and 2) the skip-sample recursive least squares pitch position estimator. The multiband pitch period estimator first filters the speech waveform into four bands across the frequency regions where a fundamental is expected to occur. The bandwidths of these filters are chosen so that only one of the outputs will be expected to contain the fundamental. Zero-crossing pitch detectors operate on the outputs of each of the filters. The information derived from the zero-crossing detectors is used as a basis for logical operations to produce pitch period estimates. The skip sample recursive least squares technique is based on a recursive least squares linear predictive coder. The coder operates on a lower sampling rate than a linear predictive coder and it uses fewer coefficients than the predictive filter. This approach permits the original sampling time resolution to be retained. The method produces a sharp residual signal whose pitch pulses can be used to determine the period.

The future trend is towards efficient low-bit rate coding that enhances the perceptual quality and intelligibility of speech. The coding of the residual signal is one way of arriving at the desired goal. This thesis presents such an idea along with a novel approach to pitch extraction. The next section presents the organization of the thesis.

1.3 Organization of the Thesis

Chapter II presents the basic ideas associated with the concept of the prediction residual. A discussion of the mechanism of speech production as related to the makeup of speech articulation is presented in speech science terms. A model of the vocal tract is presented in mathematical terms and the residual is presented in an algorithm form. The method of short-time analysis is presented. A new method for determining pitch implementation is presented using the residual waveform as the source function.

Chapter III presents some of the general ideas associated with coding of speech along with some applications. The method of transform coding (TC) is compared to the method of sub-band coding (SBC). The equivalence of the two methods is shown under certain conditions. The Articulation Index (AI) and the phoneme transitional information related to speech intelligibility are discussed along with their incorporation into the coding scheme to enhance the perception of speech. The results of the distribution of energy from the prediction residual of the phonemes are presented.

Chapter IV presents the design of the energy based sub-band coding algorithm. The basic ideas associated with the sub-band coding are discussed as related to the proposed coding scheme. The adaptive quantization is presented to explain the allocation of bits. The result on signal-to-noise ratio (SNR) performance measurements are presented. The computation for coding the prediction residual is presented.

Chapter V presents a summary and suggestions for further study. The appendixes give a sample of the related speech science definitions, computer programs for coding the prediction residual, a brief review of the concept of Articulation Index and sonagrams of speech data.

CHAPTER II

PREDICTION RESIDUAL AND THE PITCH EXTRACTION

2.1 Introduction

Recent work in the area of speech analysis and synthesis is based upon a model that separates the glottal flow from the vocal tract. That is, the speech production is represented by a convolution model where the input corresponds to the glottal volume velocity and the vocal tract by a filter. Recent models have assumed an all pole filter to represent the vocal tract [5]. The filter coefficients are determined by using the method of linear prediction. By using the inverse filter, the speech can be deconvolved to obtain the prediction error or residual. The block diagram representing this is shown in Figure 3. The residual produces a peak where the prediction is bad, representing pitch period designations. As the prediction becomes more accurate, the residual appears as a noisy signal.

Most synthesis models use a filter excited by either a train of quasi-periodic pulses or a random noise source [60]. The periodic source excites the filter for voiced sounds. The noise source excites the filter for unvoiced sounds. The prediction residual is applicable for voiced or unvoiced sounds because the residual is an approximate signal of the corresponding input sources that generate these sounds. The detailed description of the prediction residual is discussed in Section 4 of this chapter.





I
The linear predictive techniques described so far have been used successfully for time-domain speech analysis and synthesis [5] [30]. The linear predictive coding (LPC) techniques have been used in communications in the past; however, it was applied to speech only recently [5] [7]. The use of linear prediction in describing the transfer function of the vocal tract avoids the complexity of Fourier analysis. The slowly time varying aspects of speech can be taken into consideration by updating the filter coefficients every so often.

Two significant contributions have been made by Weiner [61][62] and Shannon [63]. Weiner's work describes prediction and filtering of random, time series data. Shannon's results describe the information content of a message, related to band-width and time requirements of that message, related to band-width and time requirements of that message. The background of this chapter uses Weiner's method as applied to stationary data. Shannon's results are implicitly used in the coding scheme.

Section 2.2 describes the basis of human speech production. Section 2.3 discusses the vocal tract model as a discrete time invariant linear filter. Section 2.4 describes a parallel between the glottal waveform and the residual signal. Section 2.5 reviews linear prediction analysis. Section 2.6 discusses short-time analysis. Section 2.7 describes the implementation of operations for the calculation of the prediction residual. Section 2.8 presents a novel pitch extraction technique.

2.2 Mechanism of Speech Production

Man's system of communication is by speech. Speech is produced through the human vocal system in a continuous fashion. However, speech

signals are composed of a sequence of discrete sounds called phonemes. Although phonemes are not bricks, they are the basic sounds that serve to make a complete word in any language. The connection or arrangement of these sounds is based on certain rules. It is the study of these rules and the way these sounds fit together that is called linguistics. The basic linguistic element is called a phoneme. Its distinguishable variations are called allophones [2].

Speech in humans is produced by a physical acoustic system consisting of principally four parts: lungs, vocal tract, nasal tract and vocal cords (see Figure 4). The lungs supply the volume of air necessary to produce speech. The vocal tract and nasal tract act as filters to shape the waveform. The velum, a small flap of skin, acts as a switch to close the entrance to the nasal tract. When closed, it removes any effect the nasal tract may have on the sound produced. The vocal cords, tongue, teeth and palate are parts of the filter or constriction mechanism. An elongated opening between the folds of the skin which make up the vocal cords is called the glottis.

The vocal tract provides the column of air, which is set to vibration by the excitation of the glottis. In an average male, the vocal tract is about 17 centimenters in length. The cross-sectional area which is determined by the position of the tongue, lips, jaw and velum varies from zero, i.e., complete closure, to approximately 20 square centimeters.

Speech sounds produced by the system can be separated into three distinct classes according to their mode of excitation. The voiced sounds are produced when air is permitted to escape in quasi-periodic pulses by the vibratory actions of the vocal cords. This sets the acoustic system to vibrating at its natural frequencies. These resonant

VOCAL SYSTEM (CROSS SECTIONAL VIEW)



- 1 LIPS
- 2 TEETH
- 3 TEETH RIDGE
- 4 HARD PALATE
- 5 SOFT PALATE (VELUM)
- 6 UVULA
- 7 BLADE OF TONGUE

8 - FRONT OF TONGUE
9 - BACK OF TONGUE
10 - PHARYNX*
11 - EPIGLOTTIS
12 - POSITIONS OF VOCAL CORDS
13 - TIP OF TONGUE
14 - GLOTTIS

Figure 4. Cross-Sectional View of the Human Tract System frequencies are concentrations of energy and are known as formant frequencies. These are useful in characterizing the vocal tract configuration, as there is a one-to-one correspondence in the relationship of vocal tract configuration and formant frequencies. The fricative or unvoiced sounds are generated by forming a constriction at some point along the vocal tract and forcing air through the constriction at a velocity high enough to produce turbulence. This can be identified as wideband noise exciting the vocal tract. For an unvoiced sound the vocal cords are relaxed and partially open. The plosive sounds result from a complete closure of the vocal tract and a sudden or abrupt release of the closure.

The formants or natural resonances are numbered F_1 , F_2 , F_3 , Typically, for speech analysis, only the first three or four are used. Table I gives representative values of these for certain vowels. It has been noted that all phonemes characterize some formant structure; however, it is most noted for voiced sounds [2]. It is indicative of the first formant to be greater in frequency than the fundamental frequency of the vocal tract. The fundamental frequency is the rate of vibration of the vocal cords; whereas, the first formant represents the first concentration of energy of the vocal tract system excited at the fundamental frequency. Typically, the fundamental frequency is around 120 Hertz for men, 220 Hertz for women and 300 Hertz for children. The pitch period is the reciprocal of fundamental frequency. The pitch period has a range from three milliseconds to eight milliseconds for voiced sounds. For the unvoiced sounds, most frequencies range above 4000 Hz and it has approximately a flat spectrum. All voiced sounds are characterized by voice onset time (VOT). For example, plosives are characterized by VOT, which is

TABLE I

AVERAGES OF FUNDAMENTAL AND FORMANT FREQUENCIES AND FORMANT AMPLITUDES OF VOWELS BY 76 SPEAKERS

		i	I	ξ	æ	α	Э	μ		Λ	3
Fundamental frequencies (cps)	M	136	135	130	127	124	129	137	141	130	133
	W	235	232	223	210	212	216	232	231	221	218
	Ch	272	269	260	251	256	263	276	274	261	261
Formant frequencies (cps)											
F1	М	270	390	530	660	730	570	440	300	640	490
	W	310	430	610	860	850	590	470	370	750	500
	Ch	370	530	600	1010	1030	680	560	430	850	560
F ₂	М	2290	1990	1840	1720	1090	840	1020	870	1190	1350
an tao 📕 ang binang tao ang	W	2790	2480	2330	2050	1220	920	1160	950	1400	1640
	Ch	3200	2730	2610	2320	1370	1060	1410	1170	1590	1820
F ₃	М	3010	2550-	2480	2410	2440	2410	2240	2240	2390	1690
	W	3310	3070	2990	2850	2810	2710	2680	2670	2780	1960
	Ch	3730	3600	3570	3320	3170	3180	3310	3260	3360	2160
Formant amplitudes (db)	L	-4	-3	-2	-1	-1	0	-1	-3	-1	-5
	L_2	-24	-23	-17	-12	-5	-7	-12	-19	-10	-15
	L ₃	-28	-27	-24	-22	-28	-34	-34	-43	-27	-20

Source: Peterson and Barney, "Control Methods Used in a Study of the Vowels," <u>The Journal of the Acoustical</u> Society of America, Vol. 24, No. 2 (1952), 181. the delay from complete closure of the plosive to the beginning of voicing [66]. The VOT ranges from 25 milliseconds to 300 milliseconds depending on the phoneme.

Each phoneme has its own characterization depending on the language. This characterization is associated with place of articulation and voicing. In this thesis, discussed are the phonemes of the English language. This is not to discard the pitch inflections in Chinese, whispered vowels in Japanese or vocal clicks of South African Hottentots, but to restrict to a basic area to all languages. This is established by the International Phonetic Association (IPA). Most linguists use about 35 basic units, and six diphthongs or combination phonemes. The symbols and teletype representations of these are shown in Table II.

Phoneticians classify speech sounds by vowels and consonants, or strictly speaking in the manner and their place of production. Each phoneme has certain characteristics and is identified from the distinctive features of the speech sound. The distinctive features give a unique identification of the phoneme. These are given below [68].

1. Vocalic/Nonvocalic

presence vs. absence of a sharply defined formant structure.

Consonant/Nonconsonant
 low vs. high total energy.

3. Interrupted/Continuant

silence followed and/or preceded by spread of energy over a wide frequency region (either as a burst or a rapid transition of vowel formants) vs. absence of abrupt transition between sound and the silence.

4. Nasal/Oral

TAB	LE	ΙI

Standard IPA	Teletype Representation	Example
i	IY	b <u>ee</u> t
la di seconda di second	I A A A A A A A A A A A A A A A A A A A	b <u>i</u> t
е	EY	g <u>a</u> te
ξ	EH	<u>ge</u> t
æ	AE	f <u>a</u> t
a	AA	f <u>a</u> ther
0	ΑΟ	1 <u>a</u> wn
0	OW	1 <u>o</u> ne
U	UH	f <u>u</u>]]
u	UW	f <u>oo</u> 1
3,γ	ER	murder
α	AX	about
A	АН	but
aI	ΑΥ	hide
aU	AW	h <u>ow</u>
	ΟΥ	toy
p	Р	pack
b	В	back
t	T	time
d	D	dime
k	К	coat

REPRESENTATION OF IPA PHONEMES WITH EXAMPLES

Standard IPA	Teletype Representation	Example
g	G	goat
f	F	<u>f</u> ault
V	V	vault
τα. Τάλλα θ α το τρομογοριατικό το το το	ТН	ether
8	DH	ei <u>th</u> er
S	S	sue
Z	Z	<u>z</u> 00
ſ	SH	leash
Z	ZH	leisure
h	HH	how
n de la companya de l La companya de la comp	M	sum
n	N	sun
η	NX	sung
1	L L L L L L L L L L L L L L L L L L L	laugh
W	W	wear
j	Ŷ	young
r.	R	rate
t∫	СН	chan
d	JH	jar
hw	WH	where

TABLE II (Continued)

Source: Rabiner and Schafer, <u>Digital Processing of Speech Signals</u>, New Jersey: Prentice-Hall, 1978, p. 43.

į

spreading the available energy over wider vs. narrower frequency regions by a reduction in the intensity of certain (primarily the first) formants and introduction of additional (nasal) formants.

5. Tense/Lax

higher vs. lower total energy in conjunction with a greater vs. smaller spread of the energy in the spectrum and in time.

6. Compact/Diffuse

higher vs. lower concentration of energy in a relatively narrow, central region of the spectrum accompanied by an increase vs. a decrease of the total energy.

7. Grave/Acute

concentration of energy in the lower vs. upper frequencies of the spectrum.

8. Flat/Plain

flat phonemes in contra-distinction to the corresponding plain ones are characterized by a downward shift or weakening of some of their upper frequency components.

9. Strident/Mellow

higher intensity noise vs. lower intensity noise.

A table for the distinctive features of the phonemes of English are shown in Figure 5 [66]. As indicated above the features may be of two types. The presence or absence of each feature is expressed as a plus (+) or minus (-). For example, the vocalic category has vowels shown as plus and consonants are shown as minus.

	Distinctive Features	γ	1	I	ξ	æ	a	٨	. ว	U	u	j	r	w	1	m	n	η	5	S	f	e	7	z	۷	x	t∫	k	p	t	dz	g	ъ	8	h
1.	Vocalic/Nonvocalic	+	+	+	+	+	+	+	+	+	+	+	+	+	+	+	+	+	-	-	-	-	•	-	-	-	•	-	-	-	-	•	-	-	-
2.	Consonant/Nonconsonant	-	-	-	-	-	-	-	-	-	-	+	+	+	+	+	+	+	+	+	+	+	+	+	+	+	+	+	+	+	+	+	+	+	:
3.	Continuant/Interrupted											1							+	+	+	+	+	+	+	+	-	•-	-	-	-	-	-	-	
4.	Nasal/Oral					2						-	-	-	-	+	+	+		147 - 2- - -													•		
5.	Tense/Lax	+	+	-	-	+	+	-	+	-	+								+	+	+	+	-	-	-	-	+	+	+	+	-	-	-	-	,
6.	Compact/Diffuse	-	-	-	+	+	+	÷	+	-	-					-	-	+	+	-	- ,	-	+	-	-	-	+	+	-	-	+	+	-	-	•
7.	Grave/Acute	•	-	-	-	-	+	+	+	+	+	-	-	+	+	+	-		-	-	+	+		-	· +	+	-		-	-			+	-	
8.	Flat/Plain	+	-				-	÷	+			-	+	+	-		•		•																
9.	Strident/Mellow		× .																		+	-			+	-	-	-			+	-			

PHONEMES

Figure 5. Distinctive Features of the Phonemes of English Indicating the Presence or Absence of a Feature

မ္သ

2.3 Model of the Vocal Tract

The acoustic speech system was qualitatively described in the previous section. The acoustic tube model of the vocal tract filter can be represented as a discrete time-invariant linear filter. The modeling has been discussed in the literature [2] [7] [67]. The acoustic tube is approximated by a number of sections each having a constant cross-sectional area. The cross-sectional area is characterized by the reflection coefficients. The reflection coefficient is the percentage of a wave reflected at an acoustic tube junction. The number of sections in the acoustic tube model is related to the number of formants for a phoneme.

The formants of speech correspond to the poles of the vocal tract transfer function [67]. As pointed out in the last section, only the first three or four formants are used for speech analysis, and these frequencies are below 5000 Hz. Generally, vocal tract resonances occur about one per thousand Hertz [67]. Therefore, a bandwidth of 5 kHz is, in general, sufficient for speech analysis and synthesis. Each phoneme is set apart from the others by the frequency location of the formants.

The majority of phonemes can be represented by an all-pole model of the vocal tract [5]. It is well known that for nonnasal voiced phonemes the transfer function of the vocal tract has no zeros [69]. Nasal and glide sounds include zeros in the transfer function. Zeros and poles are necessary to approximate the nasal and glide sounds. However, it has been shown that zeros in the vocal tract can be achieved by including more poles [5].

In Figure 1, let the transfer function of the vocal tract be expressed by [7]

$$V(z) = \frac{G}{\substack{p \\ 1 - \sum_{i=1}^{r} a_i z^{-i}}}$$
(2.1)

where G, the gain; {a_i}, the filter coefficients, are a function of the cross-sectional areas of the acoustic tube. The value of P, the order of the system, is usually taken as twice the number of formants for analysis for each speech sound. Typical values for P range from 8 to 10. The value of 10 has been used for lattice network representations of the vocal tract.

It has been shown that given (2.1), a lossless tube model can be found [5] [7]. Also, given an acoustic tube with all areas positive, Equation (2.1) describes a stable system [7].

2.4 A Parallel Between Glottal Waveform

and the Residual Signal

In modern signal processing techniques, it is necessary to use as much information as can be obtained about the structure of the signal. This section discusses the characteristics of the residual signal, which is the output of the linear prediction filter. It is the difference between the actual and predicted speech signals.

The residual signal used in this thesis is obtained by using the autocorrelation method in the LPC algorithm. In doing this, the speech is Hamming windowed, where the window function is

w(n) = 0.54 - 0.46 cos
$$\left[\frac{2\pi n}{N-1}\right]$$
 $0 \le n \le N-1$

otherwise (2.2)

with N = 256. The computational details are discussed in Section 2.6.

= 0

The prediction residual is the ideal signal for the excitation function for LPC analysis [28]. It contains the actual information, rather than a pulse train or random noise as in the simplified linear prediction models [10]. The waveform that excites the vocal tract is the glottal waveform, and the residual approximates this.

The characteristics of the prediction residual are as follows: (1) it marks the pitch period, (2) it has basically a flat amplitude spectrum, (3) phasing information is embedded in the prediction residual, (4) the amplitude spectrum includes details related to the suprasegmentals of the individual and the spoken words, (5) the waveform includes the fact that voiced fricatives and stops are a combination of noise and a repetitive signal.

Figure 6 gives a comparison between a speech wave and the corresponding prediction residual for a particular phoneme. The computational aspects in obtaining these figures will be discussed later. The pitch period is marked by large spikes in the residual signal. The residual gives an excellent estimation of pitch since the glottal excitation is clearly marked.

Figure 7 displays an unsmoothed spectrum of the residual signal. The spectrum of the residual contains the formants also. The peaks of the formants are flattened; however, there is evidence of the fundamental and formant frequencies on the plot. The dashed line represents a smooth spectrum. Even in this, it is seen that there is evidence of the fundamental and the formants.

The pitch and voicing for each human is unique. It can be shown by spectrograms that individuals have unique voice prints. This uniqueness is basic to the excitation signal rather than in the vocal tract filter.









Therefore, the suprasegmentals, i.e., the intonation, dialect, melody pattern, etc., will remain unique to individuals for voiced sounds.

The voiced fricative lends more benefit to this discussion than its cognate, the unvoiced fricative. The unvoiced fricative is simply a noisy speech waveform that produces only a noisy residual signal. The fricative or stop is produced by forcing air through a constriction, such as the teeth or lips. The corresponding sound results from the turbulence and is of the noisy type. The waveform is then represented by noise that can be shown to be an unvoiced excitation source. However, the voiced fricative is a result of a constriction in the vocal tract while the vocal cords are vibrating. The residual signal from these phonemes produce a repetitive signal at the pitch period.

The artificial excitation function for voiced sounds result in speech that sounds a bit unnatural. The use of the prediction residual in coding methods would introduce naturalness in voicing. Ideally, the excitation of the vocal tract filter model should approximate the excitation of the human vocal tract. The prediction residual meets these requirements.

2.5 Review of Linear Prediction Analysis

Linear prediction analysis uses a weighted sum of P successive speech samples to predict the next speech sample. The weights are chosen such that the mean-square prediction error is minimized. Let

$$x_{n} \approx a_{1} x_{n-1} + a_{2} x_{n-2} + \cdots + a_{p} x_{n-p}$$

 $x_{n} \approx \sum_{i=1}^{p} a_{i} x_{n-i}$ (2.3)

where x_n represents the speech sample sequence and a_i is a set of predictive coefficients. In this application, the method of least squares is used. Assuming a stationary linear system [5] with time-invariant statistics, zero mean, let $\hat{x}_f(n)$ represent the best estimate, in the least mean-square sense, of x_n using the a_i , $i = 1, \ldots, P$ coefficients and let $\hat{x}_b(n)$ be the best backward prediction of x_n using the b_i , i =1, ..., P coefficients. Then

$$\hat{x}_{f}(n) = \sum_{\substack{j=1 \\ i=1}}^{p} a_{i} x_{n-i}$$
(2.4a)

$$\hat{x}_{b}(n-P-1) = \sum_{i=1}^{p} b_{i} x_{n-i}$$
 (2.4b)

Let $e_f(n)$ and $e_b(n)$ be the forward and backward prediction errors defined by

$$e_{f}(n) = x_{n} - \hat{x}_{f}(n)$$

= $-\sum_{i=0}^{P} a_{i} x_{n-i}$ (2.5a)
 $e_{b}(n-P-1) = x_{n-P-1} - \hat{x}_{b}(n-P-1)$

$$= - \sum_{i=1}^{P+1} b_i x_{n-i}$$
 (2.5b)

where it is assumed that $a_0 = -1$ and $b_{P+1} = -1$. Figure 8 gives the implementation of (2.5)

Since stationarity is assumed, it follows that the errors can be minimized by

$$E\left[\frac{\partial}{\partial a_{j}}\left(e_{f}(n)\right)^{2}\right] = 0 \qquad j = 1, \dots, P \qquad (2.6a)$$



Figure 8. Implementation for Generation of Forward and Backward Prediction Errors

$$E\left[\frac{\partial}{\partial b_{j}}(e_{b}(n-P-1))^{2}\right] = 0 \quad j = 1, ..., P$$
 (2.6b)

These reduce to

$$\sum_{i=0}^{P} b_{i} E[x_{n-i} x_{n-j}] = E[x_{n} x_{n-j}] \qquad j = 1, ..., P \quad (2.7)$$

$$\sum_{i=0}^{P} b_{i} E[x_{n-i} x_{n-j}] = E[x_{n-P-1} x_{n-j}] \qquad j = 1, ..., P \quad (2.8)$$

By defining

$$E[x_{n-i} x_{n-j}] = R_{i-j}$$
(2.9)

Equations (2.7) and (2.8) can be expressed by

$$P_{\substack{\Sigma \ i=1}} R_{i-j} a_i = R_{j} j = 1, ..., P (2.10a)$$

$$P_{\substack{\Sigma \\ i=1}} R_{i-j} b_{i} = R_{P+1-j} \quad j = 1, ..., P \quad (2.10b)$$

where $R_{i-j} = R_{j-i}$ has been used. It is clear that (2.10a) and similarly (2.10b) can be written in a matrix form, wherein the coefficient matrix is a symmetric Toeplitz matrix [71]. Furthermore,

$$b_i = a_{P+1-i}$$
 $i = 1, ..., P$ (2.11)

which can be seen by defining

j = P+1-ki = P+1-k

in (2.10a). That is,

$$\sum_{k=P}^{\Sigma} R_{k-\ell} a_{P+1-k} = R_{P+1-\ell}$$

which can be rewritten as

$$\sum_{i=1}^{p} R_{i-j} a_{P+1-i} = R_{P+1-j} j = 1, ..., P$$
 (2.12)

Comparing (2.12) with (2.10), the relation in (2.11) can be seen The forward prediction error

$$E[e_{f}^{2}(n)] = E[(x_{n} - \sum_{i=1}^{p} a_{i} x_{n-i})(x_{n} - \sum_{j=1}^{p} a_{i} x_{n-i})]$$

$$= E[x_{n}^{2} - 2\sum_{i=1}^{p} a_{i} x_{n} x_{n-i}] + E[\sum_{j=1}^{p} a_{j} \sum_{i=1}^{p} a_{i} x_{n-i} x_{n-j}]$$

$$= E[x_{n}^{2}] - E[\sum_{i=1}^{p} a_{i} x_{n} x_{n-i}]$$

$$= R_{0} - \sum_{i=1}^{p} a_{i} R_{i} \equiv E_{p}$$
(2.13)

where (2.7a) has been used to obtain (2.13).

The cross-correlation between the forward and backward prediction errors is derived in the following. Let

$$C_{p+1} = E[e_{f}(n) e_{b}(n-P-1)]$$

= $E[x(n) x(n-P-1)] - E[\sum_{j=1}^{P} a_{p+1-j} x_{n}x_{n-j}]$
- $E[\sum_{j=1}^{P} a_{i} x_{n-i} x_{n-P-1}] + E[\sum_{j=1}^{P} a_{p+1-j} \sum_{i=1}^{P} a_{i} x_{n-i} x_{n-j}]$

$$= R_{P+1} - \sum_{i=1}^{P} a_i R_{P+1-i}$$
(2.14)

where again (2.7a) has been used to obtain (2.14).

It is clear that (2.13) and (2.14) correspond to P coefficients. In the following, a recursive method will be used wherein the coefficients a_i will be updated. For this reason, let

$$E_0 = R_0$$
 (2.15)

$$E_k = R_0 - \sum_{i=1}^k a_i^{(k)} R_i$$
 $k = 1, ..., P$ (2.16)

$$C_{k+1} = R_{k+1} - \sum_{i=1}^{k} a_i^{(k)} R_{k+1-i}$$

 $k = 0, 1, ..., P-1$ (2.17)

where $a_i^{(k)}$ are determined from (2.10a) by using

$$\sum_{i=1}^{k} R_{i-j} a_{i}^{(k)} = R_{j} \quad j = 1, \dots, k \quad (2.18)$$

Durbin's method [72] [73] can now be used to solve for $a_i^{(k)}$ in (2.18). The corresponding equations are

$$E_{o} = R_{o}$$
(2.19)

$$k_{j+1} = \frac{C_{j+1}}{E_{j}}$$

$$a_{j+1}^{(j+1)} = k_{j+1}$$

$$a_{i}^{(j+1)} = a_{i}^{(j)} - k_{j+1} a_{j+1-i}^{(j)}$$

$$i = 1, 2, ..., j \quad (2.20)$$

$$E_{j+1} = E_{j} (1 - k_{j+1}^{2})$$

The predictive coefficients are obtained from

$$a_i = a_i^{(j+1)}$$
 $i = 1, 2, ..., P$

Interestingly, the prediction residual E_j in (2.20) is readily available in the algorithm for the predictor of order j. The coefficients k_j generated in (2.20) are usually referred as PARCOR coefficients. These have some interesting characteristics [9] [28].

- 1. $|k_j| \leq 1$
- 2. Since $|k_j|$ is unity bounded, a set quantization levels can be determined.
- 3. The PARCOR coefficients are the result of the orthogonalization of the auto-correcation matrix.

In order to show the application of this system, the transfer function and the algorithm to acquire the prediction residual is derived below.

The transforms of $e_{f}(n)$ and $e_{b}(n-P-1)$ in (2.5) can be expressed in terms of

$$E_{f}(z) = -\sum_{i=0}^{P} a_{i} z^{-i} X(z)$$
 (2.21a)

$$z^{-(P+1)} E_{b}(z) = -\sum_{i=1}^{P+1} b_{i} z^{-i} X(z)$$
 (2.21b)

where $E_{f}(z)$, $E_{b}(z)$ and X(z) are the transforms of $e_{f}(n)$, $e_{b}(n)$ and x(n), respectively. Note that a_{0} and b_{P+1} in (2.21) are each equal to -1. For simplicity, let

$$A_{p}(z) = 1 - \sum_{i=1}^{p} a_{i} z^{-i}$$
 (2.22a)

$$B_{p}(z) = z^{-(P+1)} - \sum_{i=1}^{P} b_{i} z^{-i}$$
 (2.22b)

With these, (2.21) can be written as

$$E_{f}(z) = A_{p}(z) X(z)$$
 (2.23a)

$$E_b(z) = z^{P+1} B_p(z) X(z)$$
 (2.23b)

It is clear that (2.22) was implemented in Figure 8 using the direct form. Next, the lattice network implementation of (2.22) is discussed below. In order to do this, recall the relation $b_i = a_{P+1-i}$ given in (2.11). With the relation, (2.21b) can be written as

$$B_{p}(z) = z^{-(P+1)} - \sum_{i=1}^{P} a_{P+1-i} z^{-i}$$

$$= z^{-(P+1)} - \sum_{j=1}^{P} a_{j} z^{-(P+1)+j} \qquad (2.24)$$

$$= z^{-(P+1)} A_{p}(z^{-1}) \qquad (2.25)$$

From this it follows that

$$A_p(z) = z^{-(P+1)} B_p(z^{-1})$$
 (2.26)

Equations (2.20), (2.25) and (2.26) will now be used to derive the lattice implementation. To develop the recursive equation for the lattice formulation, some of the above equations have to be written in a recursive manner. It is clear that (2.22) can be rewritten in the form

$$A_{j+1}(z) = -\sum_{\substack{\Sigma \\ i=0}}^{j+1} a_i^{(j+1)} z^{-i}$$
 j = 0, 1, ..., P-1 (2.27a)

$$B_{j+1}(z) = -\sum_{i=1}^{(j+2)} b_i^{(j+1)} z^{-i} \qquad j = 0, 1, ..., P-1 \quad (2.27b)$$

where the superscripts on a_i and b_i are included to denote that (j+1)th order is implemented rather than a Pth order. Also

$$a_0^{(j+1)} = -1$$
 (2.28)

$$b_{j+2}^{(j+1)} = -1$$
 (2.29)

have been used. The remaining $a_i^{(j+1)}$ can be expressed in terms of $a_i^{(j)}$ using (2.20); $b_i^{(j+1)}$ are related to $a_i^{(j+1)}$ by [see (2.4)]

$$b_{i}^{(j+1)} = a_{j+1-i}^{(j+1)}$$
 (2.30)

Using (2.20), (2.29) and (2.30) in (2.27a)

$$A_{j+1}(z) = 1 - \sum_{i=1}^{j+1} (a_i^{(j)} - k_{j+1} a_{j+1-i}^{(j)}) z^{-i}$$

= $A_j(z) - k_{j+1} \sum_{i=1}^{j+1} b_i^{(j)} z^{-i}$
= $A_j(z) - k_{j+1} B_j(z)$ (2.31)

Using (2.24)

$$B_{j+1}(z) = z^{-(j+2)} A_{j+1}(z^{-1})$$
 (2.32)

Equation (2.31) can be rewritten as

$$A_{j+1}(z^{-1}) = A_j(z^{-1}) - k_{j+1} B_j(z^{-1})$$
 (2.33)

Substituting (2.32) in (2.33) and simplifying

$$B_{j+1}(z) = z^{-1}[B_j(z) - k_{j+1} A_j(z)]$$
 (2.34)

Equations (2.31) and (2.34) define the algorithm. The implementation of these is shown in Figure 9, where the generation of k_{j+1} is also included. The detailed structure of the optimum inverse filter as an analysis model is shown in Figure 10a. The corresponding synthesis model is shown in Figure 10b. The output of the synthesis filter is the input speech signal. From the analysis section, transform of the prediction residual is $A_p(z)$.

2.6 Short-Time Analysis

The concept of short-time Fourier analysis [76] [77] is fundamental for coding the residual signal. For a quasi-periodic signal such as speech, the short-time or time-dependent Fourier analysis allows for a detailed study.

The speech signal, x(m), m = 0, 1, ..., L-1, from Equation (2.3) is segmented into r sections such that short-time spectral analysis can be used. It is assumed that L = rN, where N corresponds to the number of samples in each section. This assumes the use of the formula

$$\Sigma = w(nD-m) = 1$$
 (2.35)

where w(m) corresponds to a band limited function to a frequency of 1/2D, and D is the period (in samples) between adjacent samples of the shorttime transform of the signal [77]. In all practical cases, w(m) is a time limited signal and, therefore, its spectrum cannot be band limited. The effects of this non-band limited case are discussed in a recent paper [93]. It has been shown that the aliasing errors are small and can be



Figure 9. Detailed Structure of PARCOR Implementation for jth Stage



a.) Analysis Model



b.) Synthesis Model

Figure 10. Analysis and Synthesis Models for Lattice Structure

neglected if D is properly chosen. For a Hamming window, D = (N/4) [77]. In addition to the aliasing errors, end effect errors have to be considered also [94]. This is necessary since L if finite. The LPC analysis is applied to the windiwed signal resulting in the windowed residual signal. The overlap-add of this signal is the residual signal, which has error identified earlier.

The short-time Fourier transform of the residual signal $e_{f}(m)$ can be defined as [67]

$$X_{n}(e^{j\omega_{k}}) = \sum_{m=-\infty}^{\infty} [e_{f}(m) w(nD-m)] e^{-j\omega_{k}m}$$
(2.36)

where $\omega_k = (2 \text{ k/N})$, k = 0, 1, ..., N-1, and w(m) corresponds to a window. For a particular value of n, Equation (2.36) can be implemented using FFT. This is used in this thesis. A brief review of this is presented below.

Let

$$e_n(m) \approx e_f(nD+m) w(-m) -\infty \le m \le \infty$$
 (2.37)

Using this in (2.36),

Further, let m = Nr+q, $-\infty \le r \le \infty$, $0 \le q \le N-1$. With these,

$$X_{n}(e^{j\omega_{k}}) = \sum_{\substack{\Sigma \\ r=-\infty}}^{\infty} \left[\sum_{q=0}^{N-1} e_{n}(Nr+q) e^{-j\omega_{k}(Nr+q)} - j\omega_{k}n \right] e^{(2.39)}$$

Noting that $e^{-j\omega_k Nr} = 1$,

$$X_{n}(e^{j\omega_{k}}) = \sum_{\substack{\alpha = 0 \\ q=0}}^{N-1} \sum_{\substack{\alpha = -\infty \\ r=-\infty}}^{\infty} e_{n}(Nr+q) = \sum_{\substack{\alpha = -j\omega_{k} \\ q=0}}^{-j\omega_{k}q} e^{-j\omega_{k}n}$$
(2.40)

For simplicity, let

$$u_{n}(q) = \sum_{r=-\infty}^{\infty} e_{n}(Nr+q) \qquad 0 \le q \le N-1 \qquad (2.41)$$

Note that $u_n(q)$ is periodic with period N. Now Equation (2.40) can be written, and is

$$X_{n}(e^{j\omega}k) = e^{-j\omega}k^{n} \begin{bmatrix} N-1 & -j\omega k^{q} \\ \Sigma & u_{n}(q) e^{-j\omega}k^{q} \end{bmatrix}$$
(2.42)

Observe that $X_n(e^{j\omega}k)$ is represented as $e^{-j\omega}k^n$ times the DFT of the sequence $u_n(q)$. Therefore, (2.42) can be written as

$$e^{j\omega_{k}n} X_{n}(e^{k}) = \sum_{\substack{\alpha=0 \\ q=0}}^{N-1} u_{n}((m-nD))_{N} e^{-j\omega_{k}q}$$
(2.43)

Equation (2.42) represents the DFT form, where $((\cdot))_N$ corresponds to the modulo N.

The following procedure can be used to compute (2.43).

1. The windowed sequence, $e_n(m)$, can be computed from (2.37). The sequence can then be divided into r sections of N samples each, where in this thesis, L = 4096, N = 256, D = 64, and r = 16.

2. The N-point DFT of $u_n((m-nD))_N$ can be computed to obtain (2.43) using FFT.

The above procedure is given here for generality. Due to the limitation of the disc space and to reduce computational time, a slightly different procedure is used in computing the spectral analysis. The residual signal is rectangular windowed to 256 points, spectrum analyzed and then averaged. This is used only for phonemes discussed in the thesis. No overlapping was used. The errors associated with this method are quantified in previously mentioned references [93] [94].

2.7 Implementation of Operations for the Calculation of the Prediction Residual

In this section, the formulation of the prediction residual from the speech input is presented. The implementation of the operations to calculate the prediction residual represents the analysis model for LPC. The analysis model consists of the speech as the input, the vocal tract model, the correlation coefficients and the residual as the output.

The analog speech signal is band limited to 3600 Hertz using a second order Butterworth filter. This signal is digitized at the rate of 8000 samples/second. The algorithm for digitization is named DIGITIZ and the computer program is included in Appendix B.

The results in the last section are used to obtain the windowed digitized data. This allows to process the speech in short segments. The underlying assumption for most speech processing schemes is that the properties of the speech signal change relatively slowly with time [67]. This assumption leads to short-time methods which isolate the signal during the segment of windowing. The window is a 256-point Hamming window and is overlapped at 64-point intervals. The windowing is computed by program WINDOW in Appendix B.

The windowed signal is passed to program AUTO [7]. This program uses the autocorrelation method for solving the matrix equation (2.10) for the predictor coefficients [61]. The other matrix values solved for are the reflection coefficients or PARCOR coefficients. These values are passed for use in the lattice formulation.

The lattice method represents a recursive algorithm for a solution of the prediction residual. This method guarantees stability. Note that the PARCOR coefficients are bounded. The program to calculate the residual by the lattice formulation is INVERS and is included in Appendix B.

Figure 11 illustrates a block diagram showing the sequence of operations related to the calculation of the prediction residual, $e_f(n)$.

2.8 A Novel Approach to Pitch Extraction

2.8.1 Types of Problems Associated with Pitch Extraction

The pitch extractor is of prime importance in most speech processing systems, as the pitch is one of the basic parameters in speech analysis and synthesis studies. In low-bit rate systems, it is an essential component [2] [7]. Speech with a constant fundamental frequency is perceived as a monotone or of a synthetic nature; variable pitch lends to speech a melody. An accurate pitch extractor is a challenging area of speech processing.

The difficulty in accurately determining pitch is due primarily to the time varying aspects of the glottal excitation. Since the model of the vocal tract assumes quasi-periodic changes occurring along the acoustic tube, the glottal response is not predicted accurately. This innaccuracy is due to the nonuniform train of periodic pulses that occur with the golttal waveform. The simple model of the vocal system excitation, i.e., periodic uniform pulses or Gaussian noise, eases the measurement of





of the period of the pitch. However, when the pitch and the waveform are changing within a period which occurs with frequency shifts, difficulty arises.

The second problem associated with the measurement of pitch is due to the nonseparability of the vocal tract model from the glottal excitation. That is, the separation of the formants and the fundamental frequency may not be possible and therefore the detection of the pitch period is difficult. This interaction can be seen most often during transitional regions of formants when the articulatory elements are changing.

The third problem is the detection of the beginning and ending of the pitch period. Part of this problem occurs in the definition of beginning and ending of the pitch period. In examining the speech waveform, it is necessary to always be consistent with the method because different definitions will often lead to different results. This is seen in Figure 12. In Figure 12, one can detect the period of zero crossings before the maximum peaks or detect the period between the maximums. However, the two methods do not always give the same answers. The discrepancy between the two is due to the slowly time-varying properties of glottal excitation.

The fourth problem that arises is the decision to ascertain which segment of speech is voiced or unvoiced. In particular, some algorithms have problems distinguishing between low-level speech and unvoiced speech. In transitional analysis, it is difficult to pinpoint the difference between the two.

In addition to the above problems, the pitch detection is hindered further when the signal is a transmitted speech signal. During the





transmission of a speech signal over a telephone line, there are degradations that occur that can change the signal to make pitch detection difficult. These include: 1) phase distortion, 2) amplitude modulation of the signal, 3) crosstalk between messages, 4) clipping of high-level sounds. Furthermore, as the signal travels through the telephone lines, the lines act as a bandpass filter with approximate band edges $f_1 = 200$ Hertz and $f_2 = 3200$ Hertz. The fundamental frequency is usually less than 200 Hertz and therefore is removed by the bandpass action of the line. The pitch must be regenerated by using harmonics.

The next section discusses advantages and disadvantages associated with the use of the prediction residual for pitch extraction.

2.8.2 Advantages and Disadvantages for Using

the Prediction Residual as a Source for Pitch

Extraction

The prediction residual solves the problem of vocal tract excitation. Earlier, it is stated that there is inaccuracy in determining glottal response when using the simple model for excitation. When using the twosource model for the vocal system excitation, i.e., quasi-periodic pulses and random noise, a simple algorithm can be used for extraction of pitch. The residual can be used as a single source as an approximation to the glottal excitation, and, therefore, a simple method can be used to employ the residual to extract pitch.

It is well known that the residual represents the deconvolution of the speech from the vocal tract [7]. For each vocal tract configuration, a different set of formants and a variation in harmonics of the fundamental frequency in the spectrum is acquired. The pitch markings are determined by residual spikes in the time-domain. This can be used to extract the pitch accurately.

The advantage of an accurate estimation of pitch will aid to the perception of speech. Any enhancement to perception is important to any analysis-synthesis speech system. The discussion which follows includes other reasons for using the prediction residual as a source for pitch extraction.

Referring to Figure 12, it is shown where errors can occur when the speech signal is used for pitch extraction. Figure 13 shows the residual, over 256 samples, characterized by spikes which represent the pitch period. It can be seen that it is not necessary to account for the zero crossings or maximums. It is simply a matter of tracking absolute maximums within the range of the established pitch period. It has been shown that if the interval of analysis is small enough the residual can be used to extract pitch accurately [28]. Future transmission rates will require a system that can do an acceptable performance for extracting pitch.

An application for using the residual signal for pitch extraction is with embedded coding. The advantage with the residual signal is that an absolute pitch can be determined in a frame. At higher transmission rates, the coding of the residual can be accomplished more efficiently. Therefore, a pitch extraction method can be employed easily. However, it is not feasible to transmit the residual with low transmission rates; consequently, the higher rates must extract the pitch and transfer this to the lower rates. Since the residual demonstrates a very accurate representation of the pitch, the frame-by-frame analysis of the pitch from the prediction residual would enhance pitch in an embedded coding scheme.


Figure 13. Prediction Residual Waveform for Phoneme /æ/ over 256 Sample Segment

However, a disadvantage associated with the residual may occur in a high noise environment. It has been shown earlier that the residual is a **com**bination of periodic and noisy signals. In a high noise environment, the noise may overcome the residual signal. If the noise has amplitude in the range of pitch markings, the signal would require enhancement to extract pitch adequately. On the other hand, low noise contributes to the flatness of the spectrum of the signal and enhances pitch extraction.

Several advantages and disadvantages have been discussed. It can be readily seen that the residual is an ideal signal for extraction of the pitch. The next section discusses the implementation of the pitch extractor.

2.8.3 A Novel Pitch Extractor

The last few sections have described the prediction residual as the result from the linear prediction analysis. It has been shown that the prediction residual contains much information needed for extracting pitch. It is a simple problem to pick appropriate peaks to extract the pitch. It is this problem of pitch extraction that has interested many authors recently.

Examining Figure 13, a repetitive waveform is seen at the period called the pitch period. Note that the waveform has a noisiness which implies a flat amplitude spectrum. It should be noted that for voiced sounds there are other peaks that are also repetitive. These are evidence of the formant frequencies. They are somewhat dampened; however, this is to be expected since the linear predictive filter has the characteristic of spectrally flattening the signal.

It is well known for

$$x(t) = A \cos (2\pi f_0)$$
 (2.44)

where A is the constant maximum amplitude of the signal, and f_0 is the fundamental frequency, the spectrum is

$$X(f) = \frac{A}{2} \delta(f-f_0) + \frac{A}{2} \delta(f+f_0)$$
 (2.45)

It can be said that the speech waveform is a combination of sinusoids of the type given in (2.44) summed together in a quasi-periodic fashion. The residual signal, $e_f(n)$, can be described in a similar fashion. Therefore, it follows that the spectrum of $e_f(n)$ has impulses at the fundamental and its harmonics identified here by f_0 , f_1 , ..., f_n . The maximum amplitude is centered at f_0 , the fundamental frequency [75]. The higher frequencies are all harmonics, or multiples of f_0 . An <u>a priori</u> estimate of f_0 for a speech sound can be found using the residual as input. If the spectrum is available, then the frequency of the maximum amplitude determines an estimate of f_0 . This estimate is found to be relatively accurate for speech and the prediction residual. In the following, a procedure for extracting the fundamental is given.

The initial step is square the residual. This has a dual benefit in addition to making all calculations positive. First, it makes large quantities larger and second, any small or noise-like quantities are made smaller. The new data corresponding to the set of squared samples are placed in frames of 256 samples each.

Following the initial step, the original sample rate is used to determine the time difference between maximums. It is assumed that the maximums mark the beginning of a new pitch period as set by a threshold. The threshold is used to select the next maximum. The data set is passed through a peak picker. The peak picker uses the threshold to determine the next peak (maximum). The time between the two peaks is calculated by a differencer function. The system is ready to set a pitch value from the time between the two peaks.

At this point, the <u>a priori</u> estimate of the pitch and the pitch value from the differences are averaged. An error check is made for erroneous pitch values. The error check compares a range of pitch from a low to a high value. Should the averaged value be less or more than a set threshold, an update is sent to recalculate the last averaged pitch in the frame. The process is continued until the end of the frame where the pitch is set. The procedure for estimating pitch is shown in Figure 14. The next section discusses the results in using the pitch extractor.

2.8.4 Pitch Extraction Results

The PITCH program was applied to 39 phonemes, including 16 vowels and diphthongs and 23 consonants. Each sound was held from one-quarter second to one second by a male speaker at normal intensity. Recordings were made on a SONY Model TC-106A tape recorder under anechoic conditions. The sounds were low-pass filtered by a Butterworth filter with a cutoff frequency of 3600 Hertz and samples at 8000 Hertz with nine quantization bits and one sign bit. The computer system quantization level setting was ±10 volts. This gave a quantization level of 20 millivolts. The digitized sound was sampled for 1.5 seconds using 12000 data points for storage. With a limited computer system memory, the beginning of the sound was found and 4096 points were saved. The sound was stored for later use and labeled with an appropriate name. Due to the processing

PITCH ESTIMATION



Figure 14. Block Diagram of Pitch Extraction Method

of the INTERDATA 70 computer system, all processing in program PITCH is done in 256-point blocks.

The unified recursive solution to solve Equation (2.10a) is by program AUTO [7]. The efficient resursive solution was discussed in an earlier section. The program INVERS processes the data. The residual data is stored for use in the program FFTMGR. The computation of the spectrum is performed in this program. Spectral values are stored for use in PITCH.

Samples can be plotted for any segment of the sound to aid in visual determination of the pitch period. An example is shown in Figure 13. The results show that the method presented here is an adequate and accurate method for determining pitch period. It is compared to Peterson and Barney's data [111]. Table III gives a comparison between this data and the results obtained from this method. From this table, it can be seen that the results are good. The voiced/unvoiced decision is not a product of PITCH. The FFTMGR routine produces an energy level for the determination of voicing. Voicing errors were made 25 percent of the time. This is due to the fact that the threshold is set to a low level. However, the error check will restrict any wide variance of pitch. If the calculated fundamental frequency is larger than a set threshold value of 400 Hertz, then the corresponding sound is considered as an unvoiced sound and no further calculations are made. These two checks allow for accurate measure of pitch and voiced/unvoiced decision.

Error-free pitch estimation is critical to the overall performance of any low-bit rate coding system. Coding systems that incorporate the residual signal for estimation of the pitch are accurate and adequate. Accuracy can be enhanced by using the residual in a minimum noise

environment. The residual signal formulates a true pitch per frame that can be used at low, synthetic transmission rates. There are several reasons that have been discussed to show that this is a novel approach to pitch extraction and is summarized below. First, it is a two-stage method that estimates the residual spectrum and uses time samples of the residual to calculate the approximation of the pitch. Second, the calculation is done by a thresholding technique which uses the square of samples. Finally, the extraction of the pitch includes an error check that estimates wide variances of the pitch during each frame. From these, it can be seen that this method can be considered as a hybrid technique.

TABLE III

a yr		Fundamenta]	Frequency (Male) from
	Phoneme	from Peterson-Barney [111]	Proposed Method of Pitch Extraction
i set	,.,	100	100
	/1/	136	129
	/1/	135	130
	/ε/	130	125
	/œ/	127	135
	/α/	124	123
	$ \supset $	129	135
	/μ/	137	126
	/U/	141	151
	/Λ/	130	123
	181	133	140

COMPARISON OF FUNDAMENTAL FREQUENCIES

2.9 Summary

In this chapter, the characteristics of the prediction residual were presented. There is a parallel between the glottal waveform and the residual. The mechanism of speech production and a model of the vocal tract is discussed. Short-time spectral analysis is presented. A review of linear prediction analysis is discussed. A description of the implementation of operations for the calculation of the prediction residual is discussed. A novel approach to pitch extraction is presented.

CHAPTER III

SUB-BAND CODING OF THE PREDICTION RESIDUAL

3.1 Introduction

The average rate that speech is conveyed between humans is about ten phonemes per second. It has been shown that the information rate of speech does not exceed 60 bits/second [2] [67]. For the information content to be preserved, the human must be able to extract the representation of the speech signal at this rate. It is important that the speech is intelligible to the listener, and this aspect is the fundamental consideration of coding speech.

There are two concerns in coding speech signals. First, the message content of the speech must be preserved. The content includes linguistic rules to form thoughts for humans to communicate. Second, the speech signal should be represented so that it can be transmitted. At the receiver, the signal should contain the message without serious degradations.

The interest in speech coding has led researchers to consider techniques that enhance signal quality, reduce transmission rate and cost, without considering the complexity of the coding algorithm [64]. The principle is to enhance the perceptual aspects of speech through the coding method. In this chapter, some basic ideas associated with speech coding are discussed. These include transform coding and sub-band-coding.

Since the speech sounds are characteristically different than most acoustic sounds, it is necessary to consider the properties that include the formants and energy of phonemes. Perceptual aspects that contribute transitional cues for humans to discriminate and differentiate speech sounds are discussed in this chapter. It is known that when human listeners are exposed to speech, available to them are a set of responses that are highly over-learned [65]. The minimum discrimination necessary for absolute differentiation of speech sounds is discussed. Recently, speech coding techniques have contributed efficient methods to enhance the coding of speech signals with few degradations. This chapter discusses some of these methods in Section 3.2. Section 3.3 presents a discussion of the transform coding. Section 3.4 presents the method of sub-band coding in detail. Section 3.5 discusses the determination of frequency bands by the Articulation Index. Section 3.6 presents aspects associated with transitional cueing information for the preception of certain pho-Section 3.7 discusses perception of intelligible speech. Section nemes. 3.8 describes the basis for coding the prediction residual at the rate of 9600 bits/second.

3.2 Coding Methods

The oldest form of speech coding device is the channel vocoder invented by Dudley [78]. Each of the channels has center frequency ω_k . For each of the channels, the time-dependent Fourier transform is represented as a cosine wave with center frequency ω_k which is phase and amplitude modulated corresponding to the magnitude and phase angle, respectively of each transform. Therefore, each channel is thought of as a bandpass

filter with center frequency $\omega_{\mbox{$k$}}$ and impulse response w(n). This is shown in Figure 15.

The analysis section consists of a bank of channels as in Figure 15 with frequencies distributed across the speech band. Figure 16 shows a complete channel vocoder analyzer.

The basic diagram for the synthesizer is somewhat different. The specific channel controls the amplitude of its contribution to a particular channel; while the excitation signals control the detailed structure of the output of a given channel. The voiced/unvoiced decision serves to select the appropriate excitation generator, i.e., random noise for unvoiced speech and pulse generator for voiced speech. A block diagram is shown for the synthesizer in Figure 17. Channel vocoders operate in the range of 1200 bits/second to 9600 bits/second. They are also referred to as source coders and produce speech of a synthetic nature when at bit rates below 4800 bits/second.

A major contribution of a channel vocoder is the reduction in bit rate; however, direct representation of the pitch and voicing information is not achieved. Therefore, this can be considered as a weakness.

The LPC vocoder is a very important application of linear predictive analysis in the area of low bit rate encoding of speech. It is shown in Figure 18.

The basic LPC analysis parameters consists of a set of P predictor coefficients, the pitch period, a voiced/unvoiced parameter and a gain parameter. The vocoder consists of a transmitter which performs the LPC analysis and pitch detection. These parameters are coded and transmitted. They are decoded and synthesized to output speech. This category of

CHANNEL VOCODER COMPONENT





CHANNEL VOCODER ANALYZER



Figure 16. Block Diagram of Channel Vocoder Analyzer

CHANNEL VOCODER SYNTHESIZER





73

Ø

LPC VOCODER



Figure 18. LPC Vocoder

. .

<u>.</u> .

coding is of the vocoder type. In the following, the discussion of various aspects is included.

Speech coding can be divided into two broad categories, waveform coders and vocoders [79]. There has been some metion of a few types of vocoders earlier. Waveform coders generally attempt to reproduce the original speech waveform according to some fidelity criteria. On the other hand, vocoders model the input speech according to some speech production model; then, synthesize the speech from the model. The basic make-up for coding the prediction residual in this thesis is of a vocoder model. However, techniques of waveform coding are also used. It has been shown that waveform coders tend to give better quality speech that is robust; whereas, vocoders tend to be more synthetic [64] [79]. Borrowing from the techniques of efficient waveform coders, it is conceivable to define an acceptable coding algorithm to meet quality standards at low-bit rates of transmission. A primary interest has been to produce the transmitted speech with the minimum bit rate and still meet acceptable quality [80]. Previously mentioned were methods available to date for coding of the residual. Efficient methods to improve the coding techniques are presented for coding the prediction residual.

It has been recognized that there are two efficient methods of waveform coding [79]. These are: (1) transform coding (TC) [81] and (2) sub-band coding (SBC) [36] [37]. These are characterized as frequencydomain coders, whereas examples of PCM, differential PCM, and DM are the time-domain coders. Frequency-domain coders are perceptually better than time-domain coders because they tend to exploit the pitch of the speech waveform for bit rates below 16000 bits/second. They tend to look at the spectrum of speech in blocks, whereas the predictive systems look at

adjacent samples. These two methods will be explained in detail in the next two sections.

3.3 Transform Coding

With transform coding (TC) [81], the system of speech samples is grouped into blocks, where each block corresponds to the windowed segment of the speech signal. These blocks of speech are transformed into a set of transform coefficients; then, the coefficients are quantized independently and transmitted. An inverse transform is taken at the receiver to obtain the corresponding block of reconstructed samples of speech (see Figure 19).

A basic assumption in this method is that the speech source is stationary and has a variance of σ^2 . The successive source output samples are arranged into the N-vector X; this vector X is linearly transformed using a unitary matrix A, i.e.,

$$(= AX$$
 (3.1)

where A, in general, is complex, and

$$AA^* = I$$
 (3.2)

where * denotes the transpose conjugate. The elements of \hat{Y} are the transform coefficients. These are independently quantized, yielding, \hat{Y} . The vector \hat{Y} is transmitted to the receiver and then inverse transformed. Then

$$\hat{\mathbf{X}} = \mathbf{A}^* \hat{\mathbf{Y}} \tag{3.3}$$

Since the vector \hat{X} is reconstructed output, distortion is involved. For unitary matrices the averaged mean-squared overall distortion of the

IMPLEMENTATION OF TRANSFORM CODING



Figure 19. Block Diagram of the Implementation of Transform Coding (After Zelenski and Noll, 1977).

ī

transform coder is equal to the quantization error [82]

$$\overline{D} = \frac{1}{N} \cdot E\{(X - \hat{X})^{\mathsf{T}} \cdot (X - \hat{X})\}$$
$$= \frac{1}{N} \cdot E\{(Y - \hat{Y})^{\mathsf{T}} \cdot (Y - \hat{Y})\}$$
(3.4)

where E{ } represents the expectation. The minimization of \overline{D} will yield an optimum bit-assignment rule and an optimum transform matrix A [81]. Let J_i be the number of bits/sample needed for the coefficient Y_i (an entry in the Y vector) of variance σ_i^2 so that the mean-squared distortion D_i = E[Y_i - \hat{Y}_i)²] is not exceeded. Then [82]

$$J_{i} = \delta + \frac{1}{2} \log_{2} \left[\frac{\sigma_{i}^{2}}{D_{i}} \right]$$
 (3.5)

where δ is a correction factor that takes into account the performance of a practical quantizer. The optimum number of bits for the quantizer can be obtained by minimizing the average distortion

$$\overline{D} = \frac{1}{N} \sum_{i=1}^{N} D_i$$
 (3.6)

with the constraint of a given average bit rate

$$\overline{R} = \frac{1}{N} \sum_{i=1}^{N} J_i = \text{constant}$$
(3.7)

The optimum bit assignment is [81]

$$J_{i} = \overline{R} + \frac{1}{2} \cdot \log_{2} \frac{\sigma_{i}^{2}}{\left[\begin{array}{c} N \\ \pi & \sigma_{i} \\ i=1 \end{array} \right]^{1/N}} \text{ bit/sample } i = 1, 2, ..., N$$

The average distortion is found to be

78

(3.8)

$$\overline{D} = 2^{2\delta} \cdot 2^{-2\overline{R}} \cdot \begin{bmatrix} N \\ \pi & \sigma^2 \\ j=1 \end{bmatrix}^{1/N}$$
(3.9)

Here the distortion introduced by the transform coding scheme depends on the distribution of variances. In addition, \overline{D} is found to be the geometric mean of the variance. This leads to the solution of A matrix. Let $R_{\chi\chi}$ and $R_{\chi\gamma}$ be the covariance matrices of X and Y, then

$$\det R_{yy} \leq \frac{\pi}{i=1} \sigma_i^2 \qquad (3.10)$$

and for any unitary matrix A

$$\det R_{XX} = \det R_{yy}$$
(3.11)

In particulre, the variances σ_i^2 are along the diagonal of R_{yy}; then,

$$\det R_{XX} = \frac{\pi}{i=1} \lambda_i \qquad (3.12)$$

where λ_i are the eigenvalues of R_{XX} . Therefore, the minimum distortion is found if the variances, σ_i^2 , are equal to the eigenvalues of R_{XX} [81]. The Karhunen-Loève transform (KLT) has the property that $\sigma_i^2 = \lambda_i$ for all i.

Other unique properties of KLT are: (1) transform coefficients are uncorrelated, (2) the covariance in the KLT domain is diagonal, and therefore, the transform coefficients can be quantized independently without the loss of performance [83].

It has been noted that the KLT gives optimum performance; however, there is a lack of a fast algorithm for the computation of the coefficients. In addition, the computation is quite complex. Since speech is a quasi-periodic signal, transform coding would not be efficient unless adaptive methods are used. However, this area still needs additional studies. Zelenski and Holl presented promising results. Tribolet and others [38] have done additional work in this area also. Zelenski and Noll experimented with the Walsh-Hadamard transform (WHT), the discrete slant transform (DST), the discrete Fourier transform (DFT), and the discrete cosine transform (DCT) to compare with the KLT. All these have fast algorithms and are signal independent. Zelenski and Noll found that the basis vectors of the DCT and KLT are close; however, the KLT is signal dependent. It has been shown that the performances of the DCT and KLT are similar [84]. The studies of Tribolet and others found TC to be complex and costly; however, this method proves to be superior when compared to other systems [38].

3.4 Sub-Band Coding

It is desired to retain the basic components of speech composition and phonemic quality. TC is a very efficient method of completing the endeavor; however, due to cost and complexity, it was discarded. The method of sub-band coding [36] has some very distinct advantages whereby the original goal can be met in order to secure as much of the speech signal as possible. One criterion, perceptual in nature, is the retention of transitional information. Also, the intelligibility of speech can be maximized by the use of the Articulation Index [29], which is discussed in Appendix C.

With sub-band coding the frequency spectrum is partitioned such that each sub-band contributes accordingly to the speech intelligibility which is quantified by the Articulation Index. The Articulation Index is a weighted fraction representing, for a given speech channel and noise condition, the effective proportion of the normal speech signal which is available to a listener for conveying speech intelligibility [86]. The **spe**ech spectrum can be divided into 20 frequency bands contributing 5 percent each to the Articulation Index. In this case, the frequency spectrum can be bandpass filtered in such a way that they contribute equally to the Articulation Index. An example given by Crochiere and others [36] in Table IV addresses a sub-band partitioning of four bands between 200 to 3200 Hz.

TABLE IV

SUB-BAND PARTITIONING EXAMPLE

Sub-Band No.	Frequency	Range	(Hz)
1	200 -	700	
2	700 -	1310	
3	1310 -	2020	
4	2020 -	3200	

Obviously, there are other possibilities of partitioning the speech band [37]. Each band contributes an equal 20 percent to the total Articulation Index. The total Articulation Index is 80 percent, which corresponds to a word intelligibility of approximately 93 percent [36]. Sub-band coding has another advantage which involves quantization. Each sub-band is quantized separately and each band contains its own distortion and, therefore, quantization noise could be considered separately for each band [36]. Furthermore, because of the nature of the spectrum of speech, the detectability of this distortion is not the same at all frequencies.

Since the proposed method is based upon sub-band coding the residual, the presentation is in terms of the prediction residual, $e_f(k)$.

For the following discussion, assume that the sub-bands are partitioned as shown in Figure 20. Let the width of each of these bands be identified by

$$W_n = \omega_{n+1} - \omega_n$$
 $n = 1, 2, ..., N=4$ (3.13)

where ω_n corresponds to the edges of these bands. The implementation of the sequence of operations leading from the residual to the coded output for transmission is shown in Figure 21. Also, shown in Figure 21 is the implementation at the receiver. From this figure, it follows that

$$r(k) = (e_{fn}(k) \cos \omega_n k) * h_n(k)$$
 (3.14)

where $e_{fn}(k)$ corresponds to the output of the nth bandpass filter and $h_n(k)$ corresponds to the impulse response of the nth lowpass filter. It is clear that

$$W_n \leq 2\omega_n \tag{3.15}$$

in order that the frequency bands are properly separated. Then r(k) is decimated to the rate $2W_n$ from the original sampling frequency. This signal is then encoded and multiplexed with the other channels. At the receiver, the signal is demultiplexed, decoded, interpolated, demodulated



Figure 20. Partitioning of Frequency Spectrum into Four Sub-Bands (After Crochiere, 1976)



Figure 21. Sequence of Operations Relating to the nth Sub-Band

and bandpass filtered to give $\hat{e}_{fn}(k)$. This is shown in Figure 21. The nth sub-band is then summed with other bands to produce $\hat{e}_{f}(k)$, which is the sub-band coded and decoded version of the signal. The total implementation of the system will be discussed later.

3.4.1 Sub-Band Coding and Transform Coding

Earlier it was pointed out that frequency-domain coders can be considered as a good basis for an efficient coder. In this section the relationship between sub-band coding and transform coding is discussed.

Considering the ideal case, in which there are M sub-bands corresponding to the M samples, let the discrete cosine transform (DCT) of the residual signal, $e_f(k)$, k = 0, ..., M-1, be represented by [84]

$$\alpha_{0} = \frac{1}{\sqrt{M}} \sum_{k=0}^{M-1} e_{f}(k)$$

$$\alpha_{n} = \frac{\sqrt{2}}{\sqrt{M}} \sum_{k=0}^{M-1} e_{f}(k) \cos \frac{(2k+1)n\pi}{2M}$$

$$n = 1, ..., M-1 \quad (3.16)$$

Correspondingly, the residual signal $e_{f}(k)$ is given by

$$e_{f}(k) = \frac{1}{\sqrt{M}} \alpha_{0} + \sqrt{\frac{2}{M}} \sum_{n=1}^{M-1} \alpha_{n} \cos \frac{(2k+1)n\pi}{2M}$$
 $k = 0, 1, ..., M-1$

which obviously corresponds to the inverse discrete cosine transform (IDCT). Using

$$\omega_{n} = \frac{(2k+1)nII}{2M}$$
(3.18)

in Figure 21, it is seen after modulation and low-pass filtering

 $r(k) = \alpha_n \qquad k = 0, 1, ..., M-1$ (3.19)

(3.17)

Since there are M sub-bands corresponding to the M frequencies, and since r(k) is a constant, it follows that after the decimation only one point is given for each band and that value is α_n . The encoder in Figure 21 codes the DCT coefficient. This points out the fact that in the ideal case (i.e., filters and modulators are ideal), the sub-band coding will be equivalent to the discrete cosine transform coding. Obviously, the discussion above can be generalized for the case wherein there are N sub-bands (N < M) rather than M.

It is clear that where the components in the sub-band coder are nonideal, the r(k) are not equal to α_n . Further work is necessary in quantifying the difference between r(k) and α_n [85].

Noting the simplicity in the sub-band coder and also noting the relationship between the transform coder and sub-band coder, the sub-band coder is more practical.

3.5 Determination of Frequency Sub-Bands Based on Articulation Index

The Articulation Index (AI) is a weighted fraction representing, for a given speech channel and noise condition, the effective proportion of the normal speech signal which is available to a listener for conveying speech intelligibility [29].

In this section, the methods of determining how to achieve maximum intelligibility based on using the AI are examined. There are two methods for computing AI. The first method, called the 20-band method by French and Steinberg [86], is based on measurements or estimates of the spectrum of the speech and noise present in each of the 20 continuous bands of frequencies. Each band contributes equally to the speech intelligibility. The second method, known as the octave-band method, is derived from the first method. It requires measurements of the speech **and** noise present either in certain one-third-octave-band or in certain full octave bands.

Some researchers consider these two, i.e., one-third-octave-band and full octave-band measurements, as different methods. The octave-band method is not as sensitive to variations in the speech and noise spectra as the 20-band or the one-third-octave-band method. An example where it falls apart is in situations where an appreciable fraction of the energy of the masking noise is concentrated in a band of frequency that is one octave or less in width; under these conditions, the one-third-octaveband or the 20-band method would be better to use.

The 20 frequency bands are those specified by Beranek for male voices [87]. These bands are shown in Table XXIV in Appendix C. In order to use the 20-band method to calculate the AI, the peaks of the spectrum of the speech signal (PSS) must be approximated first. The level depends on if the speech is spoken through earphones or a loudspeaker. There is an adjustment to either case of -65 dB which is considered as the over-all long-term rms sound-pressure level of an idealized speech spectrum. However, with the loudspeaker, an additional amount is adjusted according to Table V [29]. This is due to the assumption that the room is semireverberant; whereas, earphones do not present reverberance.

These corrections are obtained from experiments conducted in a reverberant room using a loudspeaker and from experiments conducted in an anechoic chamber [29].

Also an additional correction must be added to correct for the noise spectrum. This is shown in Table VI [29]. The noise that reaches the

listener's ear is assumed to be that of a steady-state nature. All noises in the listener's environment and the noise in transmission systems are combined to arrive at the noise spectrum level.

TABLE V

ADJUSTMENTS TO THE SPECTRUM OF THE SPEECH SIGNAL

Maximum Spectral Values of Speech Signal	Amount to be Subtracted
85 dB	0 dB
90	2
95	4
100	7
105	11
110	15
115	19
120	23
125	27
130	30

The corrected noise spectrum (NS) has the effect of masking the speech signal. The noise spectrum increases at a faster than normal rate when the band sensation level of the speech sound exceeds 80 dB [86]. This band sensation level is defined as the difference in decibels between the sound integrated over a frequency band and the sound pressure level of that band when the speech sound is at the threshold of audibility in **an** anechoic room. The increase in masking is taken into account in the calculation of AI by adding to the PSS. If the band sensation level of the sound exceeds 80 dB at the center frequency of a band, then the PSS is increased by the amount that is shown in Table VI.

TABLE VI

ADJUSTMENTS FOR NOISE SPECTRUM

Band	Sensation Level	Added Amount
	80	. 0
	85	1
	90	2
	95	3
	100	4
	105	5
	110	6
	115	7
	120	8
	125	9
	130	10
	135	11
	140	12
	145	13
	150	14

The noise spectrum level (NS) is compared to PSS at the mid-frequencies of the 20 bands given in Table XXIV in Appendix C. Values that are zero or less are set to zero. When PSS exceeds the noise by 30 dB, then that difference is set to 30. This is due to the limitation on the dynamic range of speech [87].

The Articulation Index is defined as

$$AI = \sum_{n} W_{n} \cdot (\Delta A)_{max}$$
 (3.20)

where

 $(\Delta A)_{max}$ is the contribution from one band and has a maximum value of 0.05.

W_n is the percent of maximum contribution by any one band and

$$W_n = \frac{PSS - NS}{30}$$
(3.21)

where 30 represents the dynamic range of the speech band and is a normalized so that W_n is limited to unity. Therefore, for 20 bands, the normalization is limited to 600. An illustrative example is given by Kryter [29].

Consider the one-third-octave-band and octabe-band method. The center and cut-off frequencies for these are shown in Tables VII and VIII [29].

With the one-third-octave and octave-band methods, the correction levels shown in Table V should be considered for signals received from the loudspeaker. Also, the NS must be calculated from Table VI, and the weighting factors need to be computed from (3.21) for each band. These are then summed to give the AI for the speech system operating under the noise conditions and the level of speech.

TABLE VII

FREQUENCIES RELATED TO ONE-THIRD-OCTAVE-BAND METHOD

One-Third-Octave Band	Center Frequency
179 - 224	200
224 - 280	250
280 - 353	315
353 - 448	400
448 - 560	500
560 - 706	630
706 - 896	800
896 - 1120	1000
1120 - 1400	1250
1400 - 1790	1600
1790 - 2240	2000
2240 - 2800	2500
2800 - 3530	3150
3530 - 4480	4000
4480 - 5600	5000

TABLE VIII

FREQUENCIES RELATED TO OCTAVE METHOD

Octave Band	Center Frequency
180 - 355	250
355 - 710	500
710 - 1400	1000
1400 - 2800	2000
2800 - 5000	4000

To consider how the different methods compare for the same speech signal and masking noise, Kryter computed the AI for each of these methods. For 20-band method, AI = 0.38; for one-third-octave method, AI = 0.33; and for octave-band method, AI = 0.28. Since the 20-band method is the basic method from which all others are derived, it provides the "correct" AI and the others are compared to this AI.

The AI can be compared to estimated speech intelligibility scores as shown by graph in Figure 22. It is noted that the intelligibility score is highly dependent on the constraints placed on the message communicated.

The greater constraint (for instance, the smaller the amount of information content in each item of message), the higher the percent intelligibility score for a given AI. No single AI can be used as a criterion for an acceptable communication value. It is a function of messages transmitted and the enunciation of the talker [29].





The Articulation Index is a good quantitative measure of speech intelligibility. Speech communication can be enhanced by an equal application of AI across the speech spectrum in the sub-band coding. In the next section, the transitional cueing of phonemes, another aid that adds to speech communication is discussed.

3.6 Transitional Information

The human speaks in an uninterrupted and continuous fashion to communicate thoughts. The underlying basis for communication is the phonemic structure that connects itself by means of transitional cues for the perception of certain phonemes [1]. It is the transitional information that must be enhanced to aid the perception needed for absolute discrimination of speech-like sounds [2]. Transitional cues are a set of frequency shifts which occur in the second-formant region where a consonant and a vowel join. The perception of a given phoneme is strongly conditioned by the transitional information of its neighbors [2].

The identification of phonemes has been studied under various conditions by a group at the Haskins Laboratories [1]. Many of their experiments have used synthetic syllables. The combinations of syllables included consonant-vowel (CV) syllables. The consonant is usually a stop out of a group of phonemes with the same voicing. The vowels were maintained at two formants. Further work has been done by Rabiner [88] for synthesis of phonemes by rules. These concluded that one frequency variable of the consonant was generally adequate to distinguish that a consonant of the group was uttered. To further distinguish the

consonant, the stop-vowel formant transitions were necessary to perceive the consonant.

Figures 23(a) and 23(b) illustrate the stop-vowel formant transitions. In Figure 23(a), the vowel /a/ has first and second formants occurring at 700 Hz and 1200 Hz respectively. It is seen that the consonants /b/, /d/ and /g/ demonstrate a different rise or fall in the second formant region. The second formant varies because each consonant has a different place of articulation. The place of articulation for /b/, /d/ and /g/ are front, middle and back, respectively. It is seen in Figure 23(a) that the consonants appear to commence from some trajectory determined by their place of articulation.

The trajectory point is further illustrated in Figure 23(b). This figure uses the consonant /d/ and three vowels, /a/, /i/ and /u/. It is shown that the consonant /d/ has a loci of points that commence in the region of 1600 Hz for the second formant. It has been shown that consonants exhibit this property of transition from a particular frequency to the steady-state value of the vowels [1].

The consonants that are perceptually heard with falling second formants to the vowel /a/ are /d/ and /g/. The consonant /b/ is heard with a rising second formant to the vowel /a/. It is noted that a shift in second formant frequency is bounded. With falling transitions of the second formant, /g/ is heard for steady-state levels of frequency between 2280 and 3000 Hz; however, between 1320 and 2280 Hz the sound could be /g/ or /d/; and, below 1320 Hz, it is identified as /d/ [1].

The importance of second-formant transitions is shown for perceptual purposes. Differences in the acoustic speech signal are due to the excitation and vocal tract configuration for different consonants. These




will differ as shown in Figure 23 by the transition regions. The consonant transitions are the principal cues for the perception of a partitular consonant. The transition occurs because the vocal tract has one shape for the vowel and one shape for the consonant. The change in the vocal tract and the effect that the glottal pulse has on vowels has been addressed recently [12]. Later, the coding aspects of transitional information will be discussed.

3.7 Relation of Perception

to Intelligible Speech

A topic that has been mentioned several times before is perception. Perception related to the Articulation Index and transitional information together for discrimination of speech sounds. A quantitative description of speech perception is not possible. However, in a qualitative sense, speech perception can be enhanced when the intelligibility of speech is increased. In this section, several aspects of speech perception will be discussed to show the need to address this subject.

Speech perception can be defined as the ability for humans to discriminate and differentiate the character of speech sounds. Discrimination is examined along fundamental dimensions of the hearing mechanism and, in general, one dimension at a time. The ear takes measurements and makes differential comparisons. These comparisons may be of frequency and intensity. The over-learned senses of the brain distinguishes the speech from other periodic waves. Further, the speech must be broken in to its discrete elements, the phonemes. Once the signal is perceived as speech, there are other factors that determine the fundamental characteristics of recognizing intelligible speech.

The ability to recognize and understand speech determines intelligibility. The intelligibility of speech may be affected in several ways [86]. These may include echoes, phase distortion, or reverberation. Unnatural sounding speech can influence intelligible understanding of speech sounds. The intensity of the speech may affect intelligibility of speech received by the ear. Noise in a transmission medium may affect intelligibility by masking the speech. The talker and listener have several factors that can cause unacceptable intelligibility related to the speech [86]. These are given below:

- a. The basic characteristics of the speech can be destroyed.
- b. The electrical and acoustic instruments which operate between the talker and the listener may not be adequate.
- c. The condition under which the communication takes place may not be acceptable.
- d. As a result of c., the behavior of the talker and listener may be modified by the characteristics of the communication system.

The perception of intelligible speech is related to the amount of information spoken. This is shown in Figure 22. The exactness with which the listener identifies speech sounds is related to the size of the vocabulary and the sequence or context of the message. As seen from Figure 22, the more predictable the message is, the better the intelligibility. It has been shown that as the vocabulary size increases, a higher signal-to-noise ratio is necessary to maintain performance [2].

Perceptual aspects of speech are influenced greatly by semantics and context. The ability to predict the speech utterance enhances intelligibility. The grammatical rules of a language are part of the human over-learned senses [65]. Consequently, the language prescribes a certain allowable sequence of words. The semantic factors occur as part of the rules because certain words must be associated with meaningful units [66]. It has been shown that intelligibility of speech is substantially higher when a grammatically correct and meaningful sentence is spoken than when using the same words randomly [65]. The over-learned senses reduce the number of alternative words from the context, and therefore, the listener has improved intelligibility.

The application of speech perception is an adaptive process. The listener uses the detection procedure within the reception system of the ear to determine the speech communication process. The listener can absolutely identify speech when given the basic sound elements of the speech. The sound elements are discriminated and differentiated from other periodic sounds to perceive speech. If the speech is intelligible, the exactness is not only related to how good the transmission medium is but also to the length of the utterance and its context. These concepts are applied in the next section to aggregate a coding algorithm for transmission of perceptually enhanced speech.

3.8 Basis of Coding the Predictional Residual

A coding method is presented to perceptually enhance the speech. The method uses sub-band coding (SBC) for coding the prediction residual. Besides SBC being conceptually simple, it has the additional advantage that each sub-band is quantized separately and each band contains its own distortion. It should be pointed out that the input to the sub-band coder is the residual signal rather than the speech signal. Some of the reasons for this approach are:

a. A more efficient bit distribution based on energy/frame.

b. A more pronounced pitch information in the residual signal, andc. An ideal input for the synthesizer at the receiver.

In an earlier section, the advantages of using the Articulation Index in SBC have been discussed. Each sub-band is selected such that each contributes equally to the Articulation Index [36]. However, it has been shown that "satisfactory" performance can be expected if this equal contribution to the articulation criterion can be met within a factor of two [37] [87]. This relaxation of the criteria was allowed for integerband sampling with good results [36] [37]. That is, the sub-bands are between $m_i \omega_i$ and $(m_i + 1) \omega_i$, where m_i is an integer. The method has popularity because it eliminates the need for modulators. Even though the integer-sampling method requires less hardware, the selection of subbands using the articulation criteria would give better perception. There has been some research done in the selection of the sub-bands by this method [37]. Also, it should be pointed out that the sub-band selection depends on the multiplexing of the encoded speech [37]. This subject will be further discussed in Chapter IV.

The coding scheme of the residual is based on enhanced transitional cues. It has been shown that the second formant is important for perceptual purposes. The exact development will be discussed in this section.

The spectrum of the signal is used for calculation of the energy. The energy can be represented by [108]

$$E \approx \frac{1}{N} \frac{N-1}{\sum_{k=0}^{\infty} |E_{f}(k)|^{2}}$$
 (3.22)

where $E_{f}(k)$ corresponds to the discrete Fourier transform (DFT)

coefficients of the signal $e_f(k)$, which can be computed by using the fast Fourier transform (FFT) algorithm.

Equation (3.22) is applied to the prediction residual to compute the energy. The spectrum of the prediction residual is partitioned into four sub-bands as stated before. Using (3.22), the energies in each sub-band can be expressed by

$$E_n = \frac{1}{N} \sum_{k=0}^{N-1} |E_{fn}(k)|^2$$
 n = 1, 2, 3, 4 (3.23)

where $E_{fn}(k)$ is the DFT coefficient of the signal corresponding to the nth sub-band.

Now the total energy can be expressed by

$$E_{T} = \sum_{n=1}^{\infty} E_{n}$$
 (3.24)

Among speech sounds, E_T has wide variance. Previous researchers have not studied the variations in E_T of the speech sounds for each prediction residual. This aspect is discussed in the next section.

3.8.1 Energy Distribution

This section gives the results on the energy data for phonemes. The goal of the energy study is to distinguish between vowels, nasals and noisy sounds. This data is used in the next chapter to determine the bit distribution in the coding algorithm.

The phonemic data used in this thesis was obtained from recordings of a number of monosyllabic utterances of a male talker made in an anechoic chamber. These utterances were lowpass filtered to 3600 Hertz. The lowpassed filtered signal was then digitized at 8000 Hertz using the program DIGITIZ. The digitized data is stored on the INTERDATA computer system disk in data file BURGE.DAT.

For future use, the sentence data in digitized format [146] is stored on the IBM 370 computer system. The data was lowpass filtered to 4000 Hertz and samples at 8000 Hertz. This data is stored in the files listed in Table IX with a description of the data. Representative sonagrams of Table IX are shown in Appendix D.

TABLE IX

SENTENCE DATA

Sentence Description	File
"The pipe began to rust while new"	OSU.ACT10161.SPEECH1
"Add the sum to the product of these three"	OSU.ACT10161.SPEECH2
"Open the crate but don't break the glass"	OSU.ACT10161.SPEECH3
"Oak is strong and also gives shade"	OSU.ACT10161.SPEECH4
"Thieves who rob friends deserve jail"	OSU.ACT10161.SPEECH5
"Cats and dogs each hate the other"	OSU.ACT10161.SPEECH6

The phonemic utterances used in this thesis are shown in Table X. Table X represents a wide variety of speech sounds. The consonants /b/ and /h/ are used to utter syllables of the form consonant-vowel-consonant (CVC) with the consonant /d/ in the final position for the vowels, such

T/	٩B	LE	Х

DUC	NE	MIC	DAT	Δ
rπu	INC	MILC.	DAT	٦.

No.	Utterance	No.	Utterance	No.	Utterance	No.	Utterance
1		41	/y/	81	151	121	/hid/
2	/i/	42	/y/	82	/bit/	122	/hId/
3	/i/	43	/m/	83	/bIt/	123	/hɛd/
4	/1/	44	/m/	84	/bɛt/	124	/hæd/
5	/I/	45	/n/	85	/bæt/	125	/h/d/
6	/ε/	46	/n/	86	/bAt/	126	/h⊃d/
7	/2/	47	/n/	8/	/h⊃t/	12/	/hUd/
8	/ 32 /	48	/ŋ/	88	/bUt/	128	/hud/
.9	/ <i>a</i> e/	49	/D/	89	/fut/	129	/n/a/
10		50	/ D/	90		130	/nald/
11	/ //	51	/۵/	91	/D3/a/	131	/n_la/
12	/a/	52	/۵/	92		132	/naud/
13	/a/	53	/g/	93	/D_1/	133	/noud/
14		54	/g/	94	/Dau/	134	/neid/
15		55	/p/	95	/ DOU/	135	/njua/
10	/0/	50	/p/	96	/Delt/	130	/awa/
1/	/0/	5/	/t/	97	/ju/	13/	/ala/
10	/u/	58		98	/W11/	138	/ara/
19	/u/	59	/ t/	99	/111/	139	/aya/
20	/ 31	60	/ K/	100	/r11/	140	/ama/
21	/ 31	62	/K/	101	/y11/	141	/ana/
22	/ 1 1 /	62	/11/	102	/1111/	142	/sev
23	/ . [/	64	/.11/	103	/111/	143	/aba/
24 25		65	/ J/	104	/551/	144	
26		66	/] /	105		145	/aya/
27	/21/	67	/ []/	100		140	/aµa/ /ata/
28	/aU/	68		102	/gii/ /nI1/	1/12	/aca/
29	/01/	69		100	/+11/	1/0	/aka/
30	/01/	70	121	110	////	150	/ana/
31	/el/	71	101	111	/hI1/	150	/aja/
32	/eI/	72	/ 8/	112	/iI1/	152	/ava/
33	/ ill/	73	/ = /	113	/+/11/	153	/a%a/
34	/iU/	74	/_/: /f/	114	/vI1/	154	/aza/
35	/w/	75	/f/	115	/8ae t/	155	/afa/
36	/w/	76	/0/	116	/aIl/	156	/aθa/
37	/1/	77	/0/	117	/fI1/	157	/asa/
38	/1/	78	/s/	118	/bæ0/	158	/a∫a/
39	/r/	79	/s/	119	/sIl/		,,
40	/r/	80	151	120	//11/		

as /hid/. The vowel /a/ is used to utter nonsense syllables of the form vowel-consonant-vowel (VCV) in both initial and final positions, such as /aba/. A set of minimal units using the final form -/II/ (-ill) is used for the consonants also. Some of the other syllables used are English words. The basic sounds are found in Table II.

The phonemes are analyzed by the algorithms in Appendix B. The energy data is shown in Table XI, normalized by the sound $/\alpha e /$ for the first 81 phonemes in Table X. The energy in the phoneme $/\alpha e /$ corresponds to the largest compared to each of the other phonemes. The data is calculated by the program ENERGY. From Table XI, it can be seen that the energy of the prediction residual divides the phonemes into classes by phonemic aggregations.

It is well known that with simple LPC methods [60], the excitation function is a set of periodic pulses or random noises which can be identified as high or low energy excitation functions. However, by using the energy data in Table XI, the phonemes can be grouped into three classes, namely high energy, low energy and noise groups. The high energy group includes the vowels and diphthongs. The plosive, fricative and unvoiced phonemes make up the noise group. The low energy group is composed of glides and nasals. It follows that an ideal excitation signal for speech would enhance perception by considering a three-tier classification rather than the conventional two-source model. This would include a source for vowels, a source for nasals and glides, and a source for fricatives. This is the result of the phoneme energy study of the prediction residual. A normalized energy distribution by phoneme for each sub-band is shown along with the energy bands in Figure 24.

TAE	BLE	ΧI

Phoneme	Total Frequency	Band	SB1	SB2	SB3	SB4
/i/	.44	•	. 58	.46	. 33	.31
/I/	.75		.51	.46	.75	.45
/ε/	.84		.65	.47	1.00	.54
/œ/	1.00		1.00	1.00	1.00	1.00
///	.72		.6/	.57	.45	.41
/a/	./2		.64	.69	.59	. 35
7_7	.03		.00	.00	.70	.42
/ 0/	.24		.23	.25	.20	20
/u/ /ɛ/	61		62	.10	64	.20
/aI/	1.00		1.00	.79	.68	.65
$/\Box I/$.44		.75	.45	.20	.32
/aU/	1.00		1.00	.95	. 90	.78
/oU/	.56		1.00	.31	.21	.53
/eI/	.86		1.00	.64	.66	.49
/jU/	. 32		.67	.21	.22	.12
/w/ .	.24		.35	.24	.10	.23
/1/	.24		.29	.08	.10	.29
/r/	.14		.24	.12	.09	.07
/ 9/	. 1 1		.20	.08	.08	.0/
/111/	. 34		.05	.25	.22	.19
/ 11/	.22		.45	.17	24	18
/ŋ/ /h/	.37		.07	.37	14	17
/d/	.21		.63	.27	.14	.21
/ ɑ/	.31		.50	.18	.14	.16
/g/	.18		.27	.08	.07	.08
/t/	.45		.46	.32	.26	.25
/k/	.32		.63	.23	.19	.20
/h/	.45		.46	.24	.31	.31
/j/	.53		.51	.44	.31	.58
/t∫/	.23		.46	.16	.10	.11
/v/	.16		.29	.13	.09	.09
/8/	.17		.32	.13	.12	.10
/2/	.24		.44	.17	.19	.15
/1/	.07		.04	.07	.05	.12
15/	08		.05	.06	.06	.03
151	.10		.06	.07	.07	.18

ENERGY BY PHONEME FOR PREDICTION RESIDUAL





Based on the above discussion, phonemes can be classified into three energy groups: (1) high energy (HE), (2) low energy (LE) and (3) noise (N). To do this, the normalized residual phoneme energies (second column in Table XI) are the first tabulated; from this, there are clear breaks in the energy levels and therefore three energy groups formed. These breaks are used to identify the threshold values for a particular energy group. For the high energy group, let T_{11} be the threshold value. That is, any phoneme that has normalized residual energy greater than T_{11} is classified into the high energy group. Similarly, T_{22} and T_{33} are the established threshold values for low energy and noise phonemes respectively. The three groupings are given in Table XII. The threshold values T_{11} , i = 1, 2, 3, can be identified from Figure 24. These are for the entire frequency range.

TABLE XII

PHONEME	ENERGY	GROUPINGS

Phonemes
i, I, ε, <i>α</i> ε, a, Λ, ⊃, U, u, ξ
m, n, n, z, w, l, r, y
∫, f, b, d, g, p, t, k

For the sub-band coding, threshold values need to be computed for each band. Also, each energy group has to be divided into four subgroups corresponding to the four sub-bands. Let E_{in} be the normalized signal energy in the nth frequency band corresponding to the phoneme that is in the ith energy group. This is explicitly shown in Table XIII. For example, E_{12} represents the energy in the second frequency band corresponding to the high energy phoneme (first energy group).

The threshold values for E_{in} (referred hereafter as E_{in}^{T}) in Table XIII will now be established using columns 3, 4, 5 and 6 in Table XI.

TABLE XIII

SYMBOLIC REPRESENTATION OF ENERGY DISTRIBUTION

		F	Frequency Band					
		1	2	3	4			
	Н	E	E ₁₂	E ₁₃	E14			
Energy	L	E ₂₁	E22	E ₂₃	^E 24			
	N	E ₃₁	E ₃₂	E ₃₃	^Е з4			

 E_{in} is listed for various phonemes in columns 3, 4, 5 and 6 in Table XI. To make the classification speaker independent, the E_{in} has to be normalized by E_{T} given in (3.24). Let

$$E_{in}^{T} = \frac{E_{in}}{E_{T}}$$
 i = 1, 2, 3
n = 1, 2, 3, 4 (3.25)

From this, it is clear that

$$E_{in}^{T} \leq 1.0$$

 $i = 1, 2, 3$
 $n = 1, 2, 3, 4$ (3.26)

As before, E_{in}^{T} in (3.25) are tabulated for i = 1, 2, 3 and n = 1, 2, 3, 4. The breaks are established from this tabulation and the threshold values are obtained from these breaks. These are tabulated in Table XIV. The array in Table XIV will be referred hereafter as energy threshold matrix. This matrix will be used in computing the bit allocation scheme, which is discussed in the next chapter.

TABLE XIV

ENERGY THRESHOLD MATRIX

	Fr	requenc	cy Ban	d
	1	2	3	4
Н	.58	.27	1.0	.75
L	.50	.19	1.0	.86
N	.46	.27	1.0	1.00

3.9 Summary

In this chapter, the basis of coding the prediction residual at the rate of 9600 bits/second using the techniques of sub-band coding was presented. Transform coding and sub-band coding wwere discussed along with their relationship. The method of achieving maximum intelligibility based on the Articulation Index was presented. Transitional information of speech along with the relation of speech perception to intelligibility was discussed. Phonemes have been divided into three energy groups so that these can be used in the bit allocation scheme to be discussed in Chapter IV.

CHAPTER IV

ENERGY BASED SUB-BAND CODING ALGORITHM

4.1 Introduction

In this chapter the sub-band coding algorithm, introduced in Chapter III, is examined with the prediction residual as the input source signal. The coding algorithm combines spectral analysis and waveform coding techniques. The combination is intended to provide perceptual enhancement of the speech. The perceptual aspects of speech are a key factor in the bit distribution of the coding algorithm. The bit allocation is established by using the energy groups discussed in the last chapter. For each frame and for each sub-band, the energy $E_n = \frac{1}{N} \sum_{k} |E_{fn}(k)|^2$ is computed, where E_n indicates the energy corresponding to the nth sub-band in a given frame.

It is well known that most of the spectral density for vocalic sounds and the fundamental frequency are basically found in the sub-band number one (lowest frequency band). The intensity of the energy is substantially high. Spectrogram data can show this. The second formant resides predominantly within the second and third sub-bands and is of the low energy type. These formants determine the transitional cues for certain perceptual effects. The energy of noisy speech sounds, i.e., voiceless fricatives, plosives, etc., has a basic flat spectrum and most of the energy is above 2 kHz. The perceptual effects are discerned in this frequency range. The spectrograms show the intensity of the signal

energy represented by varying shades of gray or black areas [2]. The higher the energy, the darker the area. Spectrograms are included in Appendix D. These figures are included to show the different energy levels associated with different phonemes. From these spectrograms, it can be seen that vowels are typified by dark areas; whereas fricatives, plosives, etc., are shown in a gray area. Although all voiced sounds show a dark color on the spectrogram, Makhoul and Wolf [90] have shown that nasals and glides have a lighter shade when compared to other voiced sounds.

In this study, the energy in each frame of the prediction residual is calculated for each type of phoneme. The bits per sample in each band is allocated on an adaptive basis, using the perceptual criteria discussed in the last chapter. The next section deals with the bit allocation scheme.

The bit allocation method is incorporated into the sub-band coder, which is discussed in Section 4.3. The adaptive strategy is combined with a uniform quantizer with results presented in Sections 4.4 and 4.5. Section 4.6 gives the details of the modules for computational aspects of the coding of the prediction residual.

4.2 Bit Allocation

In this section, the bit allocation scheme is discussed using the energy groupings in Tables XII in Chapter III. In symbolic form, the bit distribution is shown in Table XV for a three-energy level--four sub-band coder, where the rows correspond to the energy levels and the columns correspond to a particular frequency band. For example, k_{23} bits per

sample are assigned for the second energy (LE) band and the third frequency band.

TABLE XV

SYMBOLIC REPRESENTATION OF BIT DISTRIBUTION

	Frequency Band				
	1	2	3	4	
High Energy (H)	_k וו	k ₁₂	k ₁₃	k ₁₄	
Low Energy (L)	k ₂₁	k ₂₂	^k 23	k ₂₄	
Noise (N)	к ₃₁	k ₃₂	k ₃₃	^k 34	

The bits are allocated by the empirical formula

$$k_{ij} = \log_2(1 + \frac{E_{ij}}{\sigma_j})$$
 $i = 1, 2, 3$
 $j = 1, 2, 3, 4$ (4.1)

where ${\rm E}_{ij}$ is the energy from Table XIII and σ_j is a normalization factor determined from the constraint

$$\sum_{j=1}^{4} k_{ij} N_{j} = C \quad i = 1, 2, 3 \quad (4.2)$$

with N_j , j = 1, 2, 3, 4, being the number of samples in each band after decimation. The value of C is equal to the total number of bits/frame

minus the number of sync bits per frame. Combining (4.1) and (4.2), it follows that

where the normalization factor, σ_j , can be determined from (4.3). Equations (4.1), (4.2) and (4.3) define the algorithm.

The normalization factor is included to take into consideration the perceptual aspects of the signal. It is used as a weighting factor for transitional cueing. It has been shown that pitch, formant areas, nasality and affrication are important for speech perception. Within the speech spectrum, these characteristics occur in certain frequency ranges. The power density of speech can indicate this conception, and is discussed below.

The speech power density spectrum is shown in Figure 25. It is clear that most of the energy is below 1000 Hertz. It has been shown by Miller and Nicely [91] that below 1000 Hz, voicing, nasality, and affrication are predominant for determination of the phonemic content. It has been pointed out that given a set of speech signals, a weight factor can be derived when the speech is separated into sub-bands. When these signals are coded properly, there is an advantage of distinguishing certain perceptual effects such as voicing, nasality and affrication. The perceptual effects can be used for calculation of bits for coding.

To compute the normalization factor properly for coding the residual signal, a bit matrix is chosen. The bit distribution that is selected is based on perceptual concepts. This matrix will be referred to as an <u>a</u> priori bit matrix. In addition to perceptual concepts, the a priori bit





matrix is selected such that the bit rate is 9600 bits/second for the sub-bands given in Table IV. The matrix is shown in Table XVI, where the entries will be referred to as k_{ij}^A to denote the <u>a priori</u> values.

TABLE XVI <u>A PRIORI</u> BIT MATRIX DISTRIBUTION

		Frequency Band				
		1	2	3	4	
High Energy	1	4	3	2	2	
Low Energy	2	3	3	3	2	
Noise	3	2	3	3	3	

The <u>a priori</u> bit matrix is based on experimental results on phonemes. A cursory inspection of Table XVI reveals that the perceptual criteria is preserved. For example, on lower bands where pitch and formant data must be preserved as accurately as possible, a large number of bits per sample are used for encoding, whereas for upper bands where fricatives and noisy sounds are predominant, fewer bits per sample are used. Note that the same number of bits for each energy group is allocated. Also, the <u>a priori</u> bit values (k_{ij}^A) are used to compute the normalization factor in (4.1).

When the energy of the speech sound is determined to be high enough, the energy threshold introduced in Chapter III selects the energy matrix (from Table XIII) and <u>a priori</u> bit values (from Table XVI). These are used to calculate the normalization factor from (4.1), and

$$\sigma_{j} = \frac{E_{ij}^{T}}{\binom{k_{ij}^{A}}{2} - 1} \qquad i = 1, 2, 3 \qquad (4.4)$$

where E_{ij}^{T} is the energy obtained from threshold matrix and k_{ij}^{A} is obtained from the <u>a priori</u> bit matrix. Figure 26 gives the distribution of $(1/\sigma_{j})$ based upon (4.4).

Equation (4.1) can now be used to allocate the bits. It should be pointed out that in using this equation, actual energy values of the signal will be used rather than the threshold values. The following steps are performed to allocate the bits.

1. Spectral estimates are computed for each sub-band.

The total energy in the frame for the entire frequency band is computed.

3. E_{ij}'s are computed.

4. Normalization factor, σ_j , is computed

5. The bits are allocated by

where E_{ij} is the energy in the jth sub-band corresponding to the ith energy group and σ_j is the normalization factor from (4.4). Figure 27 gives the flow chart for the bit allocation scheme.







Equation (4.5) has been simulated using the phonemes in Table XII. The bits are averaged for each energy group. The results of the simulations are shown in Figure 28 for each of the three energy groups. Distinctly shown is a separation of the energy groups. Note that the low energy group which contains the nasalic and glide sounds is shown to separate the high energy and noise groups. This separation supports the three-source theory of the residual signal.

Earlier, it was shown that the residual signal parallels glottal excitation. The use of the residual signal for encoding the speech and later exciting the speech synthesizer has several benefits. The bits are minimized in the first and second sub-bands, reducing the necessary transmission rate for these sub-bands. It is unnecessary to transmit twice as many bits for sounds with nasalic, glide or liquid characteristics. On the other hand, the discrimination from the noise is shown to be distinct. The benefit remains clear further, because perceptual criteria will be enhanced in all sub-bands. Discrimination of sounds can be benefited with a minimum bit allocation.

The bit distribution is shown by frame for each phoneme in Figure 29. Again, it is shown that the perceptual criteria is preserved in that the pitch and formant prediminant phonemes receiving a substantial bit allocation and fewer bits are allocated for fricative and plosive phonemes. Noting that the total number of allowed bits per frame is constant, the difference in bits per energy group is adjusted in the synthesis bits. This is discussed in detail in the next section.

4.3 Sub-Band Encoding of the Prediction Residual

The bit allocation scheme was used in the perceptual aspects of

BIT DISTRIBUTION BY SUB-BAND





speech in sub-band coding of the prediction residual. The sub-band coder partitions the frequency band of the residual signal into four sub-bands by using the bandpass filters. The partitioning of the frequency bands is shown in Figure 20. Each sub-band is low-pass translated, decimated [by the Nyquist interval obtained from (3.15)], and encoded according to the bit allocation scheme discussed above. It has been shown that separate coding of each sub-band accomplished the preferenctial perception criteria for that band [37]. The decoding of each sub-band involves an interpolation and translation back to the original band. The bands are summed to arrive at an estimate of the original residual signal (see Figure 21). This section describes the sub-band coding parameters, the relation of the sub-bands to the Articulation Index and other perceptual criteria discussed in this thesis.

The cutoff frequencies for the sub-band coder are shown in Table XVII. The guideline established for selection of cutoff frequencies is to represent an approximately equal contribution to the Articulation Index. The bands shown in Table XVII represent enough of the important frequencies such that intelligibility is preserved.

TABLE XVII

SUB-BAND CODER CUTOFF FREQUENCIES

Band	Cutoff Frequency (Hz)
1 2 3 4	$\begin{array}{rrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrr$





Figure 29. Bit Distribution for Phonemes by Frame

The integer-band sampling scheme [37] was also analyzed at the sampling rate of 8000 Hertz. The technique requires the ratio of upper to lower cutoff frequencies of the sub-band be $(m_i + 1)/m_i$, where m_i is an integer. These bands are related at the bit rate such that the data can be synchronized when multiplexed. Table XVIII is helpful in constructing the sub-bands. Previous authors have given the choice of bands that relate at other sampling rates [36] [37]. Shown in Table XVIII are integer-band sampling cutoff frequencies for an 8000 Hertz sampling rate. The integer decimation ratio is shown in Column 1 for 8000 Hertz. The band-widths, f_i are indicated in Column 2. The sampling rate, $2f_i$, is shown in Column 3. In Columns 2, 3 and 4, the cutoff frequencies are indicated implicitly. Integer-band sampling is not used in this thesis, and is given here for completeness.

To explain how each band is related, the analysis of the sub-band coder is discussed. The sub-band coder is designed for 9600 bits/second. The transmitted coder parameters include the sub-band coded prediction residual signal, PARCOR coefficients and sync bits. Table XIX represents a breakdown of sub-band coder parameters for the high energy phonemes. Table XX shown sub-band coder parameters relative to the low energy sounds. Table XXI represents those parameters relative to the noise sounds. The difference in Tables XIX, XX, and XXI are the bits allocated and the transmission rates per band, and the sync bits.

It is well known that the decimation rate shown in Column 4 of Tables XIX through XXI represent an integer number of samples available before encoding. These available samples are related to the 9600 bits/second transmission rate. The fractional representation for each frame and subband samples are shown in Table XXII.

TABLE XVIII

Decimation Ratio	fi	2f _i	3f _i	4f _i
1	4000	8000	12000	16000
2	2000	4000	6000	8000
3	1333	2666	3999	5332
4	1000	2000	3000	4000
5	800	1600	2400	3200
6	666	1332	1998	2664
7	571	1142	1713	2284
8	500	1000	1500	2000
9	444	888	1332	1776
10	400	800	1200	1600
11	363	728	1089	1452
12	333	666	999	1332
13	308	616	924	1232
14	286	5/2	858	1144
15	200	534	798	1064
10	250	500	750	1000
10	235	4/0	705	940
18	210	444	620	000
19	200	420	600	040
20	100	380	570	760
21	190	364	5/0	700
23	174	348	522	696
24	167	334	501	668
25	160	320	480	640
26	154	308	462	616
27	148	296	444	592
28	143	286	429	572
30	133	266	399	532
31	129	258	387	516
32	125	250	375	500
			,	,

INTEGER-BAND SAMPLING CUTOFF FREQUENCIES FOR 8000 HERTZ SAMPLING RATE

TABLE XIX

SUB-BAND CODER PARAMETERS RELATIVE TO HIGH ENERGY PHONEMES

Band	Cutoff Frequency (Hz)	Sampling Frequency (Hz)	Decimation Rate	Bits Allocated	Transmission Rate (b/s)
1	250 - 500	50 0	16	4.0	2000
2	500 -1000	1000	8	3.0	3000
3	1142 - 1700	1142	7	1.5	1700
4	2000 - 3000	2000	4	1.0	2000
Sync	and Synthesis		1		900
· · ·	· · ·				9600 b/s

TABLE XX

SUB-BAND CODER PARAMETERS RELATIVE TO LOW ENERGY PHONEMES

Band	Cutoff Frequency (Hz)	Sampling Frequency (Hz)	Decimation Rate	Bits Allocated	Transmission Rate (b/s)
1	250 - 500	500	16	2.0	1000
2	500 - 1000	1000	8	2.0	2000
3	1142 - 1700	1142	7	1.0	1142
4	2000 - 3000	2000	4	1.25	2500
Sync a	and Synthesis				2958
					9600 b/s

TABLE XXI

SUB-BAND CODER PARAMETERS RELATIVE TO NOISE ENERGY PHONEMES

Band	Cutoff Frequency (Hz)	Sampling Frequency (Hz)	Decimation Rate	Bit Allocated	Transmission Rate (b/s)
1	250 - 500	500	16	1.0	500
2	500 - 1000	1000	8	1.0	1000
3	1142 - 1700	1142	7	.5	571
4	2000 - 3000	2000	4	.5	1000
Sync a	and Synthesis				6529
					9600 b/s

TABLE XXII

Band	Fraction/Frame	Samples/Frame
1	.207	53
2	.312	80
3	.180	45
4	.207	53
Sync and Synthesis	.094	24
	1.000	256 Samples/Frame

REPRESENTATION OF SAMPLES FOR A FRAME FOR HIGH ENERGY SOUND

The multiplexing (see Figure 21) is simulated on the computer by first appending each of the decimated signals to 256 points per frame by adding zeros. Second, the DFT's of these are taken. Third, the transformed signals are summed. Finally, the IDFT of the summed signal is the multiplexed signal, which has 256 points. The demultiplexing in Figure 21 is simulated using the inverse process. That is, first, the decoded signal is transformed. Second, it is divided into four frequency bands. Third, these frequency coefficients in each band are appended by zeros to get 256 points. Finally, the IDFT of these signals are taken, which gives the demultiplexed signals.

Shown in each of Tables XIX through XXII is a band labeled "Sync and Synthesis." These parameters include synchronization bits and synthesis parameters for the receiver. The synchronization bits include one to

establish the beginning of a frame and three to determine if the frame contains a high, low or noise energy signal. The remaining samples in the sync and synthesis bits are allocated to the PARCOR coefficients for synthesizing the speech.

The PARCOR coefficients are distributed between the range of $|k_j| \leq 1$ and, in most cases, the entire range is not required [28]. It has been shown that the odd-ordered coefficients are somewhat skewed toward the positive side, whereas the even-ordered coefficients are someqhat skewed toward the negative side [28]. The limitation of a quantizer range results in better speech quality for a given number of bits assigned to each coefficient. These parameters have been studied in depth in the literature. Further quantization characteristics of the PARCOR coefficients can be found in [7] [28] [31] [119]. These aspects are used in adjusting the synthesis bits in Tables XIX to XXII, and is outlined below.

Specifically, the following procedure can be used in assigning bits for synthesis parameters. For high energy sounds, 20 bits can be utilized for the 10 PARCOR coefficients. The bit allocation for low energy sounds for the PARCOR coefficients is 70. For the noise energy sounds, the bit allocation is 170. Note that more bits are available for the PARCOR coefficients corresponding to the low energy and the noise signals as compared to the high energy signals. These allocations in synthesis parameters for encoding are adequate. Actual implementation of the bit allocations for the PARCOR coefficients and their effect on the coder has yet to be done.

The find tuning of quantization parameters has yet to be done. The total sub-band system requires many trade-offs in the analysis section.

In the analysis section, allowance must be made for the transmission rate for each sub-band. In the next section, the uniform quantization method is discussed.

4.4 Adaptive Uniform Quantization

The sub-band coder partitions the residual signal into four frequency bands. These banded signals are passed to the quantizer for reduction of information content. The design of the quantizer is determined by the bits allocated as discussed earlier. The amplitude of each residual signal sample is quantized into one of 2^{IBITS} levels, where IBITS is the number of bits allocated for the sub-band. The information content of the digitized signal is IBITS bits per sample. It is shown in Column 6 in Tables XIX through XXI that the information rate for each sub-band is

Information Rate = $(Sampling Freq.)_n \times I$ bits/second

$$I = 1, ..., IBITS$$
 (4.6)

where $(Sampling Freq.)_n$ is the sampling frequency for the nth sub-band.

After quantization the discrete amplitude level of the signal sample has a value expressed in binary decimal of length IBITS. The value of IBITS ranges from 1 to 5. For example, the value of 2 for IBITS yields amplitude levels of 00, 01, 10 and 11; whereas, a value of 5 would yield 32 five binary length words.

The range of the quantizer is aligned such that the amplitudes of the input residual signal will be within the range of the maximum swing of the output of the quantization levels. The method for accomplishing the assurance that no overload occurs is based on a scheme of analyzing each frame before quantization; i.e., the range of the signal is found before quantization. This is compared to the bits allocated. An adjustment is
made if needed by rounding the bits allocated to the next integer. The method of quantization will be discussed next.

It has been shown that a characteristic of sub-band coded speech is that it has no sample-to-sample correlation [36] [37]. Following this, encoding is best performed by adaptive pulse code modulation (APCM) [109] [121]. Previous encoding based on differential or fixed prediction does not achieve good results for speech using sub-band coders [37]. Each sub-band utilizes a uniform quantizer characteristic. Each sub-band exhibits a different level of energy; therefore, an adaptive uniform quantizer is used utilizing a technique that shrinks and expands the quantizer by sub-band such that the signal is within the range of the maximum quantization level for that sub-band.

To implement the adaptive uniform quantizer, let the step size be denoted by Δ . Figure 30 illustrates the characteristic for the adaptive uniform quantizer [109] and will be idscussed in detail. It is well known that the uniform quantizer level produces error which follows the uniform distribution. That is, the probability density function of the quantization error Q_p is given by

$$f(Q_e) = \frac{1}{\Delta}; \frac{-\Delta}{2} \le Q_e \le \frac{\Delta}{2}$$
(4.7)

with the variance

$$\sigma^2(Q_e) = \frac{\Lambda^2}{12}$$
 (4.8)

The step size is dependent on the bits allocated.

Let the number of levels be represented by

$$NL = 2^{1}$$
 $i = 1, ..., IBITS$ (4.9)





then

$$\Delta = \frac{2}{NL} \left[e_{fn}(n) \right]_{max}$$
(4.10)

where $[e_{fn}(n)]_{max}$ is the maximum value of the nth sub-band residual signal.

In order to achieve the quantized value, let

$$\Psi = [\xi_1, \xi_2, ..., \xi_{NL}]$$
(4.11)

be an (NL)-vector used to identify the parameters of the quantizer levels such that

$$\underline{Q} = \Delta \cdot \underline{\psi} \tag{4.12}$$

where the vectors \underline{Q} and $\underline{\psi}$ are of dimension NL and represent the quantizer values. The entries in $\underline{\psi}$ are given by

$$\xi_{\ell} = -\left(\frac{NL}{2} - \ell + 1\right) \qquad 1 \leq \ell \leq \frac{NL}{2}$$
$$= 0 \qquad \ell = \frac{NL}{2} + 1$$
$$= \ell \qquad \qquad \frac{NL}{2} + 2 \leq \ell \leq NL \qquad (4.13)$$

From (4.12) and (4.13), it follows that the quantized level Q_{ℓ} in <u>Q</u> is given by

$$Q_{q} = \Delta \cdot \xi_{q} \qquad (4.14)$$

The quantized values of the residual signal are obtained by rounding it to the nearest quantized level, which is used to code the signal.

In the next section, performance measures are discussed for the quantizer and the sub-band coder.

4.5 Signal-to-Noise Ratio Performance Measurements

In the previous section, the quantization is done for the banded prediction residual. In this section, performance measuremetns will be discussed. It has been recognized in the literature that signal-to-noise ratio (SNR) is an inadequate performance measure for speech coding [109]. This idadequacy is realted to the idea that additive white noise is not a good model for error waveforms in speech quantization. Generally, most authors supplement the SNR by subjective and perceptual measurements as a rule.

The SNR is still the single most informative measure for quantizer performance [109]. If the quantizer is designed for maximum SNR, the step size can be chosen according to the probability density function of the signal [122]. However, the SNR improvement is offset by greater idle channel noise for speech [123]. The result is poorer subjective performance [123]. Therefore, to enhance SNR an adaptive quantizing technique is used based on the allocation of bits.

It has been shown that transform coding with adaptive quantizers maximizes SNR and lowers the idle channel noise [81]. Intuitively, subband coding should follow under similar conditions. With sub-band coding, the quantization noise of each band is contained within that band and therefore, minimizes the quantization noise of the coded speech [36]. Due to the characteristics of the speech spectrum, the quantization distortion is not equally detectable at all frequencies. This technique offers a means of controlling the quantization noise across the speech spectrum and, therefor a realization of improvement in signal quality [36].

135

The definition of each objective measure will be discussed next. Perhaps the most common measurement of performance is the conventional (normalized) SNR which is defined as

NSNR = -10 log₁₀
$$\begin{bmatrix} N-1 \\ \Sigma (x(k) - y(k))^{2} \\ \frac{k=0}{N-1} \\ \Sigma x^{2}(k) \\ k=0 \end{bmatrix}$$
 (4.15)

where x(k) is the input to the coder and y(k) is the output of the decoder. It is assumed that the numerator represents the noise of the coding technique, such that as the noise decreases a smaller SNR will be the result of the summation in (4.15). The advantage of this quantity is a representation of the normalization of the error between the coder input and the decoder output. For speech there is no perceptual advantage in maximizing the SNR; however, the SNR in (4.15) could be optimized for the autocorrelation of the speech [122].

Another measure similar to (4.15) is the root-mean-square error which is defined as

RMSSNR = -20 log₁₀
$$\left[\sqrt{\frac{N-1}{\sum_{n=0}^{\infty} (x(n) - y(n))^2}{\frac{n=0}{N}}} \right]$$
 (4.16)

where x(n) and y(n) are defined as before. In (4.16), the error is assumed to be of random nature, and is normalized by the factor N, the number of data points.

A third measure is defined as

136

MSSNR =
$$-\frac{1}{N} \sum_{n=0}^{N-1} 10 \log \left[\frac{(x(n) - y(n))^2}{x^2(n)} \right]$$
 (4.17)

where x(n) and y(n) are expressed as before. The representation in (4.17) defines some measure of error.

The results using (4.15), (4.16) and (4.17) are shown in Table XXIII. These are computed by program SNRCAL (see Appendix B). These results exemplify good coder performance. Note that these simulations are done without bit assignment to PARCOR coefficients. Several phonemes are used in these measurements and they give an adequate measure of the coder. However, the complete simulation should include quantization of all parameters to complete the 9600 bits/second coding algorithm. The next section discusses the computational aspects for coding and decoding the prediction residual.

TABLE XXIII

Phoneme	RMSSNR	NSNR	MSSNR	
/I/ /ε/ /ae! /۸/ /a/ /u/ /æ/ /aI/ /aU/ /oU/	29.2 37.2 35.1 32.9 30.1 36.8 29.8 31.3 34.2 29.4	36.7 36.9 37.4 34.9 38.4 38.7 38.0 37.0 38.4 37.9	18.2 19.1 17.5 15.2 18.3 17.7 18.4 18.2 18.7 16.8	
/eI/	33.4	39.0	17.0	

SIGNAL-TO-NOISE PERFORMANCE MEASUREMENT FOR SEVERAL PHONEMES

4.6 Computation for Coding the Prediction Residual

The flow chart that gives all the computer modules is given in Figure 31 for coding the residual signal. The data blocks shown in Table XXIV represent data processed and online storage during the computations.

TABLE XXIV

Data Block Name	Record Length	Number of Records	Module Used
BURGE.DAT	256	82	DIGITIZ/WINDOW
WINDOW.DAT	256	16	WINDOW/AUTO/LATTIC/INVERS
AUTO.DAT	176	16	AUTO/LATTIC/INVERS
RESIDUAL.DAT	256	16	INVERS/LATTIC/FFTMGR/SUMLPD
SPECTM.DAT	256	16	FFTMGR/RESULT/(PITCH)
BITS.DAT	16	16	FFTMGR/SUMLPD/ENCODE
PHAZ.DAT	256	16	FFTMGR/RESULT
CODE.DAT	256	16	ENCODE/DECODE
SIGNAL.DAT	256	16	DECODE/RESULT
SQNR.DAT	256	16	RESULT
SBAND1.DAT	256	16	SUMLPD/ENCODE
SBAND2.DAT	256	16	SUMLPD/ENCODE
SBAND3.DAT	256	16	SUMLPD/ENCODE
SBAND4.DAT	256	16	SUMLPD/ENCODE

DATA BLOCKS FOR PROCESSING AND STORAGE





The modules are arranged to generate and use the data in Table XXIV on the INTERDATA 70. The tape recorder inputs an analog signal to the computer while DIGITIZ computes a sampled signal and stores the digitized signal on disk in location BURGE.DAT. DIGITIZ is set up to store 4096 points. This program calls an assembly language digitizer and sequence clock for sampling. This program is flexible for sampling any analog signal and storing the signal on disk.

Program WINDOW uses the data on disk, BURGE.DAT. The data is windowed using a 256-point Hamming window. The user has the option of selecting which record of the digitized data to window. The program reports the sequence selected and also scales the data. The window data is written in data block WINDOW.DAT.

Routine AUTO calculates predictor and PARCOR coefficients using Levinson's method [61]. The program uses as input the window data, WIN-DOW.DAT. The output is an array, AUTO.DAT, containing autocorrelation coefficients, predictor coefficients, cross-correlation coefficients and reflection coefficients.

Routine INVERS uses the data from AUTO, AUTO.DAT, for use in the lattice filter implementation from Equation (2.31) and (2.34). The order of the filter is ten. The output from this program are the residual values. This output is stored in RESIDUAL.DAT. Routine LATTIC is the sames as INVERS except that LATTIC gives the user the option to produce a plot of the speech and prediction residual on CALCOMP.

The FFTMGR module is an FFT manager that includes a bit reversal and unscrambler. The input to this program is the prediction residual, RE-SIDUAL.DAT. This routine calculates the avergae spectrum, magnitude square and the energy of the prediction residual. It calculates the energy per sub-band. It uses an <u>a priori</u> estimate of the energy and bits to calculate the normalization factor and bits for each sub-band. The program writes on disk the spectrum, SPECTM.DAT, the bits allocated, BITS.DAT and the phase, PHAZ.DAT. It also gives the user the option for a plot of the spectrum on CALCOMP.

Routine SUMLPD passes the prediction residual through a digital bandpass filter. The signal is modulated, lowpass filtered and decimated as shown in Figure 21. The input to this program is the data file RESID-UAL.DAT. The outputs are the four sub-bands, SBAND1.DAT, SBAND2.DAT, SBAND3.DAT, and SBAND4.DAT.

The signal corresponding to the four sub-bands are encoded using the bits allocated in BITS.DATA by useing the program ENCODE. ENCODE allows for 32 levels of code. In case of non-integer numbers, the quantizer, QUNTIZ, rounds the bits to determine the number of quantizable levels. A uniform quantization is used to determine the code. The output is written in CODE.DAT.

Routine DECODE uses CODE.DAT as input. In the initial frame, the maximum number of quantization levels is determined. This maximum sets the level for the inverse quantizer. Then the signal is decoded and written in file SIGNAL.DAT.

The program RESULT interpolates, modulates and bandpasses the signal, SIGNAL.DAT, for reconstruction. The routine calculates signal-to-noise ratio given by (4.15) and (4.17).

4.7 Summary

In this chapter, the energy based sub-band coding algorithm was presented. The method of allocation of bits was discussed. The design of the sub-band encoding of the prediction residual was presented. The computational aspects for coding the prediction residual were discussed.

CHAPTER V

SUMMARY AND SUGGESTIONS FOR FURTHER STUDY

5.1 Summary

This thesis investigates an efficient coding of the prediction residual using the technique of sub-band coding at the bit rate of 9600 bits/ second. The energy of the prediction residual is used to distribute the bit allocation by sub-bands such that perceptual criteria is preserved. The perceptual criteria is enhanced by transition information embedded in the phoneme connections of speech by a technique that weights the energy based on a normalization factor.

Each sub-band is partitioned such that there is an equitable contribution to the Articulation Index as it is a measure of speech intelligibility. This is discussed in relation to the quality of speech. The perception of speech is described in a qualitative sense. The relationship between the Articulation Index and transitional information is described as a method of discrimination of speech sounds.

The prediction residual is discussed as a parallel to the glottal waveform. The prediction residual is formed by speech through an inverse filter. This is represented as a deconvolution of speech from the vocal tract filter.

The vocal tract filter is modeled as a recursive digital filter using the method of linear prediction. Linear prediction produces the

143

prediction residual, which is the difference between the actual and predicted speech signals. Because the prediction residual is parallel to glottal excitation, the prediction residual is an ideal pitch extractor.

A novel pitch extraction technique is presented. It is a two-stage method that estimates the residual spectrum and uses time samples of the residual to calculate the approximation of the pitch. The technique calculates a threshold which uses squared samples to extract the pitch within a frame. Also it includes an error check that estimates wide variances of the pitch within each period and is then updated.

The three-tier classification of phonemes is derived from the energy study of the phonemes for the prediction residual. It is shown that the energy of the prediction residual divides the phonemes into classes by phonemic aggregations, namely high energy, low energy and noise groups. The high energy group includes the vowels and diphthongs. The plosive, fricative and unvoiced phonemes compose the noise group. The low energy group is composed of glides and nasals.

The three-tier classification of the energy levels along with the four frequency bands allows for efficient allocation of bits per sample for each band. The above method aids in preserving perceptual criteria and preserves pitch-formant data by the allocation of a large number of bits per sample in the lower bands. Since fricative and noisy sounds are predominant in the upper bands, a smaller number is used in the lower bands. The perceptual criteria is further enhanced by a normalization factor.

The normalization factor is perceptual in nature and is used as a weighting factor for transitional cueing. The derivation of the

normalization factor is discussed. Additional variations are given for the relationship of the three phonemic classes to the normalization factor.

The sub-band coder is designed based on the normalization factor, the energy data, and the bit allocation. The parameters are computed on a frame-by-frame basis. The sub-bands are constructed such that the bit rate of the data from each band can be synchronized when multiplexed at 9600 bits/second. The integer-band sampling scheme is analyzed at the sampling rate of 8000 Hertz for a 9600 bits/second transmission rate. The sub-band coder is designed to transmit the coded prediction residual signal, synthesis parameters and sync bits at the 9600 bits/second rate.

An integral part of the sub-band coder is the quantizer. The encoding of the signal is designed based on adaptive pulse code modulation. Uniform quantization is used. The characteristics of the quantizer are discussed in detail. Performance of the quantizer is described in terms of signal-to-noise ratios (SNR) for objective criterion for quality. The conventional (normalized) SNR is used for representing the error of the coder input and the decoder output. The mean-square SNR is used for an indication of gross error. These SNR measurements are only an indication for quantizer performance. Generally, the SNR must be supplemented by subjective and perceptual measurement as a rule. However, the SNR measurements in this thesis are used without listeners.

In the following, some extensions to the present effort are suggested. Appropriate references are indicated.

5.2 Suggestions for Further Study

5.2.1 PARCOR Coefficient Study of Sensitivity

The PARCOR coefficients introduced in Chapter II have been thoroughly investigated because of their importance to speech analysis and synthesis [9] [28] [31]. The priority is geared toward the synthesis of speech; in that given the prediction residual and PARCOR coefficients, the speech signal can be adequately regenerated. An extension of the present work would enhance present efforts in this area by studying the sensitivity of PARCOR coefficients with respect to the sub-band coding of the prediction residual.

5.2.2 Sub-Band Coding Using Subjective

Measurements

The present work can be further advanced by the use of sub-band coding the prediction residual at various bit rates. The synthesized signal would then be used in a comparative study for various bit rates. The perceptual question concerning the method should be geared towards a recording of the synthesized speech so that a set of listeners could hear the results.

5.2.3 Energy Threshold Matrix Study

The introduction of the energy threshold matrix (ETM) in Chapter III requires further study. In this work it is seen that the ETM is highly dependent of perceptual criteria; consequently, several variations would benefit the present work. In some instances, it is necessary to bias the energy group to enhance the perceptual aspects; but this is unknown until the energy distribution is computed. The results of ETM are dependent on transmission rates; however, given one transmission rate, several ETM may be equally applicable to the coding.

5.2.4 Integer-Band Coding of the Prediction

<u>Residual</u>

The integer-band coding method introduced in Chapter IV for use with the prediction residual has not been considered in this thesis. It is simple to implement and would minimize the need for modulators. Previous authors have studied this for speech; however, the subject has not been studied for the prediction residual [36] [37].

5.2.5 Prediction Residual and Noise

A study that would greatly benefit the speech coding area is to mask the prediction residual with white noise. That is,

$$z(k) = e_{f}(k) + v(k)$$

where $e_{f}(k)$ represents the discrete samples of the prediction residual signal and v(k) represents the discrete damples of the white noise.

The enhancement of the pitch period markings would be of major importance in this study. Further, the synthesized signal-to-noise ratio performance measurements would also be of interest. The speech waveform has been examined in noise stripping environments; however, the prediction residual in a noise environment has results that are promising [19] [58]. An aid to characterization of the signal would be to use the Laplacian or Gamma distribution, as with the speech. However, these distributions are questionable for the prediction residual since the waveform is different. **Determining the probability distribution of the prediction residual may be a study in itself.**

5.2.6 Modeling the Prediction Residual

The prediction residual in this thesis is obtained by inverse filtering the speech signal. Under certain conditions, it is not easy to code the inverse filter; however, if a model was determined that is similar to the signal, it would be of benefit for synthesis. An extension of the work in Chapter II would be to compare the speech and the prediction residual. It would be necessary to identify the essential parameters that can be derived from the residual signal, such as pitch, phase in f_0 , formant characteristic and noise between pitch period pulses. The end results would approximate an expression that compares with the actual residual pulse. This in turn could be compared with Flanagan and Rosenberg's work [2] [12] [32].

BIBLIOGRAPHY

- Delattre, P. C., A. M. Liberman, and F. S. Cooper. "Acoustic Loci and Transitional Cues for Consonants." <u>The Journal of the</u> <u>Acoustical Society of America</u>, Vol. 27, No. 4 (1955), 769-773.
- (2) Flanagan, J. L. <u>Speech Analysis Synthesis and Perception</u>. New York: Springer-Verlag, 1972.
- (3) Dudley, H. "Remaking Speech." <u>The Journal of the Acoustical Society</u> of America, Vol. 11 (1939), 165.
- (4) Dudley, H., R. R. Riesz, and S. S. A. Watkins. "A Synthetic Speaker." <u>Journal of the Franklin Institute</u>, Vol. 227, No. 6 (1939), 739-764.
- (5) Atal, B. S., and S. L. Hanauer. "Speech Analysis and Synthesis by Linear Prediction of the Speech Wave." <u>The Journal of the</u> <u>Acoustical Society of America</u>, Vol. 50, No. 2 (Part 2) (1971), 637-655.
- (6) Atal, B. S., and M. R. Schroeder. "Adaptive Predictive Coding of Speech Signals." <u>Bell System Technical Journal</u> (1970), 1973-1990.
- (7) Markel, J. D., and A. H. Gray, Jr. <u>Linear Prediction of Speech</u>. New York: Springer-Verlag, 1976.
- (8) Markel, J. D. "Digital Inverse Filtering A New Tool for Formant Trajectory Estimation." <u>IEEE Transactions on Audio and Elec-</u> <u>troacoustics</u>, Vol. AU-20, No. 2 (1972), 129-137.
- (9) Itakura, F., and S. Saito. "A Statistical Method for Estimation of Speech Spectral Density and Formant Frequencies." <u>Electronics</u> and Communications in Japan, Vol 53-A, No. 1 (1970), 36-43.
- (10) Un, C. K., and D. T. Magill. "The Residual-Excited Linear Prediction Vocoder with Transmission Rate Below 9.6 Kbits/s." <u>IEEE Transactions on Communications</u>, Vol. COM-23, No. 12 (1975), 1466-1474.
- (11) Markel, J. D., A. H. Gray, Jr., and H. Wakita. <u>Linear Prediction of Speech Theory and Practice</u>. Santa Barbara, California: Speech Communications Research Laboratory, Inc., SCRL Monograph No. 10, 1973.

- (12) Rosenberg, A. E. "Effect of Glottal Pulse Shape on the Quality of Natural Vowels." <u>The Journal of the Acoustical Society of</u> America, Vol. 49, No. 2, Part 2 (1971), 583-590.
- (13) Dunn, J. G. "An Experimental 9600-Bits/s Voice Digitizer Employing Adaptive Prediction." <u>IEEE Transactions on Communication Tech-</u> nology, Vol. COM-19, No. 6 (1971), 1021-1032.
- (14) Gibson, J. D., S. K. Jones, and J. L. Melsa. "Sequential Adaptive Prediction and Coding of Speech Signals." <u>IEEE Transactions</u> on Communications, Vol. COM-22, No. 11 (1974), 1789-1797.
- (15) Cohn, D. L., and J. L. Melsa. "The Residual Encoder An Improved ADPCM System for Speech Digitization." <u>International Communi-</u> cations Control Conference Record (1975), 30-26 to 30-30.
- (16) Goldberg, A. J., A. Arcese, T. McAndres, R. Chueng, and R. Freudbert. <u>Kalman Predictive Encoder</u>. Needham Heights, Massachusetts: GTE Sylvania, Report to Defense Communications Engineering Center, Contract No. DCA 100-74-C-0058, 1975.
- (17) McDonald, R. A. "Signal-to-Noise and Idle Channel Performance of Differential Pulse Code Modulation Systems - Particular Applications to Voice Signals." <u>Bell System Technical Journal</u> (1966), 1123-1150.
- (18) Qureshi, S. U. H., and G. D. Forney. "Adaptive Residual Coder An Experimental 9.6/16 KB/S Speech Digitizer." <u>EASCON</u> 1975 Record (1975), 29A-29E.
- (19) Berouti, M., and J. Makoul. "High Quality Adaptive Predictive Coding of Speech." <u>International Conference on Acoustic, Speech</u> and Signal Processing Record (1978), 303-306.
- (20) Esteban, D., C. Galand, D. Manduit, and J. Menez. "9.6/7.2 KBPS Voice Excited Predictive Coder (VEPC)." <u>International Conference on Acoustics, Speech and Signal Processing Record</u> (1978), 307-311.
- (21) Melsa, J. L., D. L. Cohn, J. D. Gibson, R. Kolstad, D. Kopetzky, G. Lauer, and J. Tomkic. <u>Study of Sequential Estimation Method</u> <u>for Speech Digitization</u>. Notre Dame, Indiana: University of Notre Dame, Report to Defense Communications Engineering Center, Contract No. DCA 100-74-C-0037, June 16, 1975.
- (22) Atal, B. S., and M. R. Schroeder. "Predictive Coding of Speech Signals and Subjective Error Criteria." <u>IEEE Transactions on</u> <u>Acoustics, Speech and Signal Processing</u>, Vol. ASSP-27, No. 3 (1979), 247-254.
- (23) Cohn, D. L., and J. L. Melsa. "A New Configuration for Speech Digitization at 9600 Bits Per Second." <u>International Conference on</u> <u>Acoustic, Speech and Signal Processing Record</u> (1979), 550-553.

- (24) Chang, C. S. "An Improved Residual Encoder for Speech Compression." <u>International Conference on Acoustic, Speech and Signal Pro-</u> <u>cessing Record</u> (1979), 542-545.
- (25) Magill, D. T., C. K. Un, and S. E. Cannon. <u>Speech Digitization Excitation Study</u>. Menlo Park, California: Standford Research Institute, Report to Defense Communication Engineering Center, SRI Project 1526-8.
- (26) Dankberg, M. D., and D. Y. Wong. "Development of a 4.8-9.5 Kbps RELP Vocoder." <u>International Conference on Acoustic, Speech</u> and Signal Processing Record (1979), 554-557.
- (27) Viswanathan, R., W. Russell, and J. Makhoul. "Voice-Excited LPC Coders for 9.6 KBPS Speech Transmission." <u>International Conference on Acoustics</u>, Speech and Signal Processing Record (1979), 558-561.
- (28) Kang, G. S. <u>Application of Linear Prediction Encoding to a Narrow-</u> <u>band Voice Digitizer</u>. Washington, D. C.: Naval Research Laboratory, NRL Report 7774, 1974.
- (29) Kryter, K. "Methods for the Calculation and Use of the Articulation Index." <u>The Journal of the Acoustical Society of Ameri-</u> <u>ca</u>, Vol. 34, No. 11 (1962), 1689-1697.
- (30) Itakura, F., and S. Saito. "Analysis Synthesis Telephone Based on the Maximum Likelihood Method." <u>The Sixth International Con-</u> gress on Acoustics Record (1968), C17-C20.
- (31) Makhoul. J. "Stable and Efficient Lattice Methods for Linear Prediction." <u>IEEE Transactions on Acoustics, Speech and Signal</u> <u>Processing</u>, Vol. ASSP-25, No. 5 (1977), 423-428.
- (32) Flanagan, J. L. "Some Properties of the Glottal Sound Source." Journal of Speech Hearing Research, Vol. 1 (1958), 99-116.
- (33) Rabiner, L. R., B. S. Atal, and M. R. Sambur. "LPC Prediction Error - Analysis of its Variation with the Position of the Analysis Frame." <u>IEEE Transactions on Acoustics, Speech and</u> Signal Processing, Vol. ASSP-25, No. 5 (1977), 434-442.
- (34) Goodman, L. M. "Channel Encoders." <u>Proceedings of the IEEE</u>, (1967) 127-128.
- (35) White, C. E. "Bits of Voice." <u>Telecommunications</u>, Vol. 12, No. 4 (1978), 46-48.
- (36) Crochiere, R. E., S. A. Webber, and J. L. Flanagan. "Digital Coding of Speech in Sub-Bands." <u>Bell System Technical Journal</u>, Vol. 55, No. 8 (1976), 1069-1085.

- (37) Crochiere, R. E. "On the Design of Sub-Band Coders for Low-Bit-Rate Speech Communication." <u>Bell System Technical Journal</u>, Vol. 56, No. 5 (1977), 747-770.
- (38) Tribolet, J. M., P. Noll, B. J. McDermott, and R. E. Crochiere. "A Study of Complexity and Quality of Speech Waveform Coders." <u>International Conference on Acoustics, Speech and Signal Pro-</u> <u>cessing Record</u> (1978), 586-590.
- (39) Barabell, A. J., and R. E. Crochiere. "Sub-Band Coder Design Incorporating Quadrature Filters and Pitch Prediction." <u>International Conference on Acoustics</u>, <u>Speech and Signal Processing</u> <u>Record</u> (1979), 530-533.
- (40) Crochiere, R. E. "A Novel Approach for Implementing Pitch Prediction in Sub-Band Coding." <u>International Conference on Acous-</u> <u>tics, Speech and Signal Processing Record</u> (1979), 526-529.
- (41) Mathews, M. J., J. E. Miller, and E. E. David, Jr. "Pitch Synchronous Analysis of Voiced Sounds." <u>The Journal of the Acoustical</u> <u>Society of America</u>, Vol. 33, No. 2 (1961), 179-186.
- (42) Pinson, E. N. "Pitch-Synchronous Time-Domain Estimation of Formant Frequencies and Bandwidths." <u>The Journal of the Acoustical</u> <u>Society of America</u>, Vol. 25, No. 8 (1963), 1263-1273.
- (43) Sondhi, M. M. "New Methods of Pitch Extraction." <u>IEEE Transactions</u> on <u>Audio and Electroacoustics</u>, Vol. AU-16 (1968), 262-266.
- (44) Dubnowski, J. J., R. W. Schafer, and L. R. Rabiner. "Real-Time Digital Hardware Pitch Detector." <u>IEEE Transactions on Acous-</u> <u>tics</u>, Speech and Signal Processing, Vol. ASSP-24 (1976), 2-8.
- (45) Noll, A. M. "Cepstrum Pitch Determination." <u>The Journal of the</u> Acoustical Society of America, Vol. 41 (1967), 293-309.
- (46) Schafer, R. W., and L. R. Rabiner. "System for Automatic Formant Analysis of Voiced Speech." <u>The Journal of the Acoustical</u> <u>Society of America</u>, Vol. 47, No. 2, Part 2 (1970), 634-648.
- (47) Markel, J. D. "The SIFT Algorithm for Fundamental Frequency Estimation." <u>IEEE Transactions on Audio and Electroacoustics</u>, Vol. AU-20 (1972), 367-377.
- (48) Miller, N. J. "Pitch Detection by Data Reduction." <u>IEEE Trans-actions on Audio and Electroacoustics</u>, Vol. ASSP-23 (1975), 72-79.
- (49) Gold, B. "Computer Program for Pitch Extraction." <u>The Journal of</u> <u>the Acoustical Society of America</u>, Vol. 34, No. 7 (1962), 916-921.

- (50) Gold, B., and L. R. Rabiner. "Parallel Processing Techniques for Estimating Pitch Periods of Speech in the Time Domain." <u>The</u> <u>Journal of the Acoustical Society of America</u>, Vol. 46 (1969), 442-448.
- (51) Rabiner, L. R., M. J. Cheng, A. E. Rosenberg, and C. A. McGonegal. "A Comparative Performance Study of Several Pitch Detection Algorithms." <u>IEEE Transactions on Acoustics, Speech and Sig-</u> nal Processing, Vol. ASSP-24-, No. 5 (1976), 399-418.
- (52) Ross, M. J., H. L. Shaffer, A. Cohen, R. Frendberg, and H. J. Manley. "Average Magnitude Difference Function Pitch Extractor." <u>IEEE Transactions on Acoustics, Speech and Signal Processing</u>, Vol. ASSP-22 (1974), 353-362.
- (53) Maksym, J. N. "Real-Time Pitch Extraction by Adaptive Prediction of the Speech Waveform." <u>IEEE Transactions on Audio and Elec-</u> troacoustics, Vol. AU-21, No. 3 (1973), 149-153.
- (54) McGonegal, C. A., L. R. Rabiner, and A. E. Rosenberg. "A Semi-Automatic Pitch Detector (SAPD)." <u>IEEE Transactions on Acoustics, Speech and Signal Processing</u>, Vol. ASSP-23 (1975), 570-574.
- (55) Wise, J. D., J. R. Caprio, and T. N. Parks. "Maximum Likelihood Pitch Estimation." <u>IEEE Transactions on Acoustics, Speech and</u> <u>Signal Processing</u>, Vol. ASSP-24, No. 5 (1976), 418-423.
- (56) Markel, J. D. "Application of a Digital Inverse Filter for Automatic Formant and F₀ Analysis." <u>IEEE Transactions on Audio and</u> <u>Electroacoustics</u>, Vol. AU-21, No. 3 (1973), 154-160.
- (57) Itakura, F., and S. Saito. "On the Optimum Quantization of Feature Parameters in the PARCOR Speech Synthesizer." <u>International</u> <u>Conference on Speech Communication and Processing Record</u> (1972), 434-437.
- (58) Boll, S. F. "<u>A Priori</u> Digital Speech Analysis." (Unpub. Ph.D. Dissertation, University of Utah. 1973).
- (59) Barnwell, T. D., J. E. Brown, A. J. Bush, and C. R. Patisaul. <u>Pitch</u> <u>and Voicing in Speech Digitization</u>. Atlanta, Georgia: Georgia Institute of Technology, Research Report No. 3-21-620-74BU-1, 1974.
- (60) Makhoul, J. "Linear Prediction: A Tutorial Review." Proceedings of the IEEE, Vol. 62, No. 4 (1975), 561-580.
- (61) Levinson, N. "The Wiener RMS (Root Mean Square) Error Criterion in Filter Design and Prediction." <u>Journal of Mathematics Physics</u>, Vol. 25 (1947), 261-278.
- (62) Robinson, E. A., and S. Treitel. "Principles of Digital Wiener Filtering," <u>Geophysics</u> Prospectus, Vol. 15 (1967), 311-333.

- (63) Shannon, C. E. "A Mathematical Theory of Communication." <u>Bell System Technical Journal</u>, Vol. 27, No. 3 (July 1948), 379-423, and (October 1948), 623-656.
- (64) Flanagan, J. L., M. R. Schroeder, B. S. Atal, R. E. Crochiere, N. S. Jayant, and J. M. Tribolet. "Speech Coding." <u>IEEE Transac-</u> tions on Communications, Vol. COM-27, No. 4 (1979), 710-737.
- (65) Tobias, J. V., ed. <u>Foundations of Modern Auditory Theory</u>, <u>Vol. II</u>. New York: Academic Press, 1972.
- (66) Dew, D., and P. J. Jensen. <u>Phonetic Processing The Dynamics of</u> Speech. Ohio: Charles E. Merrill Publishing Company, 1977.
- (67) Rabiner, L. R., and R. W. Schafer. <u>Digital Processing of Speech</u> Signals. New Jersey: Prentice-Hall, 1978.
- (68) Jakobson, R., C. G. M. Fant, and M. Halle. <u>Preliminaries to Speech</u> <u>Analysis - The Distinctive Feature and Their Correlates</u>. Cambridge: MIT Press, 1969.
- (69) Fant, G. <u>Acoustic Theory of Speech Production</u>. The Hague, The Netherlands: Mouton, 1960.
- (70) Jayant, N. S., ed. <u>Waveform Quantization and Coding</u>, New York: IEEE Press, 1976.
- (71) Grenander, U., and G. Szegö. <u>Toeplitz</u> Forms and <u>Their Applications</u>. Berkeley, California: University of California Press, 1958.
- (72) Durbin, J. "Efficient Estimation of Parameters in Moving Average Models." Biometrika, Vol. 46, Parts 1 and 2 (1959), 306-316.
- (73) Durbin, J. "The Fitting of Time-Series Models." <u>Review of Institu-</u> <u>tion of International Statistics</u>, Vol. 28, No. 3 (1960), 233-243.
- (74) McGonegal, C. A., L. R. Rabiner, and A. E. Rosenberg. "A Subjective Evaluation of Pitch Detection Methods Using LPC Synthesized Speech." IEEE Transactions on Acoustics, Speech and Signal Processing, Vol. ASSP-25, No. 3 (1977), 221-229.
- (75) Wightman, F. L., and D. M. Green. "The Perception of Pitch." <u>Amer-ican Scientist</u>, Vol. 62, No. 2 (1974), 208-215.
 - (76) Allen, J. B. "Short-Term Spectral Analysis and Synthesis and Modification by Discrete Fourier Transform." <u>IEEE Transactions on</u> <u>Acoustics, Speech and Signal Processing</u>, Vol. ASSP-25, No. 3 (1977), 235-238.
 - (77) Allen, J. B., and L. R. Rabiner. "A Unified Theory of Short-Time Spectrum Analysis and Synthesis." <u>Proceedings of the IEEE</u>, Vol. 65, No. 11 (1977), 1558-1564.

- (78) Dudley, H. "The Vocoder." <u>Bell Labs Record</u>, Vol. 17 (1939), 122-126.
- (79) Jayant, N. S. "Waveform-Coding of Speech." Submitted for publication, Journal of the Acoustical Society of India.
- (80) Sambur, M. R. "An Efficient Linear Prediction Vocoder." <u>Bell System Technical Journal</u>, Vol. 54, No. 10 (1975), 1693-1723.
- (81) Zelinski, R., and P. Noll. "Adaptive Transform Coding of Speech Signals." <u>IEEE Transactions on Acoustics, Speech and Signal</u> Processing, Vol. ASSP-25, No. 4 (1977), 299-309.
- (82) Huang, J. J., and P. M. Schultheiss. "Block Quantization of Correlated Gaussian Random Variables." <u>IEEE Transactions on Communication System</u> (1963), 291-296.
- (83) Brown, J. L., Jr. "Mean Square Truncation Error in Series Expansions of Random Functions." <u>Journal of the Society of Indus-</u> <u>trial and Applied Math</u>, Vol. 8, No. 1 (1960), 28-32.
- (84) Ahmed, N., T. Natarajan, and K. Rao. "Discrete Cosine Transform." <u>IEEE Transactions on Computers</u>, Vol. C-23 (1974), 90-93.
- (85) Esteban, D., and C. Galand. "Application of Quadrature Mirror Filters to Split Band Voice Coding Schemes." <u>International Con-</u> <u>ference on Acoustics, Speech and Signal Processing Record</u> (1971), 191-195.
- (86) French, N. R., and J. C. Steinberg. "Factors Governing the Intelligibility of Speech Sounds." <u>The Journal of the Acoustical</u> Society of America, Vol. 19, No. 1 (1947), 90-119.
- (87) Beranek, Leo L. "The Design of Speech Communication Systems." <u>Pro-ceedings of the IRE</u>, Vol. 35 (1947), 880-890.
- (88) Rabiner, L. R. "Synthesis by Rule." (Unpub. Ph.D. Dissertation, Massachusetts Institute of Technology, 1968).
- (89) Lieberman, P. <u>Speech Acoustic and Perception</u>. New York: Bobbs-Merrill Company, Inc., 1972.
- (90) Makhoul, J., and J. Wolf. <u>Linear Prediction and the Spectral Analysis</u>. Cambridge: BBN Report No. 2304, 1172.
- (91) Miller, G. A., and P. E. Nicely. "An Analysis of Perceptual Confusions Among Some English Consonants." <u>The Journal of the</u> <u>Acoustical Society of America</u>, Vol. 27, No. 2 (1955), 338-352.
- (92) Magill, D. T., E. J. Craighill, D. W. Ellis, and C. K. Un. <u>Speech</u> <u>Digitization by LPC Estimation Techniques</u>. Menlo Park, California: Standford Research Institute, ARPA, DDC-Ad-A001931 and AD-785-738, 1974.

- (93) Allen, J. B., and R. Yarlagadda. "Digital Poisson Summation Formula and an Application." Submitted for publication, <u>IEEE Trans-</u> action on Acoustics, Speech and Signal Processing.
- (94) Allen, J. B., and L. R. Rabiner. "Unbiased Spectral Estimation and System Identification Using Short-Time Spectral Analysis Methods." Submitted for publication, <u>IEEE</u> <u>Transactions on Acous-</u> tics, Speech and Signal Processing.
- (95) Markel, J. D. "FFT Pruning." <u>IEEE Transactions on Audio and Elec-</u> troacoustics, Vol. AU-19, No. 4 (1971), 305-311.
- (96) Atal, B. S., and M. R. Schroeder. "Linear Prediction Analysis of Speech Based on a Pole-Zero Representation." <u>The Journal of</u> <u>the Acoustical Society of America</u>, Vol. 64, No. 5 (1978), 1310-1318.
- (97) Elias, P. "Predictive Coding." IRE Transactions on Information Theory, Parts 1 and 2 (1955), 16-33.
- (98) Esteban, D., and C. Galand. "32 KBPS CCITT Compatible Split Band Coding Scheme." <u>International Conference on Acoustics</u>, Speech and Signal Processing Record (1978), 320-325.
- (99) Noll, P. "A Comparative Study of Various Quantization Schemes for Speech Encoding." <u>Bell System Technical Journal</u>, Vol. 54, No. 9 (1975), 1597-1614.
- (100) Wiggins, R. H. Formation and Solution of the Linear Equations Used in Linear Predictive Coding. Bedford, Massachusetts: Mitre Corporation, Report Electronic Systems Division, Report Nos. MTR-2835, ESD-TR-74-301, 1974.
- (101) Goldberg, A. J., R. L. Freudberg, and R. S. Bheung. <u>Adaptive Multi-level 16 KB/S Speech Coder</u>. Needham Heights, Massachusetts: GTE Sylvania, Report to Defense Communications Engineering Center, Contract No. DCA 100-76-C-002, 1976.
- (102) Goldberg, A. J., and H. L. Shaffer. "A Real-Time Adaptive Predictive Coder Using Small Computers." <u>IEEE Transactions on Com</u>munications, Vol. COM-23, No. 12 (1975), 1443-1451.
- (103) Qureshi, S. U. H., and G. D. Forney. "A 9.6/16 KB/S Speech Digitizer." <u>International Communication and Control Conference Rec-</u> ord (1975), 30-31 to 30-36.
- (104) Qureshi, S. U. H., and G. D. Forney. <u>Codex Speech Digitizer Ad-vanced Development Model</u>. Newton, Massachusetts: CODEX Corporation, Report to Defense Communications Engineering Center, Contract No. DCA 100-76-C-0026, 1976.
- (105) O'Neal, J. B., and R. W. Stroh. "Differential PCM for Speech and Data Signals." <u>IEEE Transactions on Communications</u>, Vol. COM-20, No. 5 (1972), 900-912.

- (106) Burge, L. L., and R. Yarlagadda. "An Efficient Coding of the Prediction Residual." <u>International Conference on Acoustics</u>, Speech and Signal Processing Record (1979), 542-545.
- (107) Ahmed, N., and K. R. Rao. <u>Orthogonal Transformations for Digital</u> <u>Processing</u>. New York: Springer-Verlag, 1975.
- (108) Oppenheim, A. V., and R. W. Schafer. <u>Digital Signal Processing</u>. Englewood Cliffs, New Jersey: Prentice-Hall, Inc., 1975.
- (109) Jayant, N. S. "Digital Coding of Speech Waveforms PCM, DPCM and DM Quantizers." <u>Proceedings of the IEEE</u>, Vol. 62, No. 5 (1974), 611-632.
- (110) Tribolet, J. M., and R. E. Crochiere. "Frequency Domain Coding of Speech." (Unpub. paper).
- (111) Peterson, G. E., and H. L. Barney. "Control Methods Used in a Study of the Vowels." <u>The Journal of the Acoustical Society</u> of <u>America</u>, Vol. 24, No. 2 (1952), 175-184.
- (112) Winkler, M. R. "High Information Delta Modulation." <u>IEEE</u> International Convention Record, Part 8 (1963), 260-265.
- (113) Ross, M., H. L. Shaffer, A. Cohen, R. Freuberg, and H. J. Manley. "Average Magnitude Difference Function Pitch Extractor." <u>IEEE Transactions on Acoustics, Speech and Signal Processing</u>, Vol. ASSP-22, No. 5 (1974), 353-361.
- (114) Newell, A., J. Barnett, and J. W. Forgie. <u>Speech Understanding</u> System. Amsterdam: North-Holland Publishing Co., 1973.
- (115) Papoulis, A. <u>Probability, Random Variables and Stochastic Pro-</u> cesses. <u>New York: McGraw-Hill Book Co., Inc., 1965</u>.
- (116) Oppenheim, A. V., ed. <u>Applications of Digital Signal Processing</u>. Englewood Cliffs, New Jersey: Prentice-Hall, Inc., 1978.
- (117) Kalman, R. E., and B. S. Bucy. "New Results in Linear Filtering and Prediction Theory." <u>Journal of Basic Engineering, Trans-</u> <u>actions of the American Society of Mechanical Engineers</u> (1961), 95-108.
- (118) Noll, A. M. "Clipstrum Pitch Determination." <u>The Journal of the Acoustical Society of America</u>, Vol. 44, No. 6 (1968), 1585-1591.
- (119) Itakura, F., and S. Saito. "Digital Filtering Techniques for Speech Analysis and Synthesis." <u>Seventh International Con-</u> <u>gress on Acoustics Record</u> (1971), 261-264.
- (120) Noll, P. "Effects of Channel Errors of the Signal-to-Noise Performance of Speech-Encoding Systems." <u>Bell System Technical</u> <u>Journal</u>, Vol. 54, No. 9 (1975), 1615-1636.

- (121) Papoulis, A. The Fourier Integral and Its Application. New York: McGraw-Hill, 1962.
- (122) Noll, P. "A Comparative Study of Various Quantization Schemes for Speech Encoding." <u>The Bell System Technical Journal</u>, Vol. 54, No. 9 (1975), 1597-1614.
- (123) Stroh, R. W., and M. P. Paez. "A Comparison of Optimum and Logarithmic Quantization for Speech PCM and DPCM Systems." IEEE <u>Transactions on Communications</u>, Vol. COM-21, No. 6 (1973), 752-757.
- (124) Paez, M. C., and Glisson, T. H. "Minimum Mean-Squared-Error Quantization in Speech PCM and DPCM Systems." <u>IEEE Transactions</u> on <u>Communications</u>, Vol. COM-20, No. 4 (1972), 225-230.
- (125) Crochiere, R. E., L. R. Rabiner, N. S. Jayant, and J. M. Tribolet. "A Study of Objective Measures for Speech Waveform Coders." <u>Proceedings of the Zurick Seminar on Digital Communications</u> (1978), 261-267.
- (126) American Standards Association. <u>American Standard for Preferred</u> <u>Frequencies for Acoustical Measurement</u>. New York, 1960.
- (127) Strube, H. W. "Determination of the Instant of Glottal Closure from the Speech Wave." <u>The Journal of the Acoustical Society</u> <u>of America</u>, Vol. 56, No. 5 (1974), 1625-1629.
- (128) Atal, B. S. "Automatic Speaker Recognition Based on Pitch Contours." <u>The Journal of the Acoustical Society of America</u>, Vol 52 (1972), 1687-1697.
- (129) Rosenberg, A. E., and M. R. Sambur. "New Techniques for Automatic Speaker Verification." <u>IEEE Transactions on Acoustics</u>, <u>Speech</u> <u>and Signal Processing</u>, Vol. ASSP-23 (1975), 169-176.
- (130) Levitt, H. "Speech Processing Aids for the Deaf: An Overview." <u>IEEE Transactions on Audio and Electroacoustics</u>, Vol. AU-21 (1973), 269, 273.
- (131) Crochiere, R. E. "An Analysis of 16 Kb/s Sub-Band Coder Performance: Dynamic Range, Tandem Connections, and Channel Errors." <u>Bell System Technical Journal</u>, Vol. 57, No. 8 (1978), 1069-1113.
- (132) Beek, B., E. P. Neuberg, and D. C. Hadge. "An Assessment of the Technology of Automatic Speech Recognition for Military Applications." <u>IEEE Transactions on Acoustics, Speech and Signal</u> <u>Processing</u>, Vol. ASSP-25, No. 4 (1977), 310-322.
- (133) Tremain, T. E., J. W. Fussell, R. A. Dean, B. M. Abzug, M. D. Cowing, and P. W. Boudra, Jr. "Implementation of Two Real-Time Narrowband Speech Algorithms." <u>International Conference on</u> <u>Acoustics, Speech and Signal Processing Record</u> (1976), 461-472.

(134)	Crochiere, R. E. Personal Communication, August 7, 1978.
(135)	Jayant, N. S. Personal Communication, August 7, 1978.
(136)	Flanagan, J. L. Personal Communication, August 7, 1978.
(137)	Esteban, D., and C. Galand. Personal Communication, August 11, 1978, and April 3, 1979.
(138)	Markel, J. D. Personal Communication, February 21, 1979.
(139)	Allen, J. B. Personal Communication, February 22, 1979.
(140)	Sondhi, M. M. Personal Communication, April 23, 1979.
(141)	Tremain, T. E. Personal Communication, June 18, 1979.
(142)	Kang, G. S. Personal Communication, June 26, 1979.
(143)	Alde, R. Personal Communication, May 16, 1979, and June 18, 1979.
(144)	Klein, M. Personal Communication, May 16, 1979, and June 18, 1979.
(145)	Sonderegger, R. Personal Communication, August 29, 1978.
(146)	Belfield, W. Personal Communication, June 12, 1978.

APPENDIXES

APPENDIX A

DEFINITIONS RELATED TO SPEECH SCIENCE

- <u>Articulation Index</u> A weighted fraction representing, for a given speech channel and voice condition, the effective proportion of the normal speech signal which is available to a listener for conveying speech intelligibility. It is computed from acoustical measurements or estimates, at the ear of a listener of the speech spectrum and of the effective masking spectrum of any noise that may be present.
 Allophone - A manifold acoustic variation of a phoneme.
- <u>Coding</u> The means by which an analog waveform is discretized then further represented in one of the well-known methods, e.g., Pulse Code Modulation.
- <u>Cognate</u> A complimentary pair of fricatives. One is voiced, the other is unvoiced; however, the place of articulation is the same.
- <u>Consonant</u> Those speech sounds which are not exclusively voiced and mouth-radiated. There are fricative, stop and nasal consonants.
- <u>Communication</u> The means by which any transmission, emmission or reception of signs, usages or intelligence of any nature is conveyed.
- Excitation Function The representation of the glottis in the vocal tract by mathematical modeling in the synthesis of voice.
- Formant The resonance component of a speech sound. Generally, it is associated with the phonetic quality of a vowel.
- 9. <u>Fricative</u> The speech sound produced by a noise excitation of the vocal tracts. This noise is generated by turbulent air flow at some point along the constriction in the vocal tract. If the vocal cords operate with the noise, the fricative will be voiced; otherwise, it is unvoiced.
- 10. Glide (liquid) Those sounds characterized by gliding transition of

162

the vocal tract and influenced by the context in which the sound occurs, commonly referred to as semi-vowels.

- 11. <u>Glottis</u> The orifice between the vocal cords.
- Intelligibility The perceptual effect of understanding speech sounds.
- 13. <u>Language</u> The set of principles mastered by the speaker in which resides at his grasp an infinite set of sentences. It is a system of human communication based on speech sounds used as arbitrary symbols.
- <u>Nasal</u> The group of consonants made with complete closure of the mouth making the radiation sounds come from the nostrils.
- Phoneme The basic speech sound element used which serves to keep words apart.
- 16. <u>Plosives (Stops)</u> Those speech sounds which begin with complete closure of the lips. The lungs build up pressure behind the closure, suddenly release an explosion marking the voice onset time.
- 17. <u>Pitch</u> The difference in the relative vibration frequency of the human voice that contributes to the total meaning of speech, the fundamental frequency.
- 18. Quality The ability to identify the character of speech sounds.
- 19. <u>Place of Articulation</u> The part of the vocal tract where constriction occurs. Three places of articulation are: labial, alveolar, palatal; i.e., front, middle, and back of the mouth.
- 20. <u>Speech Perception</u> The ability of humans to discriminate and differentiate speech sound with their over-learned senses.
- <u>Suprasegmentals</u> The features of stress, pitch, intonation, melody,
 etc., that occurs simultaneously with speech sounds in an utterance.

- <u>Transitional Cues</u> The loci of frequency determined by the place of articulation connecting phonemes.
- <u>Unvoiced</u> Speech sounds that occur without the vocal cord source operating.
- 24. <u>Vocal Tract</u> The acoustic tube which is nonuniform in cross-sectional area beginning with the lips and ending with the vocal cords. For the adult male, it averages 17 centimeters in length and varies from 0 to 20 square centimeters in cross-section.
- <u>Voiced</u> Speech sounds that are produced by the vibratory action of the vocal cords.
- 26. <u>Voice Onset Time</u> The delay from complete closure of a plosive to the beginning of voicing. Generally averages 25-30 milliseconds.
- <u>Vowel</u> Speech sounds produced exclusively by the vocal cord, i.e.,
 voiced, excitation of the vocal tract.

APPENDIX B

COMPUTER PROGRAMS FOR CODING THE PREDICTION RESIDUAL The flow graph in Figure 24 gives all the programs used in this thesis. These programs were coded for the INTERDATA 70. Each of the modules is discussed below.

B.1 DIGITIZ

This program is an implementation of analog-to-digital (A/D) conversion. It actuates the equipment and the A/D converter which is a part of the computer system. The input is from an analog tape recorder. The output corresponds to the quantized signal with amplitudes of ± 10 volts peak-to-peak in steps of 20 millivolts. This data is stored on disk in area BURGE.DAT. The samples are grouped in 16 records sequentially with 256 samples per record.

B.2 LOOK

This program operates on any data set. It was developed as an information tool for scanning the data. It has an option to have the output on a CRT or on a line printer.

B.3 WINDOW

This routine uses a 256-point Hamming window, shifts by 64 points, uses a 256-point window, and the process is continued to the end of the file. The input is the sampled speech data, BURGE.DAT. It conveniently informs the user that the sequence is being windowed. The output is scaled, windowed data that is written in file WINDOW.DAT.

B.4 AUTO

This program calculates the inverse filter coefficients, crosscorrelation coefficients, partial correlation coefficients, and auto correlation coefficients. The input is the windowed speech data. A sample of each of the coefficients is printed. They are written in the file AUTO.DAT.

B.5 INVERS and LATTIC

These routines are in implementation of the lattice filter. The input is the windowed speech data. The output is the error signal or the prediction residual. This output is written in the file RESIDUAL.DAT on the disk.

Routine LATTIC provides the user the option of a plot of each frame for the input speech and the prediction residual. The user must also enter the two character names of the sound for the frame desired.

B.6 FFTMGR

This program calculates the Fourier spectrum of the speech input. It calculates the energy per frame, splits this into the predetermined sub-bands for sub-band energy, and it computes the normalization factor and the bit allocation. It uses as input the residual signal and outputs the spectrum, phase and bits. These are written in the files SPECTM.DAT, PHAZ.DAT and BITS.DAT, respectively.

B.7 ENCODE

Routine ENCODE codes a signal based on bits allocated. It uses uniform quantization using the adaptive strategy discussed in the main
part of the thesis to determine the number of levels and rounds the in**dividual samples** to the nearest level. The input is the number of bits **and the sub-band** signal. The coded signal is written in file CODE.DAT.

B.8 DECODE

This routine decodes the integer data in the file CODE.DAT. It determines the largest code level and calculates the allocated bits from this level. It also sets a maximum quantization level. This decoded signal is written in the file SIGNAL.DAT.

B.9 SUMLPD

This routine computes the sub-band prediction residuals using the digital bandpass filters, modulator, lowpass filters, and decimator. The inputs are the signal spectrum and phase. The outputs are the decimated sub-bands. These are written respectively in the files SBAND1.DAT, SBAND2.DAT, SBAND3.DAT, and SBAND4.DAT.

B.10 RESULT

This routine uses the signal to compute signal-to-noise (SNR) ratios. It uses as input the decoded signal and the residual signal. The output is a normalized SNR and an average mean squared SNR. The user has the option of producing a plot. If used, one must input the twocharacter sound names. The data is written in the file SQNR.DAT.

B.11 SNRCAL

The routine calculates from any two 256-point data arrays the SNR. The input is two arrays of length 256 or less number of points. The program outputs a mean-squared SNR, a root-mean-square (RMS) SNR and a conventional (normalized) SNR.

B.12 PITCH

This routine estimates the fundamental frequency of a speech utterance. The input is the speech array prediction residual signal and the spectrum of the signal. The program outputs the pitch.

1		IMPLIGIT INTEDER+2 (I-H)
2		THTFREP#2 MEFF(12608), PLNK, NENE
3		DATA F. O. M. N. LE/S. , 25 " . X" FFFF"/
4		DAIR HHF/221/
5		NPITE(5, 160)
6	1 (94)	FUMPHATCE HATER TRIDGER VALUE THRESHOLD FOR BACKUP ">
7	384	FOR INTCISIONAL HERE FOR SPECIFIC IDENTIFICATION >
Ð		CHLL INFUT(5,F,O)
	1 899	WEITE (5, 3AA)
		FEAD(1) 111) NHIE
11	111	FORMAT(182)
12		M=1F1X(F+, 5)
13		N+I1+32
4		L=1FIX(G+. 5)+32
15		H#1(F(5,1333)
6	1333	FORMATC'ENTER CR TO START CONVERSION /
17		1 TOR "FIRE" CR TO STOP PROGRAM")
8		READ(5, 114)1
9	110	FORMAT(181)
214		1FCI NE PINKOOD TO 2000
21	30	CALL A15(4)(9, 1, 1, N, NSTAT)
22		JF (N I T H) 60 TO 30
23		CHLL DIG112(BUFF(1), BUFF(12000), 1, 8, 1ER)
24	18	IF (IFP. F() 0) 60 TO 10
24		WPITE(5, 269)
26	299	FORMAT('DONE')
27		DO 15 1=1.7900
28		IF(PUFF(1) 0T.L) 00 TO 25
29	15	CONTINUE
30		60 10 88
31	25	1=1-54
32		IFAT LE BOI-L
3		EVERALDAVIENE
-		1-110-1+4645
35		DO 20 K=1, TEND
36	28	HFF(K)=HFF(K)/32
37	26	CALL SYSIN(56, L, IST, ISTDEV, BUFF(I), BUFF(IEND), 2, 0, 0)
R		IFCISTORY FO 0000 TO 1010
34		CHLL LOFPRCISTORY)
69		PRUSE 1
41		60 TO 26
42	1914	FORMATCIAZ
47	1.01.0	HRITECS, 10142 NOME
		C/1 1011 Jul. 1841 3

- 43 1011 WPITE(2, 1012) FUFF(J), BUFF(J+1), BUFF(J+2), BUFF(J+3), BUFF(J+4) 46
 - +, ##FF(J+5), M#F(J+6)

LOOK INPLICIT INTEGER+2 (I-N) DIMENSION REPRY(256) INTEGER+2 BUFF(128) 1 2 3 ERUIVALENCE CARRAY(1), BUFF(1)) 4 HRITE(S, 111) 5 10 6 111 FORMATC' INPUT SEQUENCE NUMBER TO BE DUMPED' > FLAG=0. A CALL INFUTC-5, SN, FLAG) IF CFLAG GE 1, A) GO TO 350 NS-IFIX(SN+00. 5) 10 11 1F(NS. LE. 0) 00 TO 200 12 13 14 15 IFEC+32+(NS-1) IFEG+IFEC IENO-IREC+31 REHIND 1 REMIND 1 CHLL SYSIO(92, 1, IER, ISTD, BUFF(1), BUFF(128); 2, IREC, 8) CHLL IVERK(ISTD) IF(IEEC, EQ. ISEO) HRITE(6, 141) BUFF(1) FORMATCHA, 'SEQUENCE IS ', 1R2, /, 1X) HRITE(6, 131) BUFF FORMAT(1X, 2016) 20 1.6 1.7 1.8 1.9 20 141 21 131 IFCIREC. EQ. 1END> 00 TO 10 22 23 IFFC=IREC+1 24 25 26 27 GO TO 20 00 10 20 PENIND 1 READ(1, END=10) ARRAY WRITE(6, 380) ARRAY FORMAT(1X, 10F12, 3) 350 28 29 380 60 10 360 30 200 STOP EHD

47.	1012	FURMATC1215	n agus an			
48		00 TO 1000				
49	80	HRITE(5. 120)	•		· .	
58	1.20	FORMATC' NO S	TONME	FOLND	RETRY	>
51		00 10 1968				
52	2000	ENDFILE 1				
53		STOP				
54		END				

7 8 .

DIGITIZ

DIGITIZ

POUT THE MITHEORY DATA BY OVERLAP AND HETHOD 2 C C 3 MELTIFN BY J. PFERMILL MED L. BURGE 4 С INFLICIT INTEGER+2 (I-H) 5 4 • DIFENSION H(64), DATA(256) 7 THIRSEN +2 INCHI(256) FAILVHEFICE (DATA(129), INDAT(1)) 8 5 5 WHITE(5, 111) 10 111 FORINTC'DATA SEGRENCE TO NINDONED ?") CAL INPUT (-5, SH) 11 NELIFIX(SHI+8. 5) 12 13 18EC-32+(H5-1) 14 IFEN-IFEC 15 IFND+1FEC+30 15 51.1 -6 828 17 THEP1-6. 283185 14 C 14 U 19 C 20 C 21 22 23 READ IN THE DATA FFHING 1 FFHIND 2 DO A 1=1.64 24 8 HOIDLA A 25 26 27 28 29 30 31 15 CALL SYS10(92, 1, 1ER, 1510, INDAT(1), INDAT(256), 2, IREG, 0) IF (1510 EQ 0) 60 TO 16 CALL IOFRACISTD> PRISE 1 60 TO 5 16 IF(IFEC. GT. 18E0) 00 TO 100 C 32 r, MEPORT THE SEQUENCE WE ARE ON 33 C 34 35 36 37 11 INTAICED=INDATCED/4 1HL-HT/1.>=1HL-HT(1.)+128 WPITE(6, 112) HS, INDAT(1) FOFMATCINE, "SEQUENCE NUMBER ", 13," IS ", R2) 112 38 110061(1)=8 39 С SCALE THE DATA 44 Ċ . 41 С 42 43 DO 18 1=1, 256 18 DATA(I)=FI.OAT(INDAT(I))+SCL 44 С 45 C WINDOW THE FIRST BUFFER

45 C

00 28 1-1. 256, 64 H=0 J=1+63 00 28 K-1.J PN-FLORT (K-1.)/255. 8+THOPI N=N+1 H(N)+H(N)+8. 56-8. 46+C05(PN) 20 DATACK)=DATACK)+W(N) 60 TU 300 C 1.00 IFCINEC EQ IEND) OD TO 158 00 110 1+1, 256 110 DRTACID=F1_DRTCINDATCID>+SCL PO 120 I=1, 256, 64 N=Ø J=1+63 DO 170 K-1, J N=H+1 120 DATACKS=DATACKS=HCN) GO TO 309 150 DO 160 1+1, 256 160 DRIACIS=FLORTCINDATCISSEL DO 188 1=1, 256, 64 N=0 N1.+65 J=1+63 00 180 K=1, J N=H+1 N1=H1-1 DRTR(K)=DRTR(K)+H(N) PN=FLOAT (257-K)/255, 8+THOP1 78 -1.80 H(H1)-H(N1)-0. 54+0. 46+COS(PN) C С WRITE OUT THE BUFFER C 300 CALL SYS10(55, 2, IER, 15TD, DATA(1), DATA(256), 4, 8, 8) IF(ISTD. EQ. 0) GO TO 310 CALL IDERRCISTD> PHUSE 2 60 TO 5 310 IF (IREC. EG. IEND) OO TO 400 IREC=IREC+2 60 TO 15 468 STOP END

7

47

48

49 50

-51

52 53

54

59

69

61.

62

63

64 65 66

67

68

69

78

79

80

81.

82

83

84

85

86

87

88

89

90

91

WINDOW

WINDOW

1	C	SURPORTINE AUTO
2.		IMPLICIT INTERER+2CI-N>
3.	C	PROGRAM CH.CU.ATES INVERS FILTER COEFFICIENTS BY
4	C	LEVINGSON'S KEIHOD
5	C	N - NO. OF FOINTS
6	51	X - YEGTOR OF SAMPLED SPEECH PTS
7	C	H - AFDER OF INVERS FILTER
A	C	A - VECTOP OF INVERSE FILTER COEFFICIENTS
9	C	ALFHA - VECTOR OF CROSS CORRELATION COEFFICIENTS
18	C	RC - VECTOR OF REFLECTION COEFFICIENTS
11	C	R - MITOCUPPELATION COEFFICIENTS
12	C	REPRY - VEGTOR TO HOUSE ALL COEFFICIENTS CALCULATED
13		DIMENSION R(11), X(256), A(11), ALPHA(11), RC(11), AKRAY(44)
1.4		EPUTVH ENCE CARRAY(1), K(1)), CARRAY(12), K(1)),
15		+(AFFAY(23), ALPHA(1)), (ARRHY(34), RC(1))
16		N=256
17		115=9
18	С	FERIN WINDOWED DATA
19		FENIND 1
24		FFWIND 2
21	5	FF.FD.(1, ENG=99) X
22		H=19
23	11	MP = M + 1
24		DO 18 K = 1. MP
25		F(K) = A. B
26		L = N - K + 1 .
27		DO 10 NP - 1.L.
28		MFK =NP+K−1
29	10	P(K) - F(K)+X(NP)+X(NPK)
7A		H(1) = 1.
31		H(PHH(1) = P(1)
32		1F(11 F(), 0) BO TO 60
33		R(C1) = -R(C2)/R(C1)
34		A(2) = FC(1)
35		ALPHA(2) = R(1) + R(2)+RC(1)
36		1F(M, FQ, 1) GO TO 60
37		D0.49 MINC = 2, M
38		5 = A B
39		$p_0 20 IP = 1. MINC$
40		MNCMING-1P+2
41.	20	5 = 5 + R(MRC)+ACIP)
42		Pr(MINC) = -52ALPHA(MINC)
43		MH = MINC/2 + 1
44		60-30 IP = 2/MH
45		IF = HINC - 1P + 2
45		AT = A(IP) + RC(MINC)+A(IB)

47			ACIES & ACIES + ACCHINCIARCIPS
	78		A(IP) = AT
49			RONTHC+1) = PECHINCS
			ALEMACHINCHIS - ALEMACHINCS-ALEMACHINCSARCONNCSARCONNCS
51			IF (REPHACHINC)) 58.58.40
52	40		CONTINUE
53	58		H = MIKC - 1
54	68		CCRIT 100 F
55			HEITE(2) REPRY
56			M1 = M + 1
57		•	WRITE(5, 101)
58	101		FORMAL (1HO, "AUTOCORRELATION TERMS")
59	1.04		FORMAT (1HA, SE16, 5/SE16, 6)
64)	•		WRITE(5, 100) (R(1), 1=1, M1)
61.	1.82		FORMATCHHO, 'INVERSE COEFFICIENTS')
62			HEITE(5, 102)
63			HRITE(5, 100) (A(1), 1=1, M1)
64	1.03		FORMATICING, CROSS CORFELATION TERMS()
65			HRITE(5, 103)
66			HRITE(S, 100) (ALPHA(I), I=1, M1)
67	1.64		FORMATCING, 'REFLECTION COEFFICIENTS')
68			PRITE(5, 184)
69			WRITE(5, 100) (RC(I), I=1, M)
76			NS=N5+1
71.			HRITE(5, 119) NS
72	119		FORNAT (1H9, "PRESENTLY CALCULATING AND PROCESSING
73		+	CORRELATION DATA FOR SPEECH SECTION , 13)
74			60 TO 5
75	99		STOP
76			END

AUTO

AITO

1		IMPLICIT INTEGER+2(I-N)
2	С	IMPLEMENTATION OF LATTICE FILTER BY L. L. BURDE
3	C	A1 - INFUT OF SPEECH SAMPLES
4	c	H - HO, OF POINTS
5	C	N - DRUFR OF FILTER
6	C	PG - FEFLECTION COEFFICIENTS
7	С	PO - RESIDUEL DE SPEECH PTS OUTPUT
8	C	B - VECTOR OF BACKWARD PREDICTED SAMPLES
9	C	C - TEHPORARY VEGTOR FOR SAMPLES
1 A		DIMENSION (11), A1(256), A(11), ALPHA(11), RC(11), ARRAV(44)
11		DIMENSION C(11, 255), B(11), SI(11)
12		DIMENSION #0(255)
13		E(4) YALFIFF (AFFAY(1), R(1)), (ARRAY(12), A(1))
14		FYHIVALENCE (AFRAY(23),ALPHA(1)),(ARRAY(34),RC(1))
15		M=19
16		H=256
17		115 = 6
18	C,	FERD HINGONED DATA
19		
214		FFHIND 2
21	-	PEWINE 3
22	5	FFAD(1, EM)=99) H1
23		PEND(2) HERMY
2		H(1) = H, H
2		
<u></u>		15/2 (T 4) 6/4)-64/2-4)
24		DATE OF THE DESCRIPTION OF THE D
29		
-		SI((+1)=SI(J)+R((J)+R(J)
31		C(J+1,K)=B(J)+BC(J)+SI(J)
32	•	IF (K. FR. 1) 60 TO 20
33		H(J+1)=C(J+1, K-1)
45		(H) TO 59
35	20	F(J+1)=B. A
36	54	RO(Y)=51(3+1)
37	30	CONT IN JE
38	49	CONTINUE
39		WEITEGO RO
49		M1=1+50
41		WRITE(5, 100) (RO(1), I=1, M1)
42	100	FORMATCING, PESIDUAL VALUES FROM LATTICE FILTER'
43		A(19F12 6))
44		N5=N5+1
45		WPITE(5, 560) NS
46	- 90	FORMATCING, PRESENTLY PROCESSING DATA THRU LATTICE FILTER

1		SURROUTINE LONPAS (DEMOD, IRAND, I, NB, L, FRSE)
2	C	
3	C	CONFUTES IDEALIZED LOHPASS FILTER
4	C	
5		INPLICIT INTEGER+2(I-N)
6		INTEGER+2 IBRID(1)
7		FEAL MAG(256)
. 8		DIMENSION DEMODICED, Y(256), FRSE(1)
9		IZERO=256-IERAD(I)
1.0		IPASS-IBAND(I)
11.		WRITE(5, 100)
12	1.00	FORMATCING, "ROUTINE LOWPASS PROCESSING")
1.3		CALL FEFT (DEMOD, Y, NB, L, 1)
1.4		DO 20 K=1, 256
1.5		MAG(K)=DEMOD(K)+DEMOD(K)+Y(K)+Y(K)
1.6		FHSECKD=ATAN2(Y(K), DEMOD(K))
17	1.40	FORMATCIN / REAL INAG MAG PHASE - 1,4E15.4)
19	-	DEMOD(K)=SQRT(MAG(K))
1.9	20	CONTINUE
20		WRITE(5, 140) (DENOD(K), Y(K), MAG(K), FRSE(K), K-1, 256, 32)
21		DO 30 K=1PASS, IZERO
22	30 .	DEMOD(K)=0. 0
23		RETURN
24	-	END
25	C	
26	C	

LOWPAS

47		•	SPEECH	SECTION	*, 15>
49 50	9 9		STOP STOP		

INVERS

INVERS

ENCODE

ENCODE

	1 E 2 C	SUBROUTINE ENCODE	47	1.00	MRITE(5, 100) (IKODE(K), K=1, N) FORWIT(1H , 'ENCODED SIONAL ', 014)
	3 6	MULTINE CLASS A SIGNAL RASED ON ALL MONTELL BITS	45		1F(11. EQ. 5) 00 TO 99
	4 C	USES UNIFORM QUENTIZER, WRITTEN BY I I BUNDE	50	•	00 TO 1
	5 C	the second s	51.	99	HR17E(5, 999)
	2	1001 1011 1016066-201-00	:12	399	FOSMAT (1H1/// AAAAAAA DONE AAAAAAA')
	7	THIFGERAD KODE (20) THORE CORES	117		STOP
	A	DIMENSION DEFISION BITCH	H.A.		FNO
	4	DATE MOLT / 2 3 4 4 5 5 4 6 6 5 6 6 6 6 6 6 6 6 6 6 6 6		C	
		10 10 10 10 10 10 10 10 10 10 10 10 10 1		ž	
		*15, 27, 23, 27, 23, 24, 25, 26, 27, 28, 29, 38, 31, 32/		•	
	1				
	<u> </u>	11-235			
	3	60 19 I-1.N			
	• ·	DF7.(13=4, 4			
1	5	45+ C(1)= h , a			CHRKEN
•	a 18	1KGGE(1)=0			
	7 C				
1	9 C	pero data			
1	y C				
2	•	PE 43247 1			
2	1	Prutito 2			
2	2	BENJUA T			
2	3	De utor a			
2		56 U (10) 2			
	2 ·	FERING P			
	2		4		SARKATINE CHERENCE, SEN, IN, ESUB)
		PETPES 1100 (PETS(3), J-1, 4)	2	C	THIS PROGRAM CHECKS FOR HIGH ENERGY VS LOH ENERGY SCANOS
		PREMITCH , ALLOCATED BITS /FS. 2, FOR SUBBAND ONE /	3	C .	WRITTEN BY L. L. PUPGE
		THE ZA FOR SLARGAND THE 2, FOR SURBAND THREE /	4	C	IF PREVICES ENERGY HAS MINIMUL THE ENERGY IS RELACED
3		+F5. 2, * F(R. SJ&RHAD FOLR*)	5	C	E - VECTOR OF ENERGY POINTS
τ.	1 1	PEAD(11) DEC	6	C	SEN - VECTOR OF TOTAL ENERGY FOR N PTS
3.	2	DO 30 1-1, N	7	C	IN - FRAME MARKER
3	3 30	(DECCI)-DECCI)		- Č	ESHR - SAR-BARD ENERGY
3	4	£1T=8175(11)	ä		IMPILICIT INTEGERAZ(I-N)
3	5	HPITE(5,120) (DEC(I), I=1, N)	4 4		DIMENSION FASSA, SENAA, FEIRAA
3	5 128	FORMATCH / INPUT SIGNAL (12FR 2)	4.4		
3	7 C		3.3.		
3	9 C	FIND MAXIMUM OF THEIT STOLEN			DU 10 LEU IN
3	. c	The second of the second	1.3	3.4	
4	9	CAL HEYCODEC N DIG MINN			
	1		3.7		HY 12-HY 12+75.0
	è c	54-14 T B	1.6		1F (HV1L, 61, E(256)) 60 TO 30
			1.7		E(256)=E(256)/16. 0
		CRIL INC MANYI IZER	1.8		DO 20 1-1,4
			19	20	ESURCID=ESUBCID/16.0
		CHEL RUNT IZ (DEC, BIT, ADEC, B10, KODE, IKODE)	20	38	RETURN
- 4/	5	WRITE(8) IKODE	24		E MA

GUNT 12

GUNTIZ

N.ROE.

		SUPPORTINE GANTIZ (DEC. BITS, ODEC, BID, KOOF, BUMAR)					
5	C, -			47		00 10 38	
3	- C C	POITTINE RUNNITIZES AND CODES STONAL BY L L BURDE		48	60	CONT INNE	
5	-	1041 1CTT THIS GERARIC LAWS		50	34		
6		TATEGERS THIS FALL MANAGES		- 343		RE LURI	
~				31	•	END	
ė		NICE IS INTERACIONAL CONTRACTOR DEC(1), DEC(1)		52	-C -		
ž				53	C		
	-	0 3 14, 33		34	C		
1.4		(32 4(1)=0		55	C		
33	6						
12	C	POLAD BITS TO INTEGER DETERMINE NO. LEVELS					
. 13	-C					· ,	
14		RDIND=BITS+1.0					
. 15		IBITS+IFIX(R(AND)					
16		1044 - 20018115					
17		1(HH_L + 1)HR +1					
18		1F(19H, 6T 32) 10H - 32					
19		Geta and Contractions				CLAMP	
- 29	C						
21	ē						
22	r	PATHALITY THE REPERT IN ENCH LEVEL					
22							
		wire.= 12. 00+1132/2014_					
2.4	-	104 11-1044.72	•				
- 27	C						
26	C	SET VALUE FOR LEVELS				•	
27	r,						
28		10 10 K=1, 10H H				SUBSCHITTING OF AND CONFUL T ODEC K DEC THOSE HAS	
- 29		K1=1(48_1)-K+1		- 5	C	PORTON TIME CLIMIT CONTROL OF GUTLINE DECI IKUDE, KDE	H.)
30		K2=101A-K+2			ř.	CLEMES STOLEN TO DESCRIPT LOURS ON A	
31.		PHEH(K2)-FLOATCK1)+RING			č	ACTURS STORAT TO DEDKEST TRAFT BY T T	1,161
32		PREW(K)FLOAT(K3)+01NC			6		
33	10	CONT 114 JE				INPLICIT INTEGER+2(I-N)	
34		WRITE (5.110) IBITS THE ATM BTO		2		INTEGER42 INODE(1), KODE(1)	
35	110	FORMHIGH , (ABANT) ZATION DATA (AM ATTS FOR CONTIN				DIMENSION QUECCL, QNEWCL, DECCL)	
26		+ AVAILAPLE - 16/ MANEED RELEVELE - TENTS FUR CODING				CHICI = ABS CANENCI+1.>-DEC(K)>	
37		+ VHILE 1. EG 32 MAY UNLE OF FLOWL A DA THURSTENT				CHK2=ABS(DEC(K)-GNEH(J))	
18		LITER ANAL CONDUCTION OF STORAL SPEC.3)		1.6		IF(CHCLLE CHC2) GO TO 10	
79	100	ECHACTARY (MICHIER CARE), INTEL		1.1		GDECCK)=@#EWCJ)	
	-	FORTHER AND CONTIZED INCREMENTS -74F10.3)		1.2		IKODE(K)=KODE(J)	
	а. С			1.3		GO TO 20	
	u .	MORED SIGNAL TO REAREST LEVEL		3.4	1.0	GDEC(K)=QNEH(J+1)	
42	C			1.5		1KODE(K)=KODE(J+1)	
43		DO 39 K=1.N		16	28	RETIEN	
44		1/3 60 J=1, 1000		17		FND	
45		IF (QNENCI+1), I.E. DEC(K)) GO TO 60		18	c		
46		CALL CLAMP (OHEN, J, ODEC, K, DEC, IKODE, KODE)		19	č	•	
					~		

1	C	SURFOUT THE DECODE		47	180	FORMAT(1H , DECOVED SIGNAL ', 8F7. 3) 15(11, 59. 5) 60 TO 99
1	č	MUT WE RECIDES COOPD STONE WEITTEN MUT		49		00 TO 1
1	č	NOTTINE TECTORE CONTRACTORISTICAL PER LA T	- ENARCHE	50	99	5100
5	•			51		END
6		101610+2 KOLA (256), 1KODE (32), KODET (256)		52	C	
7		DINEUSION STONAL (255)	•	 53	ē -	
Â		DAIA 18062/1, 2, 3, 4, 5, 6, 7, 8, 9, 18, 11, 12, 13, 14, 15,	16.17.18.		-	
9		+19. 29. 21. 22. 23. 24. 25. 26. 27. 28. 29. 30. 31. 32/				
18		N=256				
11		11=1				
12	C					
13	c.	FEAD DATA, FIND MAX IN ARRAY				
14	С					
15		FFHIND 1				
15		FEWIND 2				
17	1	PERMIT FIRMAND KODE				MAX
1.8		11-13+1				
14		DO 28 1-1, N				
29	29	KOU+1(1)=KOU4(1)				
21		CALL NEXTRUCT, N. 1810, NM)				
22		E16-FLOHT(1616)				
2.7		PFI1E(5, 119) 1810, MM, N				
24		WPI(E(5, 120) (KGDF(1), I=1, N)				•
25	129	FUENAICIH / CODED SIGNAL (1815)				
25	110	FORMATCIH , VALUES FROM MAX PROORAM', 316)				
27	C					
28	C	SET LEVEL FOR INVERSE QUANTIZER		1.		SUBROUTINE HAX(KODE, N, 1810, NUN)
29	С			2	C	
30		(X) 19 INC=1, 16		3	С	FINDS MAX IN ARRAY
31		101/6=114.+2+1		- 4	C	
72		IF(IBIG FO 1000) HITS=3 32193+AL0010(BIO)		5		IMPLICIT INTEGER#2(I-N)
33	10	IF(IBIO_FQ_ING+2)_BIT5=3, 32193+ALO010(BIO)		6		INTEGER+2 KODE(1)
34		IF(BITS GT. 4. 8 AND BITS UT. 5. 2) REEVEL-10. 0		7		tt1.=N-1.
35		JE(BITS 61.3 8 AND BITS I.T. 4.2) QLEVEL=5.8		. 8		DO 20 I=1.NL
35		IF (BITS, GT 2, 8, AND, BITS, LT, 3, 2) GLEVEL=2, 5		9		IF(KODE(I), GT, KODE(I+1)) GO TO 18
37		IF (BITS. OT. 1. A. AND BITS. L.T. 2. 2) QLEVEL=1. 25		1.0		1816-KODE(1+1)
78		IF (BITS, GT., 8 HUD, BITS, LT. 1. 2) GLEVEL=0. 625		1.1		₩#= I + 1.
39		IF (BITS, GT, 5, 0) (ILEVEL=1.0, 0		1.2		GO TO 20
49	-	1F(BITS, LT. 1. 0) RLEVEL=0, 625		1.3	1.0	IB1G=KODECI)
41	ç			1.4		KODE(I+1)=1BIG
42	C	CALL DECODER		1.5	20	CONTINUE
43	C			1.6		RETURN
44		CHLL UNCODE (OLEVEL, BITS, KODE, N. SIGNAL, IKODE)		1.7		END
45		HRITE(2) STONAL		1.8	C	
45		WPITE(5,169) (SIGNAL(K),K=1,N)		1.9	С	

ORCODE

DECODE

内側のしして

RESULT

.

	C	SPACETTINE RESULT	17	C	
	č		48	C	INTERPOLATINO SIGNAL TO UP SAMPLE
-	Č.	ROUTINE CREATES SHR WRITTEN BY L. L. BURGE	49	С	
4	r i		56	•-	HR11E(6, 900)
	-	1001 1017 1016/08/201-00	51	900	FORMATCINE (PROCESSING SIGNAL ()
ē		INTEGERAZ DEUCAL, IRUCAL	52		HI-THOP1+FKEG+ISHCIFGHD)
-		DIKENSION STUDIEL (256), SPER (256), STHED(256), REM(256)	53		CHIL FFFT(SPEAL, SIMAG, NB. L. 1)
÷.		DIRENSION FRONT (256), VERCIT(256), SNR1(258), PHOT(256)	54		PO 19 I-L.N
		DINE IS THE FUEL (DEAL) CHERICOLD COUNTY ON A COUNTY OF	55		SHAGSO(1)+SREAL(1)+SREAL(1)+SIMAG(1)+SIMAG(1)
•	•	511511747 CHILL COLOR & WORLD WORLD THE CHILL CHILL OF	56	18	\$H66(1)=568T(568650(1))
		DUBUSION REFUSED DATACEST TURCESED. TTTECO	 57		WEITE(6, 940) (SREAL(1), SIMMO(1), SHAOSD(1), SHAO(1), 1-1, 120)
		FOUTVE FILE (FHAT(1), SUE1(1), DATA(1))	58	945	FOSTAT(1H . SFAL IMAD MADSO MAD . 4F11 3)
17		FOUTURE HAVE (SUBG(1), W(1), PEN(1))	59		MOVE TEHL LEBUD
		FRITYELLARE (STONE) (1), SPEAL (1))	60		ISTOP-TEAND+1
15		ELATIVE FINE CENSOS((1), SPECTM(1))	61		12-1FH(15TOP)
		FORTHER HATE (EXCTT(1), PHAT(1))	62		172-14-12
17		FORTHER FORF (STRAG(1), FETR(1), TIME(1))	63		DO 36 J-1, 128
			64		J1=Mi/VE+J-1
		COTO DELLANDA 213 213 1027 10176 26 52 55 1067	65		SREAL CIDESREAL CITS
20		TETE TITLE // 400 / / 500 / / TITLES// CIM / / 500 - / /	66		510(4)(J)=510(4)(J1)
2.			67		J2=H-J+1
2		THE TILL 2022	68		SEER (J2)=SREAL (J)
~~		(HTT 1-5 2532	69		51H6G(J2)=51H6G(J)
5.		140-0	79	38	CONTINUE
2			71	•	D() 40 1=12, 122
26		ALLELA H	72		SPEAL (I)=0. 0
27		FFF0474 28	73		51060(1)=0.0
53.		1660-1	74	48	CONTINUE
24			75		DO 45 I-LN
10			76	45	SHAG(I)=SREAL(I)=SREAL(I)+SIMAG(I)+SIMAG(I)
			77		HPITE(6, 915) MOVE, 12, 1STOP, 122
32		VENTITIINA A	78		NRITE(6, 925) (SHAQ(K), K+1, N)
	č	Tracticity-a. a	79	915	FORMAT (1H + 416)
24	č	PEND DATA	80	925	FORMATCHH / FREQ INT SIGNAL (2F12 4)
24	ž	FERT WITH	81	C	
25	•	PEUDA 1	82	C	NODULATING SIGNAL BY COSINE WAVE
37		BENTAN D	83	č	
		PENDING R	84		HELTE(6, 910)
-		FELIN A	85	910	FORMATCINI, THORALATING SIGNAL ()
40		FEUDO 2	86		DO 50 KHL N
41		FERICE FUD-991 STONE	87	58	REHCK)=COS(NN+K+ANG)
42	*	LOTTE/C. 490 \	88		CALL FEFT (REM. REIM. NB. L. 1)
42		ENDURTING & ANAL BOUTINE DEEL T PROFESSION ANALAS	89		D(1 55 K=1. N
22		ADDRESS OF THE STORE STORE STORE STORE	90		SPEAL (K)=HEM(K)
		LIDITE/E 43D) /STONW (1) Tel N)	91		SIMBO(K)=REIM(K)
46		FORMUTCH / THEIT STONE / RE7 3)	92		REIH(K)=REN(K)+REH(K)+REIH(K)+REIM(K)
	300	FURNALLIA / LARVI SLUAR, JOF7, 37			

RESUL T

RESULT

93	-55	REN(K)=REIN(K)+SHAB(K)	
94		WEITE (6. 950) (REM(K), KH1, N)	
*5	اغر 19	FORTHING H . MICHAATED SIGNAL ", 8F7. 3)	
٩.,		11=1E:n1/0+1	
97		that=JFH(IFH)	A
95		168/1-1E4/(11)	
99		DO EG K-THU, THUI	
163		EXCLT(K)+SPEAL(K)	
141		SE2CIT(P)=STMAG(K)	
182	68	CONTINUE	
167		HEITECE, 966) (EXCITCK), YEXCITCK), K=1, INN1)	
1 44	964	FORMALCH , 'EX 510', 8F9. 3)	
145		IFAIIIFAND+1	
185		TECTEMUN NE. 5) GO TO 1	
197	59	CONTINUE	
183		CHIL FEFT (EXCIT, YEXCIT, NB. L1)	
1.9.2		DO 150 1+1.N	
118		EXCITCID=EXCITCID/256.0	
111	158	YEXCIT(1)+YEXC11(1)/256.0	
112		NETTE (6, 980) (EXCIT(K), YEXCIT(K), K=1, N)	
117	988	FUERALCH , TIME STONAL (, 4E15, 3)	
114	С		
115	C	FEAD IN ACTUM, SIGNAL	
116	C		
117		FERD(3) SPECIM	
119		FEHD(4) PHAZ	
119		DO 84 K-1, N	
124		SHAG(K)+SURT(SPECTM(K))	
121		X(K)=5HAG(K)+CD5(PHAZ(K))	
122		Y(Y)=SNAG(K)+SIN(PHAZ(K))	
123	89	CONTINUE	
124		HPITE(6, 478) (X(K), Y(K), K=1, N)	
125	979	FORMATCHH , FEAL THAD ', 4F11, 3)	
125		DI) 98 K=129, N	
127		X(K)+4 H	
128		YCK)=A H	
129	90	CONTINUE	
139		HEITE(6, 920)	
131	929	FORMATCINE, CALCULATING SNR ()	
132		CALL FFFT(X, Y, NB, L, -1)	
133		DO 140 K=1, N	
134		X(K)+X(K)/256.0	
135	140	Y(K)+Y(K)/256, 0	
136		HRITE(7) X	
137		WPITE(6, 980) (X(K), V(K), K=1, N)	
	-		

1.39	C	CALL SHR ROUTINE) CALCULATE AVR SHR
140	C	
141		CALL SUR(KHAT, X, N, SNR1, SUNR)
142		DO 130 K-1, N
143		TIME(K)=FLOAT(K)
1.44		ASHK-SHK1(K)+ASHR
145	138	CONTINE
146		ASNE+ASNE/FLOAT(N)
147		WRITE(2) SNR1
148	. C	
149	C	PLOT ROUTINE
159	C	
151		NRITE(5,100)
152	100	FORMATICTED YOU WANT & SANK PLOTY)
153		READ (S. 119) ANS
154	1.10	FORMATCIA4D
1.55		1F(ANS. NE. YES) 00 TO 999
156		WRITE(S, 120)
157	120	FORMATC' INPUT SOUND NAME')
1.58		READ(5, LLA) CHAR
159		CHLL FL015(EHFF, 200, 8)
160		CALL PLOT (A. A11. 0, -3)
161		CHIL FLOT(2, 8, 7 8, 3)
1.62		CHUL SYNEOL (2. 0, 8. 0, 14, 1PHONEKE 1, 8. 8, 8)
1.63		CALL SYMEN. (999. 8, 999. 8, . 14, CHAR, 8, 8, 4)
164		CALL SYMEOL (299. 8, 999. 8, 14, "SONR ", 8, 8, 5)
1.65		CALL SYDEOR.CS. A. 7. 5. 14. TITLE, M. A. B.
1.66		CALL, NUMEFR (999, 0, 999, 0, 14, ASNR, 0, 0, 2)
1.67		CALL SYMPOLICS, 0. 6. 5 14. TITLE2. 0. 0. 8)
1.68		CALL NUMBER (595, 8, 999, 8, 14, 50NR, 8, 8, 2)
1.6.9		CHLL PLOT(8, 8, 1, 8, -3)
170		CALL SCALE(DA) 8, 6, 6, 256, 1)
1.71		CALL SCALECTINE, 7. 8, 256, 1)
172		CALL AX15(0. 0. 0. 0. 13HTIME SANPLES13.7. 0. 0. 0.
173		+TINE(257), TIME(258))
174		CHLL AXIS(A. 0, 0. 0, 214510NAL TO NOISE RATIO, 21, 6. 0, 90.
175		+DATA(257), DATA(258))
176		CALL, LINE (TIME, DATA, 256, 1, 0, 0)
1.77		CALL PLOT(9. 0. 0999)
1.78	999	STOP
1.79		END
1.80	C	
1.81	C	

1	-	SUBROUTINE SAMPLE (DEMOD, FASE, IBAND, I, NB, L, DEC)
1	2	
1	5	COPPUTES IDEALIZED DECIDITION FUNCTION
2	L.	THE TOTY INTEGERATION
ž		
ž		NTHE DE TRANSPORTE DE CARA DE CARA DE DE CARA
		DATE NEL 27.57.57.42.45.4
۰é		TENSDELICI
12	110	FORMAY (141, PONITINE SEMPLE BOUPESSING 2)
47		
14		
15	18	DEC (K) NDEM(H)(K)
16		DO 2H KATEN 255
17		
18	29	
19		1/FUA=1FU-1
29		NST=NFUH/2
21		DO 30 KHUST, NENA
22		J1=256-N51+.1
23		PHHZE(K)=FHSE(J1)
24		DEG(K)=DEMUD(J1)
25		Ja:/+1
26	30	CONTINUE
27		HP (16(5, 120)
28		HPITE(5, 130) (DEC(K), PHOZE(K), K, K=1, 256, 32)
29	120	FORMATCIN , SHIFTED SIGNAL BEFORE DECIMATION >
39	130	FUPMAT(1H , 2F10, 4, 16)
31		D0 49 K+1,256
32		XKX+DEC(K)+C05(PHAZE(K))
88	40	V(K)=DEC(K)=SIN(PHRZE(K))
34		L-7
55		CALL FFFT(X, Y, NB, L -1)
36		L±9
37		DO 59 K=1,256
38	50	DFG(K)=X(K)
39		HRITE(5,100) (DEC(K),K=1,256)
49	100	FORMAT(1H , 'DECIMATED SIONAL', 8F6. 3)
41		KETURN
42		END

\$.		SUBROUTINE SELIT (RVES, PHASE, 18H, 1, NR, L, DEMOD)
2	C.	
3	C	CONPUTES IDEALIZED BANDPASS FILTER
4	č	
5	-	IMPLICIT INTEGER+2(1-N)
Ā		INTEGER+2 IBH(1)
7		DIMENSION AVES(1), PHASE(1), DEMOD(1)
Å		DINEUSION EP(255), ¥(255), ¥(255)
ŭ		NTHENSTON PH (256)
16		The function of the states
44		LEISTLIGETAIPUCTA74 25
42		
42	104	FLENGTING CENTINE SHITT PROCESSINGLY
44	4.4949	Jat
4.5		
16		DO 10 Cal 256
47		64/2/ A A
10		
10		
1.7	10	CONTENT TO T.
20		
41		1/ERITELEN(13)
		DU AN KEIPHSS, IZERU
2.5		PHCJJ=PHHSE(K)
24		EP (J) - AVES(K)
25		,1=,1+1
26	20	CONTINUE
27		J1=256-J
28		IP6551-256-1P855
29		12EF01-256-12ER0
30		00 30 K-12EPOL, IPASS1
31		EP(J1)=AVE5(K)
32		PH(J1)=PHRSE(K)
33		.13=.11+1
34	30	CONTINUE
35		DO 49 K-L 256
36		X(K)=EP(K)+COS(PH(K))
37	48	Y(K)=FP(K)+S[N(PH(K))
38		CALL FFFT(X, Y, NE, L)-1)
39		DO 50 K+1, 256
48	56	DEHOD(K)=X(K)+COS(HN+K)/256. 0
41		WEITE(5. 118) (DEMOD(K), K+L 8)
42	110	FORMATCH , MODULATED STONAL
43		KETURN
44		END
45	C	
46	C	
	-	

SPL IT

ダウハアしだ

PITCH

PITCH

£	c		CALCULATION OF FITCH PERIOD ESTIMATION
2	C		SURROUTINE PITCH
3	C		CRUCATION OF PITCH PERIOD ESTIMATION BY L. L. BURGE
4	С		NT - VECTOR MARKING PITCH PULSE OCCUREENCES
5	С .		IDIFF - DISTANCES RETWEEN PULSES IN SAMPLE POINTS
6	С		FO - PREDICTION RESIDUAL SAMPLE POINTS
7	С		SAR - VECTOR CONTAINING SQUARE OF AMPLITUDE FOR POINTS
8	С		T - TEMPORARY VECTOR FOR STORAGE
9	С		PT - VECTOR OF PITCH PERIOD VALUES
10	C		SPECTH - VECTOR CONTAINING MAGNITUDE SQUARE SPECTRUM PTS
5.5			IMPLICIT INTEGER+2(1-N)
12			INTF6EP+2 NT(256), IDIFF(256)
13			DIMFNSION PO(256), 508(256), T(256), PT(256)
14			DIMENSION SPECTA(256)
15			AVP-9. 8
16			N=256
17			IRI_K=9
18			CDN=31, 25
19	·		DO 1 I=1,N
20			IDIFF(I)=9.8
21			PT(1)=9 0
22	1		NT(1)=4
23			TINE+0. 000125
24	-		FEHDO 1
25			FFHIND 2
26			PFHID 3
27	С		FERD SPECTRUM DATA
28			READ(1) SPECTH
29			60 29 1-1.11
30			IF(SPECTH(I), OT, SPECTH(I+1)) 00 TO 30
31			MAXSP=SPECIM(I+1)
32			FPEG-CON+FLOAT(I+1)
33			GO TO 24
34	30		MHX5P=SPECTH(I)
35			SPFCTH(I+1)=MAXSP
36	20		CONTINUE
37			PITCH=1. /FKFQ
38			WRITE(6, 900) PITCH, FREQ
39	900		FORMATCINA, "ESTIMATE OF PITCH PERIOD
40		*	PHD FURDEMENTAL
41	C		FEAD RESIDUAL DATA
42	100		READ(2, END-99) RO
43			IBIK+IBIK+1
44	C		SOLAFE DATA
45	-		00 49 Mul. N
46			SOR(H)+KO(H)+KO(H)

SORCHD=SORCHD=SORCHD WRITE(3) SOR

 47
 40

 48
 50

 54
 50

 53
 53

 53
 53

 53
 53

 53
 53

 53
 53

 53
 53

 54
 53

 55
 57

 59
 60

 61
 62

 62
 64

 65
 52

 66
 69

 69
 69

 712
 73

 77
 78

 64
 65

 83
 85

 86
 68

 89
 89

 99
 91

 HRITE(3)
 SQR

 HRITE(6,940)
 IBLK

 D0
 36

 HSL
 SQR(1)

 HSUN/4
 RV0-PITCH

 D0
 60
 L-1, N, NP

 JN-1
 JN-1
J=1.+1#2-1 DO 63 KHL J K1+K+1. IFCTCK1), GT. TCK1+133 00 TO 64 IF(T(K1) GT. MAX-T(K1+1) GD TO 63 MAX=T(K1) T(K1+1)-MAX CONTINUE WRITE(5, 910) MAX DO 52 1-1. H T(I)-SOR(I) THR. 75+MAX DO 69 KI=1.N NT(KI)+0 DO 65 K-L.J IF(T(K), GE, THR) 00 TO 66 60 TO 65 HT (JN)=K CONTINUE CONTINUE D0 67 1=1, JN 101FF(1)=IRES(NT(I)=NT(I+1)) PT(1)=FLORT(IDIFF(I))=TIME PTC=0.0 D0 70 KL=1, JN PTC=PT(KL)+PTC PTC=PTC/FLORT(JN) IF(A): (J 0010 00 00 00 IF(PTC. 4T. . 001) 00 TO 60 IF(PTC. 4T. . 02) 00 TO 60 DO 69 I=1. JN AVP=PT(1)+AVP AVP=AVP/FLOAT(JN) F8-1. /AVP IF(F8, GT. 1868. > 00 TO 68 IF(FO. L.T. 100. > HVP=. 008 F8-1/HVP

1		SUBROUTINE MRX(X, N, A10, RUM)
2	C	
3	C	DETERMINES THE NAXIMUM OF AN ARRAY BY L. L. BURGE
5		INPLUGIT INTEGER+2(1-N)
6		DIMENSION X(1)
7		N1-11-1
8		DO 29 1+1, N1
9		1F(ABS(X(1)), 0T, ABS(X(1+1))) GO TO 10
1.0		\$15=X(1+1)
11		F# #1= 1 + 1
12		GO TO 20
13	10	BIG-X(1)
14		X(1+1)-BIG
15	29	CONTINUE
16		BIG-ARS(BID)
17		PETURN
18		END

JN5-JH-1 ·		~	CINCTATY INE CIDE IN
WP.LTE.(6. 920) (NT(H), H+1, JN5)			BARNOTTING BARLED BOCKER DECKTATING RECTAINS THEN MANTAL BEARLERE BUTTO
HPITE(6, 930) (PT(1), IDIFE(1), I=1, JNS)	<u> </u>	5	PROSEST PREPARATION RESIDENCE INCOMPANY PROTOCOLOGY PROTOCOLOGY PROTOCOLOGY
1011E(5.50A) AVP. FA	3	C	ROOCHIES, LOAPASS FILTERS AND DEGINATES WRITTEN BY L.L. BUILT
	4		1MP1.1C17 1N1F(4R+2(I-N)
COLUMN TAND AND INCOME AND DEPENDENT SAD AN	5		INTEGER+2 IEAAD(S)
FORMALCING, THAN USED AS THE REPERENCE FFICE AN ARTES	6		DINENSION DEMOD(256), DEC(256)
FORMAL CHARTER OF FILL PULSE 714 (1016)	~ 7		DINENSION AVES(256), PHASE(256), FASE(256)
FORMATCING, VALUES OF SELECTED PITCH PERIOD HAD POL			EQUIVALENCE (PHASE(1), FASE(1))
+ DISTRICES //6(F10. 4, 16))			DATA 18400/6, 23, 42, 64, 90/
FORMATCING, 'DATA BLOCK NUMBER', 16)	18		LFB
60 TO 100			N=256
STOP	12		
END .	, 16		
	13		
•			RYGD(1)=0.0
	10		FHH2E(1)=0, 0
	1.6	10	CONTINUE
	17	C	
	18	C	READ IN DATA
MAX	1.9	C	
	29		KENTID 7
	21		ATHIND 1
	22		FENIND 2
	23		FEHDO 3
	24		RELITED 4
	25		REMIND A
	26		AFRICZ, END-99) AVES
	27		FF GU((A) PHASE
	26		HRITE(S. 100)
SURVITINE MAY (V. N. BIG. N. M.)	29	198	FORMATING CARAGE FORCESTING DI DEV ARABET, AV CARABARAST
	e		
DETERMINER THE MOUTHER OF AN ADDOUT MULT I PROP			
WE LE KELLINES THE MONTHUM OF HIS PROCHY BY C. L. BURGE		-	HYESCODESKICHYESCODD
THE IST THEFE AND AND	54	30	GORAL TRADE
Inpluit Integration	33	G	
DIMENSION X(1)	34	G	PROCESSING SURBANDS
N1-N-1	30	C	
00 20 1-1, N1	36		00 40 1-1.4
IF (685(X(I)), 0T, 885(X(I+1))) 60 TO 18	37		HRITE(5,110) I
BIG=X(I+1)	38	110	FORMAT (1HA), *++++ PROCESSING SURRAND +++++* ?
f##1=t+t	39)	CALL SPILIT (AVES, PHASE, IBAND, I, NB, L, DEMOD)
GO TO 20	-49)	CALL LOWPASCOEMOD, IRAND, 1, NR, L, FRSE)
BIG-X(1)	41		CALL SAMPLE (DEMOD, FASE, IBAND, I, NB, L, DEC)
X(1+1)-BIG	42		HRITE(I) DEC
CONTINUE	43	48	CONT INUE.
BIG-ARS(BID)	44		570*
PETURN	45		END
END	46	C	
		-	

PITCH

SUMLPD

SNRCAL

....

1

SNRCAL

	SUDPLOYIING SNEEDE.		<u>.</u>	UNTE THE COURSETED BEDETE
2	COLOR ATE SHE FLOW DES BOTHE STOND	49	2	BRATE THE GRADUITED REDUCTS
~	GREAT ALTE SHE FROM 200 FUINT STORM.		•	METTERS BIBL ENC. CONF. 6CNF
C.	1444 11117 1111 C.C			ENDIATION / ENG SAME / THE TIME / MACH CONE / 32, 518 3
	ITELICIE LICEPERAZULARIA	51	71.0	FORMALLING COLORS STATE FOR FOR STATE CONTRACT STATE
	DIDENSING STORPLOZODI STOEST (2007) WHECCODI HETURE (2007)	34		TIG TETT SALVE SALVE SALVE ST
	DINENDINI EKKOK(206)	53	33	5107
		514		E/R)
	DIRITIANI-II. B			
	5188AB1-0.0			
	R513-49, P			
С				
C	FEAD IN DATA			
C				Chicche
	FEHIND 1			
	FFWIND 2			
	PFAD(1, END-99) SIGNEL	1 .		SUBROUTINE UNCODE COLEVEL, BITS, KODE, NJ STONAL, IKODE)
	REHD(2) SIDEST	2	C	
С		3	C	DECODER FOR INPUT CODE, BY L. L. BURGE
C	CALCULATE CUMULATIVE SUM FOR SNR	4	C	
Ċ		5		IMPLICIT INTEGER+2(I-N)
	DO 18 1-1, N	6		INTEGER+2 KODE(1), IKODE(1)
	ACTIAL (I)+SIGNAL (I)+SIGNAL (I)	7		DINENSION SIGNAL(1)
	FEXOR(1)+(SIGURE (1)-SIGEST(1))+(STOURE (1)-STOEST(1))	8		16ITS=IFIX(6ITS+, 1)
		9		1(HL+2++1F175
		9.t		QINC+QLEVEL/FLOATCIGN.)
10	CONTINUE CONTRACTOR	11		QHEG-FLORT (10H)/2. 8
C		12		IGNEG-IFIX(ONEG+ 1)
2		13		10:05=10:4
1	UNLOUGHTE NORTHLIZED SHR & RES SHR	- 5.4		WEITE CO. 100 IONL, IBITS, GINC
L.		4 4	4.000	FORMAT (180, 1908) FP 1 FVF1 S1, 12, 24, (BITS1, 13, 24, (1801, F7, 3)
	5102 - 52743 4 71713 4154 41	4.6	2.00	DOL 10 TEL.N
	PMUS174141271	4.7		DO 20 ILL INNER
	SOMP-THANDGLOCSNR)	40		
	ENS-SORT (EM)	1.0		TELEVILLE CO TROPEZINA STONE ZINA EL DOTZIGNOTAL
_	PMS-29+ALOGLA(RMS)	3.7	24	PRODUCTS, Ed. IRUSENUSS SIGNERIS-FEDERING
Ç			20	
C	CALCULATE MEAN SQUARE EXROR	~1		
C		2.2.		DO 30 K=IQNEG, IGPUS
	DO 29 1-1, N	23		IF(KODE(I), EQ. INDDE(K)) SIGNAL(I)=FLOAT(K1)+GING
	IF(ACTUAL()) ER. 8. 83 ACTUAL()=0. 001	24		K1=K1+1
	AHRCI)=HBS(FRFOR(I)/ACTUHLCI))	25	30	CONTINUE
	JF (QNAY(I), EQ. 0, H) QNR(I)=0, 001	2.6	1.0	CONTINUE
	NNR(1)=10, 0+61 0010(0NR(1))	27		RETURN
	ASIR-UNR(1)+ASIR	28	•	END
20	CUNTIMF	2.9	. C	
	ASHK-ASHR/FLOHT (N)	38	C	

FFTMOR

FFTMCIR

			47\ C		
	c	CLEAR THE FRAME	48		CALL FFFT(51, Y1, NBITS, L, 1)
	•	TAGE 10TT TOTAL CONSTANT	49		CALL MAGSQ(Y1, S1, N, E, PHAZ)
	~		50		IN-IN+1
	2	UNCON MESS THE INCOMITIONE SCHEED OF FOURIER SPECTRUM	51		RE+E(256)/2. 0
		WEITER BY L. I. BUNGE	52		TE(IN EQ. 1) 00 TO 630
5	5	CHECKLERIES FIREFRY FOR N PTS	53		16(F(255) 1 1 A) (0 10 618
2	C C	SI - VECTOR OF IMPORT SAMPLES ON IMPUT		76	LIDTERS ALL TA BE
		SL - VECTOR OF FERL SPECTRUM POINTS ON OUTPUT FROM FET			FRANKTYANA CEPTION AT 14.4 TOTAL ENERGY FOR RICCY & S.FIE AT
	C	51 VECTOR OF INHOTHERY SPECTRUM POINTS OUTPUT FROM MAGSA			
•	C	VI - VECTOR OF MAGNITUDE SPECTRUM POINTS ON OUTPUT	67 6		COLOUR ATE EVEDAU ACIE BOARD BOARD
10	C.	N - WATEER OF POINTS	ST C		UNLIGHTIG EACHOFY SUD-DIAN DIAN
11	C	NRITSE - URDER OF FET	58 C		AND A DECEMBER THE
12	C	AVES - SPECTERL CURFFICIENTS	59		CHLL SURCELINI ESORI INI
13	C	そうしゃ うしもっとんけの だしじゃりく	69		EJJ=10, 0
14	C	SIGNA - NOFUH, 126TION CONSTRUCT	61.		E22-4. 0
15		INIF (4 P+2 KH175(3, 4)	62		£33=1.0
15		DINENSIGH S1(258), V(258), V1(258), F(258), 78175(4), BUEB(256)	63		SEN(IN)=E(256)
17		DIHLNSION AVES(238), VP(258), DATA(258), BINESTAN, BUARTAN	64 C		
18		DIFFICIENT RELEASE FOR ALL FOR DAY FOR THE CONFERENCE OF	65		IF(E(256), LT, 40, 0) 00 TO 200
19		FOULDER FOR FUSION STATES FOR THE RESIDENCE AND STATES STATES	66 C		
214		MATE WERE AN AND AND AND AND AND AND AND AND AND	67 C		CHECK IF HIGH OR LOH ENERGY BRSED ON PREVIOUS ENERGY
21			68 C		
22			69		CALL CHEKENCE, SEN, IN, ESUB)
~~		UNIT FELLOY 4, 3, 2, 3, 3, 3, 2, 3, 3, 2, 2, 37	79 2	98	E101-E(256)
~ ~ ~ ~ ~ ~		DRTH ENERGY/10. , 7. , 4. , 5. , 3. , 2. ,	71		TEGETOL LT. EX3. 00 TO 503
~		•24. , 16. , 8. , 16. , 18. , 18. , 9. Z	72 0		
		10.00 M / = 69	77 0		B PETORI CHICULATION OF NORMALIZATION CONSTANT
25		11=6	74 0		
27		NH: IT51=9	74 4		TE (ETO) (AT E11) AO TO 501
28		P4 1 I=1,256		•	TECTOL DI ELLY OU TO MAL
2.9		AVEP(I)=9.8	73		1F(E,10), 01, E222 00 10 502
<u></u>	1	AVES(1)=0.0	20 1		
31		DO 2 1 = 1,4	78 5	91	
32	2	ZBITS(I) = 0.0	79 5	3.3.	SIGPHC03=ENERGYC1, 377(18. ++(RB115(1, 3773, 321937)
33		C0H=31, 25	80		60 10 540
34		N=256	81. 5	92	00 522 J=1, 4
35		NBITS - HAITS1 -1	62 5	22	SIGHACI)=ENERGY(2, 0)/(10. ++(KHIT5(2, 0)/3. 32193))
36		l, • • #3	83		60 10 548
37	C		84 5	0 3	DO 533 J=1,4
38	Č.	READ DATA	85		\$16MA(J)=ENEROY(3, J)/(10. ++(K8115(3, J)/3. 32193))
39	č		86		1F(51GNA(J), EQ, 0, 0) \$10MA(J)=1. 0
44	-	REHIND 1	87 5	33	CONTINIE
41		FEWIND 2	88 5	48	CONTINUE
42		BEUIN 7	85		WRITE(6, 556) (\$10NA(37), J=1, 4)
		ECUTIN A	90 5	50	FORMAT(149, 151GMA VALUES =1,4F16. 6)
			91. 5	55	FORMAT (1HG, 'BITS/SAMPLE MATRIX'/1HG, 418/
	2	KGDV:1.4 CTV=731 31	52 -		+1H0, 419/1H0, 418)
	~				
	G	GREGERIE SPECTA AND ENERGY			

.

PPTMCIR

FFTMOR

	3 568	FARMATCING, CENEROY THRESHOLD MATRIX(2	3.39		WRITE(3) CRITS	
9	4 .	+1HD, 4116, 6/1HB, 4F16, 6/1HB, 4F16, 6)	1.48		WRITE(4) RVEP	
9	5 6		141	C		
ć	- r	FU PLATE STIC FOR SECTETED STONE	1.42	C	PLOTTING ROUNTINE	
		on the prior provide the store	1.43	C		
	· ·		1.44		HR11E(5,112)	
	, , ,		145	112	FORMAT ("DO YOU HANT A PLOT?")	
	7 64.945		146		READ(5, 113) ANS	
10		HF116(6,593) (B1(5(1),1+1,4)	447	443	FORMAT(184)	
10	L	TE115-8 0	1.10		TECOUS HE VES) ON TO 998	
16	?	G4 6A1 1=1.4	140		UPITECE (30)	
19	3 601	TRITS-RITS(1)+THITS	1.4.2	470	COCHOT // THENT COUNT NEWS A	
16	4	IF(16115 I.T. 2. 6) 00 TO 603	1.50	170		
16	5	17,0##IT=1C(##IT+3	1.51			
1.5	5	DO 602 1=1,4	1.52		CHEL PLUISCOUP, 200, 07	
12	7 692	78175(I)+28175(I)+8115(I)	153		CHLL PLOTOB 4, -11. 6, -37	
14	. 6.03	CONT THE	1.54		CHLL PLOT (3. 8. 7. 6, 3)	
10	. 665	FORDATCHAG, BITS ALLOCATED FOR SUBBRADS', 2% 4F12 3)	1.55		CALL SYMEOR, C2, 0, 7, 0, 14, PHONERE 7, 0, 0, 07	
11	A	60 10 651	156		CALL, SVII-04. (999. 0, 999. 0, 14, CHAR, 0. 0, 4)	
11	61A	HELIF(6, 420) IN	1.57		CALL, PLOT(A, A, 1, 0, -3)	
11	2 620	FORMATCHEW, SECTION, 14, 15 PREDOMINATELY NOISE',	158		CALL SCALE (DATR. 6. 0, 256, 1.)	
11	1	+ +++++++ NO ENERGY CALCIE ATED')	1.59		CALL SCALE (XP, 8. 0, 256, 1.)	
	. r		160		CHLI, AXISCO, 0. 0. 0. 9. SHEREQUENCY9. 8. 0. 0. 6. XP(257). XP	(256))
	č	STIMATION OF RUFPLE SPECTOM AND EVERAGE BITE	161		CHLI, AXIS(0, 0, 0, 0, 11HMA0 SQUARED, 11, 6, 0, 90, 0, DATA(2	57), DRTA(258)). A.
		ESTIMATION OF AVENUE PEGTICAL AND AVENUE PITS	162		CHLL LINE (XP, RVES, 256, 1, 0, 0)	
			163		CHI PIOT(10 8. 6. 6999)	
11	r out	D4) 14 [=1, 255	164	499	STIP	
11	8	AVES(1)=AVE5(1)+Y1(1)	445	222	END	
11		AVEP(1) HVEP(1) + PHAZ(1)	4.55	~		
12	9 10	COUT 11496	1.00	ž		
12	1	60 10 5	3.07	C.		
12	2 99	10 20 I-1,256				
12	š.	AVES(1)=AVES(1)/16. 8				
12	4	HVFP(1)=HVEF(1)/16. 0				
12	5 29	CHAT LINE				
12	5	DD 694 I=1, 4				
12	7 .664	ZBITS(I)=ZBITS(I)/16. 0				
12	4	D0 30 I+1,256				
12	9	XP(1)=(1-1)+C(N				
17	9 30	CC017 11#JE				
. 13	1	HPITE(6, 555) ((KBITS(1, J), J=1, 4), J=1, 3)				
17	2	WRITE(6.560) ((ENERGY(1.1).1=1.4).1=1.3)				
1 1	2	WPITE(6.565) (28175(1).1m1.4)				
	4 565	FORMATCHING AVERAGED BITS ALL OCATED TO SUBRAND . 24. 454 3)				
	2 ×	WITE ANT AGTA AN ALSK AND PRINTER				
		MALLE VAL ANIH ON MICH AND FRINKER				
13		WELLELZJ AVED				

-

MAGSO

2

9

1A

11

12

14

15

LATTIC

1

67

8

9

10

11.

1.2

13

14

15

16

17

18

1.9

20

23

24

25

26

27

28 29

30 31

32

33

34

35

40

42

43 44

45

46

21. 22

SUBROATINE MOUSOCK, VI, NT, E, PHBZ) IMPLIGIT INTFOFR+2(1-N) PROGRAM COMPUTES MAGINITUDE SOUNCED OF THE ARRAYS 3 C WEITTEN HY L. I. RAFGE X = FIFST REFEY FOR INFUT AND MAGINITUDE AKRAY FOR OUTPUTYI. - SECOND RAPRY FOR INFUT4 C 5 C 6 C 7 C NT - MANARK OF POINTS 8 C E - FREEGY DINENSION X(1), Y1(1), E(1), PHR2(1) DO 10 J # 1.NT PHHZ(J)=AIAH2(X(J),Y1(J)) X(J)=X(J)+X(J)+Y1(J)+Y1(J) CHLL FNEFGY(X, NT, E) 10 PF. TUPH FIID 16 C 17 C

ENERGY

1		SUPPORT THE ENERGY (X, NT, E)
2		INFI ICIT INTEGER+2(I-N)
3	С	THIS PROGRAM CALCULATES THE ENERGY PER N POINTS
4	Ċ	WRITTEN RY L. L. BURGE
5	č	X - MAGUITURE SQUARED INPUT ARRAY
6	С	NT - MANBER OF POINTS
7	C	E - ENERGY
8		DIMENSION X(1), E(1)
		L = NT - 1
		E(1) = 0.0
11		DO 10 J = 1.1.
12		E(J+1) = (8.8882444)+X(J)+E(J)
13	10	CONTINUE
4		RETURN
15		ELD

IMPLICIT INTEGER+2(1-N) IMPLEMENTATION OF LATTICE FILTER BY L. L. BURGE A), S1 - INFUT OF SPEECH SAMPLES N - NO. OF POINTS M - GEOFR OF FILTER KC - REFLECTION COFFFICIENTS C RO - PREDICTION RESIDNAL OF SPEECH PTS OUTPUT C B - VECTOR OF PHENNARD PREDICTED SAMPLES Ĉ C - TENHIWARY VECTOR FOR SAMPLES DINENSION #(11). A1(256). A(11). ALPHA(11). RC(11). ARKAY(44) ... DINENSION C(11, 256), 8(11), 51(11) DIMENSION FO(258) DIMENSION XN(256), DATA(258), DATAL(258), BUFF(58) EQUIVALENCE CAREAV(1), K(1)), CAREAV(12), A(1)) EQUIVALENCE CHERAY(23), ALPHA(1)), (AKRAY(34), KC(1)) EQUIVALENCE (DATA(1), RO(1)) EQUIVALENCE (PATAL(1), A1(1)) DATA VES / YES '/ H=10 N=25-5 NS=A IFEC-9 CON=1, 28068, С READ HINDONED DATA FENINO 1 FEHIND 2 FEHIND 3 READCH. END-99) AL 5 FEHIND 5 IREC=IFEC+1 DO 10 1-1, N 1.0 XN(1)=(1-1)+CON FEAD(2) AFRAY 8(1) = 0, 0 DO 48 K = 1, N C(1, K)=H1(K) IF(K. 0T. 1) 8(1)-81(K-1) DO 38 J=1, M 51(1)=R1(K) 51(J+1)=51(J)+&C(J)+&(J) G(J+1,K)=B(J)+RG(J)+51(J) IF (K. EQ. 1) GO TO 28 8(J+1)=0(J+1, K-1) 60 TO 50 20 H(J+1.)=0. 0 50 RU(K)=SI(J+1)

LATTIC

.

LATTIC

47	30	CONTIME				
48.	49	CONTINUE		93		STOP
49		HF-17E(3) #0		94		END
- 0		M1=1+50				
51		W/11-(6,100) (RD(1),1=1,01)				
52	189	FOR HALCING, "RESIDING WELLES FROM LATTICE FILTER"				
53		A(1#+12 6))				
54		115=115+3				
55		HEIJE (SAA) NS				SNR
56	с	FLOTTING SCHETING				
57	- ·	L# 17F (5, 119)				
58	118	FORTHIC'IS & HOT DESIRED?')				
.9		FERUS, 120) WIS				
68	129	FEENATCIAA				
61		IF (FHS HE YES) AD TO 5				
62		HEIJE (5, 130)				
67	130	FIFTIGIC DEUT SOLNO HEMES				
£.4		FFHD(5, 120) (HAK				
65		FEALW = FLOAT (IFFC)		1		SUBROUTINE SNRCKEST, RHGT, N, GNRG SUNR)
6.6		CH 1 A (1)5(H)(F. 204. A)		2	C	
67		$[B] \downarrow [F] (1] (0, 0, -11, 0, -3)$		3	C	CALCULATES SNRI WRITTEN MY L. L. HURGE
6.94		$C_{1} = 1 + 1 + 1 + 1 + 1 + 1 + 1 + 1 + 1 + 1$		4	С	
69		CHIL SYNE (-1 8.3 8. 14. PHONEME 1.98.8.8)		5		INFLIGIT INTEGER+2(I-N)
78		FWI 547464 (994 8.999 4. 14. FHER. 48 8.4)		6		DINENSION REST(1), RACT(1), ANR(1), ACTUAL (256), EXROR(256)
71		FRI _ 5116/4 (463 8, 939 8, 14, FERME 1, 69 8, 6)		7		DURISURIEO. O
72		(GL) MULLE (944 0. 964 0. 14 FEMILE 40 01)		8		SUNCHMED, O
2				9		DO 20 I=1.N
74		CHI 5 (BE FUETA & A. 285.1)		18		ACTURE CID+RACTCID+RACTCID
75		CHE 57 M + (2)1. 4 (0, 256, 1)		11		EKFORCID=(RACICID-RESTCID)+(RACTCID-RESTCID)
76		CHI & AXIS(0 0.4 0.13HTIME INTERVAL 13.4 0.0 0.		1.2		DUMSUNI=ACTURE CIS+DUMSUM
77		+x1(252), x1(252)		13		SUNDIM=FPRORCID+SUNDIM
78		CALL 6X15(0 0.0 6.845551044 .8.4 8.90 8.		1.4	28	CONTINUE
79		+(618(257), 6818(258))		15		SQUE - SUMDOM/DUMSUM
24		(614 L 11/E (X3), 60, 256, 1, 0, 0)		1.6		SOME-LOAPS OUT CONTROL
81	500	FORUNT (1H0, "PRESENTLY PROCESSING DATA THRU LATTICE FILTER		1.7		DO 16 I=1, N
R2		+ 1/1H .' FOR SPEECH SECTION 1, 15)		18		IF (NACTCI), EQ. 0. 0) NACTCI)=0. 032
83		CALL SCALE CONTAL 4. A. 256, 1)		1.9		HCTUH_CI)=RHCTCI)+RHCTCI)
R4		CHIL AXIS(9 0.5 0.13HTIME INTERVAL -13.4.0.0.		20		ERROR(I)=(RRCT(I)-REST(I))+(RHCT(I)-REST(I))
85		+X1(257), X1(258))	· · ·	21		QURCID=HESCERRORCID/HCIDHLCIDD
86		CALL AXIS(0. 9. 5. 0. SHSPEECH, 6. 4. 0. 90. 0.		22		IF CONRECTO, E.O. 0. 00 CONRECTO = 0. 0051
87		+DATA1(257), DATA1(258))		23		RNR(I)=10, 0+AL.0010(GNR(I))
HA.		CALL PLOTTA 4.5 83)		24	10	CONT INCE
89		CALL LINE (21), A1, 256, 1, 0, 0)		25		RETURN
99		CALL M. 01(6. 8. 8. 83)		2.6	-	END
91		60 TO 5		2.7	C	
42	99	CAL MOTO 8.8 8999)		28	C .	

SUB

FFT

1 2	C	SURKOUTINE SUR(YL), N. ESUR, IN) YL – VECTOR OF MARAITULE SPECTRUM POINTS N – NIRBER OF KOUTS	•	12	c
	ř			3	
- 2	5	THE ELEMENT OF SOME PROPERTY OF COTFOR		4	C
		IN - FFICE PATER		5	C
-	C	THIS FOUTTHE ESTIMATES ENERGY/SUB-BAND BY L L BUILDE		6	C
		INFLICIT INTERPENZIEN		7	C
		Dimension Vicio, Esurceo			C
				9	č
30		ESUR2=6 8		16	
11		F 51 #2 3 = 6. A		4 1	
12		E \$1#:4=61 @		42	
13		P15-4095		- 12	c
14		IF IL 1-23		1.4	č
15		IF JL 2=42			č
14		IF1L3=64		10	U .
17		IF 1L 4=90		1.6	
18	C			17	1.0
19	C	FSTINGTION OF NORMLIZED SUB-EAND DISTRIBUTIONS		18	
26	С			: 1.9	
21		DO 19 1-1, 1F1L1		20	
22	18	F5UR1=(1, 0/P15)+Y1(1)+E5UB1		21	
23		ESIR1+2 +FSIR1		22	
24		FSUB(1) = FSUB1		23	
25				24	
26	24	60169214 624161441414466169		25	20
37	20	E (2010)		26	C
20				27	Ċ
20		52 TO X 2 - 52 TO 2 X 2 X 2 X 2 X 2 X 2 X 2 X 2 X 2 X 2		29	C
20	20	27) 27 1-177227 17123 EC102-271 8/072471252107		29	30
	30	ESTANDA SETUD		36	
21				31	
				32	
		LA, AD INIFILS, IFILA		37	
	417	F5/8/4+1, 8/2/15/441(1)+25084		34	
		E SUBSER AL STRAG		35	
				76	
		NP11E(0,100) IN. (ESUB(I), IA1, 4)		32	
58	100	FORTH I CINE, SUB-BAND ENERGY DISTRIBUTED FOR SECTIONS",		38	
		+14/17ki, 4F 16. B)		20	
49		PETIMH		10	
41		END	and the second second		
42	r,			41.	447
43	C			42	
44	C			43	
				44	
				45	G
				- 46	C

	SUPROUT THE FFFTCAS TISTS COLUMN
•	
	INFLICIT INTEGER+2(1-H)
	THIS ROUTINE IS A REVISION OF
6	FAST FOURIER TRANS WITH PRUNING BY J. D. MARKEL
	X - VECTOR OF REAL DATA POINTS
C	V1 - VECTOR OF IMAGINARY DATA POINTS
ē -	M - ORDER OF FFT 2++M=NO. OF PTS.
ř.	1 - OFLER OF PRUMED DATA (N(2++L) LON
.	DIMENSION X(1), Y1(1)
	N = 244M
~	
	SEDO THE VECTLE FOR IMPOINTER POINTS
6	ZERO THE VECTOR FOR THEITWEET FORTE
C	
	DO 10 I = 1,256
1.0	91(1) · 0.0
	TWOP1 = 5.283185
	00 40 LO = 1 M
	LMX = 2++(H-LO)
	LMM = LMX
	L1X = 2+1.MX
	SCL1 - THOPIAIX
	IF (LO-N+L) 20, 30, 30
28	L111 - L2
c	
č	TTRAN 15 THE DIRECTION OF THE TRANSFORM
č	
ັນຄ	00 40 LM = 1.1MM
	APR = (IN-1)+SCI 1+17RAN
	C = C(S(AFB))
	S & SIN(APG)
	$\frac{1}{10} = \frac{1}{10} \frac{1}{10} \frac{1}{10}$
	11 = 0.017 - 0.047
	$1_{\mathbf{A}} = 1_{\mathbf{A}} (1_{\mathbf{A}}) - 1_{\mathbf{A}} (1_{\mathbf{A}}) $
	A(01) = A(01) + A(01) + A(02)
	$\mathbf{Y}_{1}(\mathbf{J}_{1}) = \mathbf{Y}_{1}(\mathbf{J}_{1}) + \mathbf{Y}_{1}(\mathbf{J}_{2})$
	X(J2) = (+11+5+12
48	$Y_1(J_2) = C + T_2 - 5 + T_1$
	CALL PRT(X, YL, M)
	RETURN
	END
C	
C	

3		SHEPTINE PRICE YI. N
2		INVLICIT INTEGER+2(I-N)
3	С	POITTINE PEVERSES AITS, 8 VI
- 4	C	A REVISION OF J. D. HERKEL
5	C	X - PERK FFT POINTS
- 6	с	Y1 - INHGINARY FET POINTS
7	C	H - OFIER OF FFIT
		INTEGER+2 L(9)
•		DIFENSION X(1), VI(1)
19		D() 28 J=1,8
11		L(J) = 1
12		IF(J-M) 18.18.28
1.1	18	1(J) = 2++(M+1-3)
1.1	29	C CALT TANKE
15		TH a 1
16		
17		
1.4		
20		
2		141 54 JA & J7, 18, 17
~		LD = L(D)
25		DO 59 35 - 36, 15, 16
~		14 = 1.(4)
		DO 50 34 - J5, L4, L5
<u></u>		L3 # 1.C32
20		100500J3 = J4, L3, L4
~		$L_2 = 1.(2)$
29		DO 50 J2 - J3, L2, L3
19		L1 = 1.(1)
31		00.50 JL = J2, L1, L2
32	_	1F(3)1-31) 30 ,30,40
33	34	R . XC300
34		X(JN) = X(J1)
3		X(J1) = R
36		FI - YICJND
37		V1(JN) = V1(J1)
` (H		Y1(J1) = F1
39	48	JN = JN + 1
49	56	CONTINUE
41		CALL USCANCE, VI. H)
42		FETURN
43		FND
44	C	
45	C	
46	ĉ	

SURPORTINE PRICE MI. IN			
THE TOTT THOFFERADOLAND			
MAITTINE REVENSES ATTS & UTS	-		-
A REVISION OF J. D. HARSEL	POSITINE	ERCHI DUE	10 FF1
X - FER FET POINTS			
Y1 - INHGINERY FET POINTS			
M - OFIER OF FEIT			
INTEGER+2 L(9)			
DIMENSION X(1), VI(1)			
D() 28 J=1,8			
L(J) = 1			
IF(J-H) 10, 10, 20			
L(J) = 2 + (M+1-3)			
CONT IN F			
JH = 1			
L8 = L(8)			
DO 58 J8 = 1,L8			
1.7 • 1 (7)			
DO 58 J7 N J9, L7, L8			
L6 # L(6)			
DO 59 J6 ► J7,L6,L7			
L5 = L(5)			
DO 59 J5 - J6, L5, L6			
14 = 1(4)			
00 50 34 - J5, L4, L5			
1.3 = 1.(3)			
00.50.03 = 04, 13, 14			
$L_2 = L_2$			
DO 59 J2 + J3, L2, L3			
$D(1, 24) = J_2, L_1, L_2$			
1F(311-31) 30 , 30, 40			
FI = TI(JN)			
\mathbf{v}_{1}			
ATCOTA = NT			

RUT

1		SURROUTINE USCRNCK, YL ND	
2		INPLICIT INTEGER+2(1-N)	
3	C	THIS FROGRAM UNSCRAMPLES FOURIER TRANSFORM OF REAL DAT	18
4	C	DUE TO FFT AFTER S. B. DAVIS	
5	C	X - AFRAY OF REAL TRANSFORM POINTS	
6	С	V1 - AREAY OF INFOLNARY TRANSFORM POINTS	
7.	C	M - UNDER OF TRANSFORM	
8		DINENSION X(1), VI(1)	
9		I = 2++M	
10		102 = 1/2	
1.1.		P1 + 3. 14159265	
1.2		SA = PI/I	
13		X(1) = X(1) + Y1(1)	
1.4		Y1(1) = 0.0	
15		LL = 1.02 + 1	
1.6		VIGLEDVIGLED	
17		D0 10 K = 2,102	
1.8		J # I-K+ 2	
1.9		AHO - 58+1K-1)	
20		C = COS(ANG)	
21		5 - SIN(ANG)	
22		6A = X(K) + X(J)	
23		BB = Y1(K) - Y1(J)	
24		CC = X(K) - X(J)	
25		DD = Y1(K) + Y1(J)	
26		XR = C+DD-S+CC	
27		XI = 5+DD+C+CC	
28		X(K) = (AA+XR)+. 5	
29		X(J) = (HA - XR)+, 5	
30	1.6	¥1(J) = (-68-XI)+. 3	
31		RETURN	
32		END	
33	С		
34	C		
33	č		
36	č	· · ·	
	-		

USCRM

APPENDIX C

ARTICULATION INDEX

ARTICULATION INDEX

The concept of the Articulation Index (AI) was advanced by French and Steinberg [86]. It is defined as a number obtained from articulation tests using nonsense syllables under the assumption that any narrow band of speech frequencies of a given intensity in the absence of noise carries a contribution to the total index, which is independent of the other bands with which it is associated, and that the totals of all the bands is the sum of the contributions of the separate bands [86]. It must be proven that there is a unique function relating syllable or word articulation to AI for any given articulation crew and choice of word list. In determining AI under these conditions, there are essentially two parameters of a linear communication system that can be varied: (a) the level of the speech above the threshold of hearing, and (b) the frequency response of the system. Here a linear system that is free from noise is assumed.

A curve of AI versus frequency is included from French and Steinberg [86] in Figure 32. The curve is derived from the syllable articulation gain and frequency responses of speech waveforms [86]. The syllable articulation is expressed as the percentage of syllables with which consonant-vowel-consonant of meaningless monosyllables are perceived correctly.

Baranek [87] pointed out two important facts. First, extending the frequency range of a communication system below 200 or above 6000 Hz contributes almost nothing to the intelligibility of speech. Second,

ARTICULATION INDEX VS. FREQUENCY



Figure 32. Composite Articulation Index vs. Cutoff Frequency of Ideal Lowpass Filters (After French and Steinberg, 1947)

each frequency band shown in Table XXV makes a 5 percent contribution to the AI, provided that the orthotelephonic gain of the system is optimal (about +10 dB) and that there is no noise present.

TABLE XXV

No.	Edges of Band	Mid-Freq (Mean)	No.	Edges of Band	Mid-Freq (Mean)
1	200 - 300	270	11	1600 - 1830	1740
2	330 - 430	380	12	1830 - 2020	1920
3	430 - 560	490	13	2020 - 2240	2130
4	560 - 700	630	14	2240 - 2500	2370
5	700 - 840	770	15	2500 - 2820	2660
6	840 - 1000	920	16	2820 - 3200	3000
7	1000 - 1150	1070	17	3200 - 3650	3400
8	1150 - 1310	1230	18	3650 - 4250	3950
9	1310 - 1480	1400	19	4250 - 5050	4650
10	1480 - 1660	1570	20	5050 - 6100	5600

FREQUENCY BANDS OF EQUAL CONTRIBUTION TO ARTICULATION INDEX

The orthotelephonic (OT) gain is defined by

OT Gain (Subjective) = 20 log (e_0/p_0) + 20 log (E_2/e_0)

+ 20 log
$$(P_1/E_2)$$
 (A.1)

where

 P_1 = free field pressure necessary to produce the same loudness in the ear as was to produce by the earphone with voltage E_2 across. e_0 = voltage produced by the microphone across the input resistor of the amplifier by a voice which produces pressure p_0 at a distance of one meter in a free field. E_2/e_0 = voltage amplification of the amplifier.

OT Gain (Objective) = 20 log
$$(e_0/p_0)$$
 + 20 log R

where

- R = ratio of the pressure produced at the eardrum of a listener by a source of sound to the pressure which would be produced by the same source at the listener's head position if he were removed from the field.
- p_e = pressure produced at the eardrum of a listener by the earphone with a voltage e₂ across it; others are the same.

The AI obtained per frequency band in Table XXV is successively added to arrive at the total AI.

APPENDIX D

SONAGRAMS



(temale)











VITA

Legand L. Burge, Jr.

Candidate for the Degree of

Doctor of Philosophy

Thesis: EFFICIENT CODING OF THE PREDICTION RESIDUAL

Major Field: Electrical Engineering

Biographical:

- Personal Data: Born in Oklahoma City, Oklahoma, August 3, 1949, the son of Mr. and Mrs. L. L. Burge.
- Education: Graduated from Douglass High School, Oklahoma City, Oklahoma, in May 1967; received Bachelor of Science degree in Electrical Engineering from Oklahoma State University in 1972; received Master of Science in Electrical Engineering from Oklahoma State University in 1973; completed requirements for the Doctor of Philosophy degree at Oklahoma State University in December 1979.
- Professional Experience: Engineer Trainee, Oklahoma Gas and Electric, summers 1969-71; student teaching assistant, School of Electrical Engineering, Oklahoma State University, 1970-71; graduate teaching assistant, School of Electrical Engineering, Oklahoma State University, 1971-72; United States Air Force, 1973 to present.

Professional Organizations: Member of Institute of Electrical and Electronic Engineers, Sigma Xi, Eta Kappa Nu.